University of Alberta

Compound Operators and Infinite Dimensional Dynamical Systems

by

Qian Wang



A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

 \mathbf{in}

Applied Mathematics

Department of Mathematical and Statistical Sciences

Edmonton, Alberta, Canada

Fall 2008



Library and Archives Canada

Published Heritage Branch

395 Wellington Street Ottawa ON K1A 0N4 Canada

Bibliothèque et Archives Canada

Direction du Patrimoine de l'édition

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 978-0-494-46445-8 Our file Notre référence ISBN: 978-0-494-46445-8

NOTICE:

The author has granted a nonexclusive license allowing Library and Archives Canada to reproduce, publish, archive, preserve, conserve, communicate to the public by telecommunication or on the Internet, loan, distribute and sell theses worldwide, for commercial or noncommercial purposes, in microform, paper, electronic and/or any other formats.

The author retains copyright ownership and moral rights in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

AVIS:

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, publier, archiver, sauvegarder, conserver, transmettre au public par télécommunication ou par l'Internet, prêter, distribuer et vendre des thèses partout dans le monde, à des fins commerciales ou autres, sur support microforme, papier, électronique et/ou autres formats.

L'auteur conserve la propriété du droit d'auteur et des droits moraux qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis. Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant.



To my parents and my family in China.

Without their selfless support, this would not have been possible.

ABSTRACT

This thesis studies the qualitative theory of linear and nonlinear infinite dimensional dynamical systems with applications mainly to parabolic partial differential equations. The objective of the study is to examine through linearization the local and global behaviour, including existence and nonexistence, of invariant structures such as equilibria and periodic solutions.

In the linear theory, the dimension of the asymptotically stable solution subspace of a linear differential equation is studied. This gives new insights into the behaviour of linear and nonlinear dynamical systems.

The nonlinear results include such topics as a generalization to infinite dimensional differential equations of a classical stability condition of Poincaré. The main idea is that a periodic orbit is stable if the system diminishes nearby 2-dimensional areas. Similar considerations give conditions for the existence as well as the stability of a periodic solution. If the system diminishes areas globally rather than locally, it is shown that nontrivial periodic solutions can not exist; this is a generalization of the wellknown 2-dimensional Bendixson condition for the nonexistence of periodic solutions.

Examples of applications to concrete differential equations are given throughout and the thesis concludes with an application of the Bendixson condition to an epidemiological model.

ACKNOWLEDGEMENT

I am grateful to all the people who made this thesis possible. First and foremost, I would like to express my deep and sincere gratitude to my supervisor, Professor James Muldowney. During my doctoral study, he never failed to be friendly and supportive and always had time to help and discuss problems. His enthusiasm, inspiration and personal guidance helped to make research life smooth and rewarding for me. Without his patience and generous support, as well as his critical comments during the thesis writing period, this thesis would not have come as far as it did. I simply could not wish for a more dedicated supervisor.

I wish to express my appreciation to my co-supervisor, Professor Michael Li, for many valuable suggestions and constructive advice especially during the development of applications of this thesis. His organization of a very productive research group and laboratory was especially valuable.

I thank my mathematics teachers in China and Canada and departmental support staff who have assisted me in many different ways. Special thanks to Professor Thomas Hillen for two courses which increased my knowledge in the area of partial differential equations and to Professor Jack W. Macki for taking great interest in my doctoral research as well as providing suggestions to improve my presentation skills.

I thank Shawn Desaulniers, for his love and support over the years. My many student colleagues, especially Hongbin Guo and Zhisheng Shuai, provided a stimulating and friendly environment in which to learn and grow.

Financial support from an Izaak Walton Killam Memorial Scholarship, a F. S. Chia PhD Scholarship, a R. Pundit Sharma Scholarship, a Josephine M. Mitchell Graduate Scholarship and a Josephine M. Mitchell Research Prize is gratefully acknowledged.

Contents

Introduction 1								
1	1 Compound Operators and Compound Equations 5							
	1.1	Exteri	or Products	5				
		1.1.1	Vector Space: Hamel Basis	8				
		1.1.2	Normed Vector Space: Schauder Basis	9				
		1.1.3	Function Space	11				
		1.1.4	$L^2(\Omega)$	11				
		1.1.5	Interpretation	13				
	1.2	Comp	ound Operators	14				
		1.2.1	Representation of Compound Operators	16				
		1.2.2	The Laplacian Operator	20				
	1.3	Comp	ound Differential Equations	21				
2	Din	nensio	n Problems	26				
	2.1	Asym	ptotically stable subspaces	27				
		2.1.1	Function Space	30				
		2.1.2	Normed Vector Space	31				
		2.1.3	Differential Equations	32				
	2.2	Dimer	nsion Problems for Steady State Solutions	35				
3	On	Poinca	aré's Stability Criterion for Periodic Orbits	47				
	3.1	Orbita	al Stability of Periodic Solutions	48				
	3.2	React	ion Diffusion Systems	55				
	3.3	Effect	of Diffusion	64				
		3.3.1	Stability of Periodic Solutions	67				

		3.3.3 Planar Reaction Diffusion Systems	75
	3.4	Scalar Reaction Diffusion Equations	81
4	Cor	nvergence Theorems	96
	4.1	Existence of Stable Steady State Solutions	98
		4.1.1 Proof of Theorem 4.9	102
	4.2	Existence of Stable Periodic Solutions	107
		4.2.1 Proof of Theorem 4.12:	112
5	Ber	ndixson Criterion	126
	5.1	Surfaces and Boundaries in Banach Spaces	128
	5.2	Bendixson Criterion	136
6	An	Example from Epidemiology	145
	6.1	Existence and Boundedness	146
	6.2	Nonexistence of Periodic Solutions	156
A	Loz	inskiĭ Measures	158
В	Sec	torial Operators	162
	B.1	Sectorial Operators and Analytic Semigroups	162
	B.2	Fractional Powers of Operators	163
Bi	blio	graphy	166

List of Tables

.

2.1	Norms of k th exterior products \ldots	·	•	•	•	 •	•	•	•	•	•	•	•	32
A .1	Norms and Lozinskiĭ measures of ${\cal A}$												•	160

List of Figures

1.1	Exterior product $u^1 \wedge \cdots \wedge u^k$
2.1	Stability of steady state solutions in the invariant subspace S_1
	and S_2
2.2	Graphs of $y = 1 - \varepsilon - \frac{\varepsilon^2}{8}$ and $y = -3 + 7\varepsilon + \frac{\varepsilon^2}{8}$
3.1	$\varepsilon = 0.2$. The black trajectory is $u^2 + (u')^2 = 2.2^2$ 90
3.2	$\varepsilon = 1$. The black trajectory is $u^2 + (u')^2 = 2.9^2$
3.3	$\varepsilon = 10$. The black trajectory is $u^2 + (u')^2 = 14.5^291$
4.1	A measure of the angle between x^1 and x^2
4.2	Evolution of the oriented infinitesimal parallelogram $v^1 \wedge v$ 122
5.1	The boundary of the 2-dimensional surface h in X
5.2	A sector T at 0 in \mathbb{R}^2
5.3	The map $\tilde{\mathfrak{R}}$. Note $v(\pm \varepsilon) = (P_1 \psi(\pm \varepsilon), P_2 \psi(\pm \varepsilon))$ 135
6.1	A diffusive SIR model

Introduction

This thesis extends the theory and applications of compound matrices to differential equations in Banach spaces. The research developed provides new tools for the study of global behaviour of infinite dimensional dynamical systems.

Most of the developments on compound operators have occurred in the context of linear and multilinear algebra. Results on Hilbert spaces are discussed in [1, 7, 8, 94, 104]. A classical body of work dealing with algebraic aspects of multiplicative compound operators in \mathbb{R}^n , usually called compound matrices, can be found in [3, 33, 47, 67, 109]. One of the best-known results is the Binet-Cauchy identity. Good historical summaries are [75–78, 91]. In contrast, the literature on additive compound matrices is quite sparse. In the final chapter of the lecture notes [110] Wielandt discusses algebraic and spectral properties of both multiplicative and additive compound matrices. The same approach is taken in the book of Marshall and Olkin [67]. Fiedler [31] presents algebraic aspects of additive and multiplicative compounds in a coordinate-free setting and has applications to stochastic matrices.

Apart from the Abel-Jacobi-Liouville equation, which is also the formula for the determinant of an $n \times n$ matrix solution of a linear differential equation, applications of compound matrices to differential equations begin in the 1970s. Other cases are considered for special equations by Mikusinski [73], Nehari [82] and less directly by Hartman [41], Corollary 3.1, Chapter IV. However, the first treatment in full generality is due to Schwarz [95] in his study on the total positivity of fundamental matrices to general linear systems. London [61] derives a large number of interesting properties of additive compounds based on the relationship between a linear ordinary differential equation and its compound differential equations, and shows how properties of compound matrices may be used to greatly simplify many classical spectral inequalities.

Muldowney [76–78] systematically develops the theory of compound differential equations in his study of linear and nonlinear problems of differential equations in \mathbb{R}^n , such as dichotomy and stability theory, orbital stability of periodic solution of nonlinear autonomous systems and establishes a new approach to higher dimensional criteria for the non-existence of periodic orbits analogous to the conditions of Bendixson and Dulac for planar systems. Li and Muldowney [52–54, 56, 58] further explore problems such as lower and upper estimates for the Hausdorff dimension of the global attractor of dissipative dynamical systems, higher dimensional generalization of the criteria of Bendixson and Dulac for the nonexistence of periodic solutions, the existence and stability of equilibria and periodic orbits of nonlinear autonomous systems on invariant manifolds and global stability problems. They solve a long-standing open problem on the global stability of the endemic equilibrium of the SEIRS models with general nonlinear incidence rate in epidemiology.

Temam [104] reviews the definitions of the exterior product of Hilbert spaces and of two multilinear operators, which are the additive and multiplicative compound operators of this thesis. Emphasis is placed on investigation of bounds on the growth of exterior products of solutions of differential equations in terms of Lyapunov exponents rather than specific representations for the compound operators. Central to the discussion in Chapter V of [104] is the behaviour of distance, area, and, more generally, k-volumes which are generated locally by the semigroup $\{S(t)\}_{t\geq 0}$ of a dynamical system. This book gives an exposition of the work of P. Constantin showing that the evolution of oriented k-dimensional volumes is related to the k-th multiplicative compound operator of the differential of the map S(t). Applications to an upper estimation of the exponential decay rate of the volume and the Lyapunov exponents to partial differential equations are also discussed in Chapter VI. The concept of evolution of k-volumes in the dynamics is also emphasized in this thesis.

In Chapter 1, the exterior product of vector spaces and additive and multiplicative compound operators are defined and compound differential equations associated with linear differential equations in a Banach space X are discussed with examples.

In the linear theory, Chapter 2 investigates the codimension of the asymptotically stable solution subspace of a linear differential equation, which gives new insights into the qualitative theory of linear and nonlinear dynamical systems. The objective of the study is to examine, through linearization, the behaviour of nonlinear dynamical systems in the neighbourhood of particular invariant structures such as equilibria and periodic orbits.

Chapter 3 establishes a generalization of the 2-dimensional Poincaré's stability criterion to differential equations in Banach spaces. It is well understood that for periodic orbits of evolutionary differential equations, the moduli of Floquet multipliers determine their stability characteristics. However, the problem of directly estimating the Floquet multipliers is inherently difficult. It is shown that the orbital stability of periodic solutions is equivalent to the stability of a related compound linear system. This permits the use of simpler techniques such as Lyapunov functions in the estimation of the multipliers. Applications to reaction diffusion equations give new insight on the effect of diffusion terms.

Chapter 4 discusses the structure of omega limits of differential equations in Banach spaces. A criterion for the existence of periodic orbits for differential equations in Banach spaces is developed. The finite dimensional motivation is the Poincaré-Bendixson theory. The characterization of the stability of steady state solutions in terms of stability of linearizations and orbital stability of periodic orbits in terms of stability of second compound differential equations, discussed in Chapter 2 and Chapter 3 respectively, are special cases of this chapter.

In Chapter 5, an infinite dimensional analogue of the Bendixson criterion for the nonexistence of periodic orbits is established. The generalized Bendixson criterion states that, if some measure of 2-dimensional surface area tends to zero with time, then there are no closed curves that are left invariant by the dynamics. In particular, there are no nontrivial periodic orbits, homoclinic loops or heteroclinic loops. In this chapter, the Bendixson conditions for general nonlinear differential equations are developed in terms of stability of associated compound differential equations.

In Chapter 6, the results of this thesis are applied to a diffusive SIR model. A SIR model is an epidemiological model that is used to study the spread of an infectious disease, such as measles, mumps and rubella. In this chapter, the Bendixson criterion of Chapter 5 is used to establish the nonexistence of periodic orbits under certain conditions. Appendix A gives a brief synopsis of relevant properties of Lozinskiĭ measures for linear operators. These arise when norms are used as Lyapunov functions in stability analysis of linear systems.

Appendix B is a summary on sectorial operators and some notations which will be used throughout this thesis.

Chapter 1

Compound Operators and Compound Equations

This chapter contains essential definitions and results that will be used extensively throughout the thesis. A theory of compound operators and compound differential equations in a Banach space X is developed.

The definitions and properties of multiplicative and additive compound operators in \mathbb{R}^n , usually called compound matrices, can be found in [3, 31, 33, 47, 67, 75–78, 91, 109, 110]. Applications of compound matrices to differential equations are discussed in [41, 52–54, 56, 58, 61, 73, 76–78, 82, 95]. Results on Hilbert spaces are discussed in [1, 7, 8, 94, 104]. In particular, when X is a Hilbert space, Temam [104] relates the evolution of oriented k-dimensional volumes to the kth multiplicative compound operator of the differential of the semigroup $\{S(t)\}_{t\geq 0}$ of a dynamical system.

This chapter introduces the exterior product of vector spaces and additive and multiplicative compound operators. Compound differential equations associated with linear differential equations in a Banach space X are discussed with examples.

1.1 Exterior Products

Let X, Y be vector spaces over \mathbb{R} . Let $\mathscr{L}(X, Y)$ denote the space of all linear functionals from X to Y and $X^* = \mathscr{L}(X, \mathbb{R})$ denote the (algebraic) dual space of X. Define a nondegenerate bilinear map $\langle \cdot, \cdot \rangle : X^* \times X \to \mathbb{R}$ by

$$\langle v, u \rangle = v(u) \tag{1.1}$$

where $v \in X^*$ and $u \in X$.

Definition 1.1. For every $k \in \mathbb{N}$ and elements $u^1, \dots, u^k \in X$, define the *kth exterior product* $u^1 \wedge \dots \wedge u^k$ by

$$u^{1} \wedge \dots \wedge u^{k} : (X^{*})^{k} \longrightarrow \mathbb{R}$$
$$u^{1} \wedge \dots \wedge u^{k}(v_{1}, \dots, v_{k}) := \det[v_{i}(u^{j})], \quad v_{i} \in X^{*}, \ i = 1, \dots, k.$$
(1.2)

Definition 1.2. For each $k \in \mathbb{N}$, the *kth exterior power* of X, $\bigwedge^k X$, is the vector space of all finite linear combinations of elements $u^1 \wedge u^2 \wedge \cdots \wedge u^k$ where $u_i \in X$, $i = 1, \dots, k$, whose vector addition and scalar multiplication are defined by

$$(u^{1} \wedge u^{2} \wedge \dots \wedge u^{k} + w^{1} \wedge w^{2} \wedge \dots \wedge w^{k})(v_{1}, v_{2}, \dots, v_{k})$$

= $u^{1} \wedge u^{2} \wedge \dots \wedge u^{k}(v_{1}, v_{2}, \dots, v_{k}) + w^{1} \wedge w^{2} \wedge \dots \wedge w^{k}(v_{1}, v_{2}, \dots, v_{k}),$
 $\alpha(u^{1} \wedge u^{2} \wedge \dots \wedge u^{k}) = (\alpha u^{1}) \wedge u^{2} \wedge \dots \wedge u^{k} = \dots = u^{1} \wedge u^{2} \wedge \dots \wedge (\alpha u^{k}),$

where $\alpha \in \mathbb{R}, u^i, w^i \in X, v_i \in X^*, i = 1, \cdots, k$.

Remark 1.1. Equation (1.2) also defines $v_1 \wedge \cdots \wedge v_k \in \bigwedge^k X^*$ as a multilinear map from X^k to \mathbb{R} and a nondegenerate bilinear map $\langle \cdot, \cdot \rangle : \bigwedge^k X^* \times \bigwedge^k X \to \mathbb{R}$ by

$$\langle v_1 \wedge \cdots \wedge v_k, u^1 \wedge \cdots \wedge u^k \rangle = u^1 \wedge \cdots \wedge u^k (v_1, \cdots, v_k).$$

In this thesis, both notations will be used.

Proposition 1.1. The vectors u^1, \dots, u^k are linearly dependent in X if and only if

 $u^1 \wedge u^2 \wedge \dots \wedge u^k = 0,$

that is, $u^1 \wedge \cdots \wedge u^k(v_1, \cdots, v_k) = 0$ for all $v_1, \cdots, v_k \in X^*$.

Proof. It is straightforward that $u^1 \wedge \cdots \wedge u^k = 0$ if u^1, \cdots, u^k are linearly dependent in X. Mathematical induction is used to prove that $u^1 \wedge u^2 \wedge \cdots \wedge u^k = 0$ implies the linear dependence of u^1, \cdots, u^k . First, it is true if k = 1. Assume that $u^1 \wedge \cdots \wedge u^k = 0$ implies that u^1, \cdots, u^k is linearly dependent in X. Next, suppose that $u^1 \wedge \cdots \wedge u^k \wedge u^{k+1} = 0$. Then for every $v_1, \cdots, v_{k+1} \in X^*$,

$$u^1 \wedge \cdots \wedge u^{k+1}(v_1, \cdots, v_{k+1}) = 0.$$

Choose v_1 such that $v_1(u^1) \neq 0$. Let $\alpha_i = -\frac{v_1(u^i)}{v_1(u^1)}, 2 \leq i \leq k+1$. Then

which implies that

$$(u^2 + \alpha_2 u^1) \wedge \dots \wedge (u^{k+1} + \alpha_{k+1} u^1)(v_2, \dots, v_{k+1}) = 0$$

for all v_2, \dots, v_{k+1} since $v_1(u^1) \neq 0$. Therefore

$$(u^2 + \alpha_2 u^1) \wedge \cdots \wedge (u^{k+1} + \alpha_{k+1} u^1) = 0,$$

and the induction hypothesis implies that $u^2 + \alpha_2 u^1, \dots, u^{k+1} + \alpha_{k+1} u^1$ are linearly dependent. Hence u^1, \dots, u^{k+1} are linearly dependent. Mathematical induction implies that u^1, \dots, u^k are linearly dependent in X if $u^1 \wedge \dots \wedge u^k = 0$.

Let X be a normed space and X' denote the continuous dual of X, consisting of the continuous linear functionals on X. In general different topologies and norms can be imposed on $\bigwedge^k X$ depending on the application. For $w \in \bigwedge^k X$, define

$$\|w\|_{\bigwedge^{k} X} := \sup_{v_{i}} w(v_{1}, \cdots, v_{k}) = \sup_{v_{i}} \langle v_{1} \wedge \cdots \wedge v_{k}, w \rangle.$$
(1.3)

Here the supremum is taken over $v_i \in X', \|v_i\|_{X'} \leq 1, i = 1, \dots, k$. For simplicity, the symbol $\|\cdot\|$ will be used instead of $\|\cdot\|_{\bigwedge^k X}$ except when the relationship with the norm in X is to be emphasized. Then for $w, w_1, w_2 \in \bigwedge^k X, a \in \mathbb{R}$,

- (*i*) $||w|| \ge 0;$
- (ii) ||w|| = 0 if and only if for every $v_i \in X'$, $w(v_1, v_2, \dots, v_k) = 0$ if and only if w = 0;

$$(iii) ||aw|| = |a|||w||;$$

$$\begin{aligned} (iv) \quad \|w_1 + w_2\| &= \sup_{v_i} \left(w_1(v_1, v_2, \cdots, v_k) + w_2(v_1, v_2, \cdots, v_k) \right) \\ &\leq \sup_{v_i} w_1(v_1, v_2, \cdots, v_k) + \sup_{v_i} w(v_1, v_2, \cdots, v_k) \\ &\leq \|w_1\| + \|w_2\|. \end{aligned}$$

Therefore, $\|\cdot\|$ is a norm on $\bigwedge^k X$. The symbol $\bigwedge^k X$ is used to denote the completion of $\bigwedge^k X$ in this norm.

Proposition 1.2. If X is a normed space, then for every $u^i \in X$, $i = 1, \dots, k$, the norm on $\bigwedge^k X$ defined by (1.3) satisfies

$$||u^1 \wedge \dots \wedge u^k|| \le 2^{\frac{k(k-1)}{2}} ||u^1|| \dots ||u^k||.$$
 (1.4)

Proof. The proof is by mathematical induction. First, equality is satisfied in (1.4) when k = 1. Assume that (1.4) holds for k, a positive integer. Then if $u^i \in X, v_i \in X', i = 1, \dots, k+1$ and $v_1 \wedge \dots \wedge \hat{v}_l \wedge \dots \wedge v_{k+1}$ denotes the exterior product of the k vectors $\{v_i : i = 1, \dots, k+1, i \neq l\}$, then

$$\langle v_1 \wedge \dots \wedge v_{k+1}, u^1 \wedge \dots \wedge u^{k+1} \rangle$$

$$= \sum_{l=1}^{k+1} (-1)^{l+1} \langle v_l, u^1 \rangle \langle v_1 \wedge \dots \wedge \hat{v}_l \wedge \dots \wedge v_{k+1}, u^2 \wedge \dots \wedge u^{k+1} \rangle$$

$$\leq (k+1) \cdot \|u^1\| \cdot 2^{\frac{k(k-1)}{2}} \|u^2\| \cdots \|u^{k+1}\|$$

$$\leq 2^k \cdot 2^{\frac{k(k+1)}{2}} \|u^1\| \|u^2\| \cdots \|u^{k+1}\|$$

$$= 2^{\frac{k(k+1)}{2}} \|u^1\| \|u^2\| \cdots \|u^{k+1}\|.$$

Thus

$$||u^1 \wedge \dots \wedge u^{k+1}|| \le 2^{\frac{k(k+1)}{2}} ||u^1|| \dots ||u^{k+1}||,$$

and mathematical induction implies that (1.4) holds for all k.

1.1.1 Vector Space: Hamel Basis

Every real vector space X has a Hamel basis $H \subset X$ such that each $u \in X$ is a finite real linear combination of elements $h \in H$, a sum of the form

$$u = \sum_{h \in H} u_h h,$$

where all but a finite number of $u_h = 0$. With each $h \in H$ associate a linear functional $\langle h, \cdot \rangle : X \to \mathbb{R}$ defined for k in H by

$$\langle h,k\rangle = \left\{ egin{array}{cc} 1 & \mbox{if } k=h \\ 0 & \mbox{if } k
eq h \end{array}
ight.$$

and, for any $u \in X$, by linearity so that $\langle h, u \rangle = u_h$. Thus

$$u = \sum_{h \in H} \langle h, u \rangle h. \tag{1.5}$$

Finally, for any $u \in X$, a linear functional $\langle u, \cdot \rangle : X \to \mathbb{R}$ is defined by linearity $\langle u, \cdot \rangle = \sum_{h \in H} \langle h, u \rangle \langle h, \cdot \rangle$. It follows that, when H is a Hamel basis of $X, u^1 \wedge \cdots \wedge u^k$ can be expressed as a finite sum of the form

$$u^{1} \wedge \dots \wedge u^{k} = \sum_{h^{i} \in H, \ i=1,\dots,k} \left\langle h^{1}, u^{1} \right\rangle \left\langle h^{2}, u^{2} \right\rangle \dots \left\langle h^{k}, u^{k} \right\rangle h^{1} \wedge \dots \wedge h^{k}.$$
(1.6)

However, if k > 1, $\{h^1 \land \cdots \land h^k : h^i \in H, i = 1, \cdots, k\}$ is not a Hamel basis for $\bigwedge^k X$. It is not a linearly independent set as

$$h^1 \wedge \dots \wedge h^k = \operatorname{sgn}\sigma \ h^{i_1} \wedge \dots \wedge h^{i_k}$$
 (1.7)

if $\sigma = (i_1, \dots, i_k)$ is any permutation of $(1, \dots, k)$. A subset H_k obtained by including just one element $h^1 \wedge \dots \wedge h^k$ of each set of k! elements related by (1.7) is a Hamel basis for $\bigwedge^k X$ and, from (1.6),

$$u^{1} \wedge \cdots \wedge u^{k} = \sum_{h^{1} \wedge \cdots \wedge h^{k} \in H_{k}} \left\langle h^{1} \wedge \cdots \wedge h^{k}, u^{1} \wedge \cdots \wedge u^{k} \right\rangle h^{1} \wedge \cdots \wedge h^{k},$$

where $\langle h^1 \wedge \cdots \wedge h^k, u^1 \wedge \cdots \wedge u^k \rangle = \det \langle h^i, u^j \rangle$.

1.1.2 Normed Vector Space: Schauder Basis

Suppose that $S = \{e^i : i = 1, 2, \dots\}$ is a Schauder (countable) basis of a real normed vector space X, then

$$u = \sum_{i} u_i e^i,$$

where $u_i = \langle e^i, u \rangle$ is the *i*th coordinate of *u* with respect to this basis. Here $\langle e^i, \cdot \rangle$ is defined as for (1.5) but now the sum (1.5) may contain infinitely many

non-zero terms when X is not finite dimensional. Analogously to (1.7), the set

$$S_k = \left\{ e^{i_1} \wedge \dots \wedge e^{i_k} : 1 \le i_1 < \dots < i_k \right\}$$

is a basis for $\bigwedge^k X$ under the norm $\|\cdot\|$ defined in (1.3). No particular order is imposed on S_k . By Definition 1.2, if $u^j \in X$, $j = 1, \dots, k$, $u^j = \sum u_i^j e^i$, $u_i^j = \langle e^i, u^j \rangle$, then

$$u^1 \wedge \cdots \wedge u^k = \sum_{(i)} u_{(i)} e^{i_1} \wedge \cdots \wedge e^{i_k},$$

where

$$\begin{array}{lll} u_{(i)} &=& \left\langle e^{i_1} \wedge \dots \wedge e^{i_k}, u^1 \wedge \dots \wedge u^k \right\rangle \\ &=& \det \left[\left\langle e^{i_r}, u^j \right\rangle \right], \quad r, j = 1, \cdots, k \\ &=& \det \left[u^j_{i_r} \right], \quad r, j = 1, \cdots, k \\ &=& u^{1 \cdots k}_{i_1 \cdots i_k}. \end{array}$$

Thus the $(i) = (i_1 \cdots i_k)$ -th coordinate $u_{(i)} = u_{i_1 \cdots i_k}^{1 \cdots k}$ of $u^1 \wedge \cdots \wedge u^k$, $1 \leq i_1 < \cdots < i_k$, with respect to the basis $\{e^{i_1} \wedge \cdots \wedge e^{i_k}\}$ in $\bigwedge^k X$ is the $k \times k$ minor determined by the rows i_1, \cdots, i_k of the matrix $[u_i^j]$.

If X is a Hilbert space with inner product $\langle \cdot, \cdot \rangle$, then $\bigwedge^k X$ is also a Hilbert space with inner product

$$\langle u^1 \wedge \dots \wedge u^k, v^1 \wedge \dots \wedge v^k \rangle = \det \left[\langle u^i, v^j \rangle \right].$$
 (1.8)

The space $\hat{\bigwedge}^k X$ is the completion of $\bigwedge^k X$ under the associated norm $\langle w_1, w_2 \rangle^{\frac{1}{2}}$. If X is separable and $\{e^i\}$ is an orthonormal basis, then $\{e^{i_1} \wedge \cdots \wedge e^{i_k} : i_1 < \cdots < i_k\}$ forms an orthonormal basis of $\hat{\bigwedge}^k X$. It follows from

$$\left\langle e^{i_1} \wedge \dots \wedge e^{i_k}, e^{j_1} \wedge \dots \wedge e^{j_k} \right\rangle = \delta^{(i)}_{(j)}$$

that

$$||u^{1} \wedge \dots \wedge u^{k}||^{2} = \langle u^{1} \wedge \dots \wedge u^{k}, u^{1} \wedge \dots \wedge u^{k} \rangle$$
$$= \sum_{i_{1} < \dots < i_{k}} \langle e^{i_{1}} \wedge \dots \wedge e^{i_{k}}, u^{1} \wedge \dots \wedge u^{k} \rangle^{2}$$
$$= \sum_{i_{1} < \dots < i_{k}} (u^{1 \dots k}_{i_{1} \dots i_{k}})^{2}.$$

Remark 1.2. A good reference on the exterior product of a Hilbert space is Temam [104], Chapter V.

1.1.3 Function Space

Let X be a vector space of real-valued functions on a set Ω . Then a linear functional $x \in X^*$ may be associated with $x \in \Omega$ by

$$\langle x, u \rangle = u(x)$$

if $u \in X$. For $u^1, \dots, u^k \in X$ and $x_1, \dots, x_k \in \Omega$,

$$\langle x_1 \wedge \cdots \wedge x_k, u^1 \wedge \cdots \wedge u^k \rangle = u^1 \wedge \cdots \wedge u^k(x_1, \cdots, x_k) = \det \left[u^j(x_i) \right].$$

Norms on $\bigwedge^k X$ may be defined by

$$||w||_{\infty} := \operatorname{ess} \sup \{|w(x_1, \cdots, x_k)| : x_i \in \Omega\}$$
(1.9)

or, if $1 \leq p < \infty$ and Ω is a measurable set by

$$||w||_{L^p} := \frac{1}{k!} \left[\int_{\Omega^k} |w(x_1, \cdots, x_k)|^p \, dx_1 \cdots dx_k \right]^{\frac{1}{p}}; \tag{1.10}$$

when these expressions are finite. If Ω is finite or countable, the integral (1.10) may be replaced by a discrete sum. For example, when $X = \mathbb{R}^n$ and $\Omega = \{1, 2, \dots, n\}$, for $u^i = (u_1^i, \dots, u_n^i), i = 1, \dots, k$,

$$\langle 1 \wedge \dots \wedge k, u^1 \wedge \dots \wedge u^k \rangle = u^1 \wedge \dots \wedge u^k (1, \dots, k) = \det \left[u_i^j \right]$$

and the l_p norm of $u^1 \wedge \cdots \wedge u^k$ on $\bigwedge^k \mathbb{R}^n$ is

$$\|u^{1} \wedge \dots \wedge u^{k}\|_{l_{p}} = \frac{1}{k!} \left(\sum_{i_{1}, \dots, i_{k}} |u^{1 \dots k}_{i_{1} \dots i_{k}}|^{p} \right)^{\frac{1}{p}} = \left(\sum_{1 \le i_{1} < \dots < i_{k}} |u^{1 \dots k}_{i_{1} \dots i_{k}}|^{p} \right)^{\frac{1}{p}}$$

where $u_{i_1\cdots i_k}^{1\cdots k}$ is the $k \times k$ minor of the $n \times k$ matrix (u^1, \cdots, u^k) determined by the rows i_1, \cdots, i_k .

1.1.4 $L^2(\Omega)$

The choice of v plays an important role in the application of exterior products. The following example on $X = L^2(\Omega)$, where Ω is a measurable set in \mathbb{R}^n , shows different representations of $u^1 \wedge u^2$ associated with different choices of v. Case 1: A pointwise representation of exterior products

For $u \in L^2(\Omega), x \in \Omega$, let x again denote the linear functional $u \mapsto \langle x, u \rangle := u(x)$. Define an inner product

$$\langle u,v\rangle = \int_{\Omega} u(x) v(x) dx.$$

For $u^1, u^2 \in X$ and $x_1, x_2 \in \Omega$,

$$u^1 \wedge u^2(x_1, x_2) = \det \left[u^i(x_j) \right].$$

Then, from (1.8),

$$\langle u^{1} \wedge u^{2}, v^{1} \wedge v^{2} \rangle = \det \begin{bmatrix} \langle u^{1}, v^{1} \rangle & \langle u^{1}, v^{2} \rangle \\ \langle u^{2}, v^{1} \rangle & \langle u^{2}, v^{2} \rangle \end{bmatrix} = \det \begin{bmatrix} \int_{\Omega} u^{i} v^{j} \end{bmatrix}$$
(1.11)

is an inner product on $\bigwedge^2 L^2(\Omega)$. Since $u^1 \wedge u^2(x_1, x_2) = \det [u^i(x_j)]$,

$$(u^{1} \wedge u^{2}, v^{1} \wedge v^{2}) = \int_{\Omega^{2}} \det \left[u^{i}(x_{j}) \right] \det \left[v^{i}(x_{j}) \right] dx_{1} dx_{2}$$
 (1.12)

is also an inner product on $\bigwedge^k L^2(\Omega)$. In fact, the inner products in (1.11) and (1.12) are equivalent since

$$\langle u^{1} \wedge u^{2}, v^{1} \wedge v^{2} \rangle = \frac{1}{2!} \left(u^{1} \wedge u^{2}, v^{1} \wedge v^{2} \right).$$
 (1.13)

This can be seen by applying the Binet-Cauchy identity for square matrices $[u], [v], \det[u] \det[v] = \det[uv]$, to the integrand in (1.12).

Case 2: A basis representation of exterior products

Suppose that $\{e^i\}$ is an orthonormal basis of $L^2(\Omega)$. Then $u = \sum_i \langle e^i, u \rangle e^i$. Define another linear map $e^i : u \mapsto \langle e^i, u \rangle$. For $u^1, u^2 \in X$, $u^1 \wedge u^2$ is a bilinear map from $\{e^i\} \times \{e^i\}$ to \mathbb{R} and

$$u^{1} \wedge u^{2}(e^{i_{1}}, e^{i_{2}}) = \det \begin{bmatrix} \langle e^{i_{1}}, u^{1} \rangle & \langle e^{i_{1}}, u^{2} \rangle \\ \langle e^{i_{2}}, u^{1} \rangle & \langle e^{i_{2}}, u^{2} \rangle \end{bmatrix}.$$

The set $\{e^{i_1} \wedge e^{i_2} : i_1 < i_2\}$ forms an orthonormal basis of $\bigwedge^2 L^2(\Omega)$ and if $u^j = \sum_i \langle e^i, u^j \rangle e^i, j = 1, 2$, then

$$u^1\wedge u^2=\sum_{(i)}u_{(i)}e^{i_1}\wedge e^{i_2}$$

where

$$u_{(i)} = \langle e^{i_1} \wedge e^{i_2}, u^1 \wedge u^2 \rangle$$

= det [$\langle e^{i_r}, u^p \rangle$], $r, j = 1, 2$
= det [$u^p_{i_r}$], $r, j = 1, 2$
= $u^{12}_{i_{j_1}}$ (1.14)

is the $(i) = (i_1i_2)$ -th coordinate of $u^1 \wedge u^2$, $i_1 < i_2$, with respect to the basis $\{e^{i_1} \wedge e^{i_2}\}$ in $\bigwedge^2 L^2(\Omega)$, which is the 2 × 2 minor determined by the rows i_1, i_2 of the matrix $[u_i^j]$ (see Section 1.1.2). It follows from (1.13) that

$$\|u^1 \wedge u^2\|^2 = \sum_{i_1 < i_2} \left(u_{i_1 i_2}^{12}\right)^2 = \frac{1}{2!} \int_{\Omega^2} \det \left[u^i(x_j)\right]^2 dx_1 dx_2,$$

where $u_{i_1i_2}^{12}$ is defined by (1.14) for any basis orthonormal with respect to the inner product $\langle \cdot, \cdot \rangle$ on $L^2(\Omega)$.

1.1.5 Interpretation

In keeping with the tradition of representing a vector u as a directed line segment, the exterior product $u^1 \wedge \cdots \wedge u^k \in \bigwedge^k X$ can be interpreted as the oriented parallelepiped determined by the ordered set of vectors $\{u^1, \cdots, u^k\}$ (see Figure 1.1). Just as ||u|| can be considered a measure of the length of u



Figure 1.1: Exterior product $u^1 \wedge \cdots \wedge u^k$

when it is represented as directed line segment, a norm $||u^1 \wedge \cdots \wedge u^k||$ may be considered a measure of the k-volume of the oriented parallelepiped. If u is referred to a basis $\{e^i\}$, then $u = \sum_i u_i e^i$ and the coordinate $u_i = \langle e^i, u \rangle$ may be considered the projection of the directed line segment onto the coordinate axis e^i . With respect to the basis $\{e^{i_1} \wedge \cdots \wedge e^{i_k}\}$ of $\bigwedge^k X$, $u^1 \wedge \cdots \wedge u^k = \sum_{\substack{(i) \\ (i)}} u_{(i)}e^{i_1} \wedge \cdots \wedge e^{i_k}$, where $u_{(i)} = \langle e^{i_1} \wedge \cdots \wedge e^{i_k}, u^1 \wedge \cdots \wedge u^k \rangle = u_{i_1 \cdots i_k}^{1 \cdots k}$ may be considered the projection of $u^1 \wedge \cdots \wedge u^k$ onto the k-dimensional subspace of X spanned by e^{i_1}, \cdots, e^{i_k} . For example, when $X = \mathbb{R}^n$ with the standard orthonormal basis e^1, \cdots, e^n , then $\bigwedge^k \mathbb{R}^n \simeq \mathbb{R}^{\binom{k}{n}}$ and the $\binom{n}{k}$ components of $u^1 \wedge \cdots \wedge u^k$ with respect to the k-dimensional coordinate subspace span $\{e^{i_1}, \cdots, e^{i_k}\}$ are the determinants $u_{i_1 \cdots i_k}^{1 \cdots i_k}$.

1.2 Compound Operators

The definitions of compound operators on vector spaces are generalized from compound matrices on \mathbb{R}^n , (see [3, 33, 47, 67, 109]) and the two multilinear operators on Hilbert spaces considered in Temam [104], Chapter V.

Definition 1.3. Let X and Y be vector spaces and $A: X \to Y$ be a linear operator. The operator $A^{(k)}: \bigwedge^k X \to \bigwedge^k Y$ defined by

$$A^{(k)}\left(u^{1}\wedge\cdots\wedge u^{k}\right):=Au^{1}\wedge\cdots\wedge Au^{k}, \quad u^{i}\in X$$
(1.15)

and extended by linearity to $\bigwedge^k X$ is called the *kth multiplicative compound* (or *exterior power*) of A.

Definition 1.4. Let X and Y be vector spaces, $X \subset Y$ and $A : X \to Y$ be a linear operator. The operator $A^{[k]} : \bigwedge^k X \to \bigwedge^k Y$ defined by

$$A^{[k]}\left(u^{1}\wedge\cdots\wedge u^{k}\right) := \sum_{j=1}^{k} u^{1}\wedge\cdots\wedge Au^{j}\wedge\cdots\wedge u^{k}, \quad u^{i}\in X$$
(1.16)

and extended by linearity to $\bigwedge^k X$ is called the *kth additive compound* of A.

Remark 1.3. In [104], Temam uses $\bigwedge^k A$ and A_k to denote what are, in this thesis, called the kth multiplicative compound operator and kth additive compound operator of A.

The multiplicative and additive compound operators have the following properties.

Proposition 1.3.

(i) If $A : X \to Y$ and $B : Y \to Z$ are linear operators, then $(AB)^{(k)}$ satisfies the Binet-Cauchy identity

$$(AB)^{(k)} = A^{(k)}B^{(k)}.$$

(ii) If $X \subset Y$ and $A, B : X \to Y$ are linear operators, then $(A + B)^{[k]}$ satisfies the additive identity

$$(A+B)^{[k]} = A^{[k]} + B^{[k]}.$$

- (iii) If $X \subset Y$ and $A : X \to Y$ is a linear operator, then $A^{[k]} = \frac{d}{dt} (I + tA)^{(k)} \Big|_{t=0} = \lim_{h \to 0} \left[(I + hA)^{(k)} I^{(k)} \right] / h$.
- (iv) If $\lambda_1, \lambda_2, \cdots$ and e^1, e^2, \cdots are eigenvalues and eigenvectors of A, then $e^{i_1} \wedge \cdots \wedge e^{i_k}, 1 \leq i_1 < i_2 < \cdots < i_k$ are eigenvectors of $A^{(k)}$ and $A^{[k]}$ with corresponding eigenvalues $\lambda_{i_1} \cdots \lambda_{i_k}$ and $\lambda_{i_1} + \cdots + \lambda_{i_k}$, respectively.

Proof. (i) Let
$$u^i \in X$$
, $i = 1, \dots, k$. Then
 $(AB)^{(k)} (u^1 \wedge \dots \wedge u^k) = ABu^1 \wedge \dots \wedge ABu^k$
 $= A^{(k)} (Bu^1 \wedge \dots \wedge Bu^k)$
 $= A^{(k)} B^{(k)} (u^1 \wedge \dots \wedge u^k)$

and this equality can be extended by linearity to $\bigwedge^k X$. Thus the Binet-Cauchy identity is proved.

(*ii*) Let $u^i \in X$, $i = 1, \dots, k$. Then

$$(A+B)^{[k]} (u^{1} \wedge \dots \wedge u^{k})$$

$$= \sum_{j=1}^{k} u^{1} \wedge \dots \wedge (A+B) u^{j} \wedge \dots \wedge u^{k}$$

$$= \sum_{j=1}^{k} u^{1} \wedge \dots \wedge A u^{j} \wedge \dots \wedge u^{k} + \sum_{j=1}^{k} u^{1} \wedge \dots \wedge B u^{j} \wedge \dots \wedge u^{k}$$

$$= (A^{[k]} + B^{[k]}) (u^{1} \wedge \dots \wedge u^{k})$$

and this equality can be extended by linearity to $\bigwedge^k X$. Thus the additive identity is proved.

(*iii*) Let $u^i \in X$, $i = 1, \dots, k$. Then

$$\begin{bmatrix} (I+hA)^{(k)} - I^{(k)} \end{bmatrix} (u^1 \wedge \dots \wedge u^k)$$

= $(I+hA)^{(k)} (u^1 \wedge \dots \wedge u^k) - u^1 \wedge \dots \wedge u^k$
= $(1+hA)u^1 \wedge \dots \wedge (1+hA)u^k - u^1 \wedge \dots \wedge u^k$
= $(hAu^1) \wedge u^2 \dots \wedge u^k + \dots + u^1 \wedge u^2 \dots \wedge (hAu^k) + O(h^2)$
= $hA^{[k]} (u^1 \wedge u^2 \dots \wedge u^k) + o(h^2)$

and thus

$$\lim_{h \to 0} \left[\left(I + hA \right)^{(k)} - I^{(k)} \right] \left(u^1 \wedge u^2 \cdots \wedge u^k \right) \middle/ h = A^{[k]} \left(u^1 \wedge u^2 \cdots \wedge u^k \right).$$

This above equality can be extended by linearity to $\bigwedge^k X$. Therefore

$$A^{[k]} = \left. \frac{d}{dt} \left(I + tA \right)^{(k)} \right|_{t=0} = \lim_{h \to 0} \left[\left(I + hA \right)^{(k)} - I^{(k)} \right] \right/ h \; .$$

(iv) The eigenvalues and eigenfunctions properties follow from

$$\begin{aligned} A^{(k)}(e^{i_1} \wedge \dots \wedge e^{i_k}) &= A e^{i_1} \wedge \dots \wedge A e^{i_k} \\ &= \lambda_{i_1} e^{i_1} \wedge \dots \wedge \lambda_{i_k} e^{i_k} \\ &= \lambda_{i_1} \cdots \lambda_{i_k} (e^{i_1} \wedge \dots \wedge e^{i_k}) \end{aligned}$$

 and

$$\begin{aligned} A^{[k]}(e^{i_1} \wedge \dots \wedge e^{i_k}) &= \sum_{j=1}^k e^{i_1} \wedge \dots \wedge A e^{i_j} \wedge \dots \wedge e^{i_k} \\ &= \sum_{j=1}^k e^{i_1} \wedge \dots \wedge \lambda_{i_j} e^{i_j} \wedge \dots \wedge e^{i_k} \\ &= (\lambda_{i_1} + \dots + \lambda_{i_k})(e^{i_1} \wedge \dots \wedge e^{i_k}). \end{aligned}$$

1.2.1 Representation of Compound Operators

Let X and Y be real normed vector spaces with Schauder bases $\{e^j\}$ and $\{f^i\}$ respectively. If $A: u \mapsto v = Lu$ is a bounded linear function from X to Y, then

$$\begin{array}{rcl} u & = & \sum_{j} u_{j} e^{j}, & u_{j} = \langle e^{j}, u \rangle , \\ L u & = & \sum_{i} v_{i} f^{i}, & v_{i} = \langle f^{i}, L u \rangle \end{array}$$

implies that

$$v_i = \sum_j a_i^j u_j, \text{ where } a_i^j = \left\langle f^i, Le^j \right\rangle.$$
(1.17)

Thus the relationship between the sequences of basis coefficients or coordinates, $\hat{u} = (u_i)$ and $\hat{v} = (v_i)$ of u and v = Lu respectively, arrayed as column vectors, satisfy (1.17) which may be written

$$\widehat{v}=A\widehat{u}$$

where $A = \begin{bmatrix} a_i^j \end{bmatrix}$ is a matrix whose rows are indexed by *i* and columns by *j*. By Definition 1.2, if $u^r \in X$, $r = 1, \dots, k$, $u^r = \sum_j u_j^r e^j$, $u_j^r =$

 $\langle e^j, u^r \rangle$, implies

$$u^1 \wedge \dots \wedge u^k = \sum_{(j)} u_{(j)} e^{j_1} \wedge \dots \wedge e^{j_k}$$

where

$$u_{(j)} = \left\langle e^{j_1} \wedge \dots \wedge e^{j_k}, u^1 \wedge \dots \wedge u^k \right\rangle = u_{j_1 \cdots j_k}^{1 \cdots k}$$
(1.18)

is the $(j) = (j_1 \cdots j_k)$ -th coordinate of $u^1 \wedge \cdots \wedge u^k$, $j_1 < \cdots < j_k$, with respect to the basis $\{e^{j_1} \wedge \cdots \wedge e^{j_k}\}$ in $\bigwedge^k X$ (see Section 1.1.2).

From (1.17), $v^r = Lu^r = \sum_i v_i^r f^i$, $v_j^r = \langle f^i, Lu^r \rangle$ implies, since

$$L^{(k)}\left(u^{1}\wedge\cdots\wedge u^{k}\right)=\left(Lu^{1}\right)\wedge\cdots\wedge\left(Lu^{k}\right)$$

the $(i) = (i_1 \cdots i_k)$ -th coordinate $v_{(i)}$ of $L^{(k)} (u^1 \wedge \cdots \wedge u^k)$, $i_1 < \cdots < i_k$, with respect to the basis $\{f^{i_1} \land \cdots \land f^{i_k}\}$ in $\bigwedge {}^k Y$ is given by

$$L^{(k)}\left(u^{1}\wedge\cdots\wedge u^{k}\right)=\sum_{(i)}v_{(i)}f^{i_{1}}\wedge\cdots\wedge f^{i_{k}}$$

where, from (1.17) and (1.18) with $(i) = (i_1 \cdots i_k), (j) = (j_1 \cdots j_k),$

$$\begin{aligned} v_{(i)} &= v_{i_1 \cdots i_k}^{1 \cdots k} = \sum_{(j)} b_{(i)}^{(j)} u_{j_1 \cdots j_k}^{1 \cdots k}, \\ b_{(i)}^{(j)} &= \left\langle f^{i_1} \wedge \cdots \wedge f^{i_k}, L^{(k)} \left(e^{j_1} \wedge \cdots \wedge e^{j_k} \right) \right\rangle \\ &= \left\langle f^{i_1} \wedge \cdots \wedge f^{i_k}, \left(Le^{j_1} \right) \wedge \cdots \wedge \left(Le^{j_k} \right) \right\rangle \\ &= \det \left[\left\langle f^{i_r}, Le^{j_s} \right\rangle \right], \ r, s = 1, \cdots, k \\ &= a_{i_1 \cdots i_k}^{j_1 \cdots j_k} \end{aligned}$$

where $a_{i_1\cdots i_k}^{j_1\cdots j_k}$ denotes the $k \times k$ minor of the matrix $A = [a_i^j]$ determined by the rows i_1, \cdots, i_k and the columns j_1, \cdots, j_k . Denote by

$$A^{(k)} = \begin{bmatrix} b_{(i)}^{(j)} \end{bmatrix} = \begin{bmatrix} a_{i_1 \cdots i_k}^{j_1 \cdots j_k} \end{bmatrix},$$
(1.19)

the matrix of the linear function $L^{(k)} : \bigwedge^k X \to \bigwedge^k Y$ referred to the bases $\{e^{j_1} \land \cdots \land e^{j_k}\}, \{f^{i_1} \land \cdots \land f^{i_k}\}.$

Similarly, if $X \subset Y$, the $(i) = (i_1 \cdots i_k)$ -th coordinate $w_{(i)}$ of $L^{[k]}(u^1 \wedge \cdots \wedge u^k)$, $i_1 < \cdots < i_k$, with respect to the basis $\{e^{i_1} \wedge \cdots \wedge e^{i_k}\}$ in $\bigwedge^k Y$ is given by

$$L^{[k]}\left(u^1\wedge\cdots\wedge u^k
ight)=\sum_{(i)}w_{(i)}e^{i_1}\wedge\cdots\wedge e^{i_k}$$

where, from (1.17) and (1.18) with $(i) = (i_1 \cdots i_k), (j) = (j_1 \cdots j_k),$

$$\begin{split} w_{(i)} &= \sum_{(j)} c_{(i)}^{(j)} u_{j_1 \cdots j_k}^{1 \cdots k}, \\ c_{(i)}^{(j)} &= \left\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, L^{[k]} \left(e^{j_1} \wedge \cdots \wedge e^{j_k} \right) \right\rangle \\ &= \left\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, \sum_{s=1}^k e^{j_1} \wedge \cdots \wedge L e^{j_s} \wedge \cdots \wedge e^{j_k} \right\rangle \\ &= \sum_{s=1}^k \left\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, e^{j_1} \wedge \cdots \wedge L e^{j_s} \wedge \cdots \wedge e^{j_k} \right\rangle. \end{split}$$

Let

$$\delta^{j_1\cdots\widehat{j_\theta}\cdots j_k}_{i_1\cdots\widehat{i_t}\cdots i_k} = \begin{cases} 1, & \text{if } i_1\cdots\widehat{i_t}\cdots i_k = j_1\cdots\widehat{j_s}\cdots j_k \\ 0, & \text{if } i_1\cdots\widehat{i_t}\cdots i_k \neq j_1\cdots\widehat{j_s}\cdots j_k \end{cases}$$

where $i_1 \cdots \hat{i_t} \cdots i_k$ denotes the k-1 numbers $i_1 \cdots i_{t-1} i_{t+1} \cdots i_k$. Since

$$\langle e^{i_1} \wedge \dots \wedge e^{i_k}, e^{j_1} \wedge \dots \wedge Le^{j_s} \wedge \dots \wedge e^{j_k} \rangle$$

$$= \sum_t (-1)^{s+t} \langle e^{i_t}, Le^{j_s} \rangle \langle e^{i_1} \wedge \dots \wedge \widehat{e^{i_t}} \wedge \dots \wedge e^{i_k}, e^{j_1} \wedge \dots \wedge \widehat{e^{j_s}} \wedge \dots \wedge e^{j_k} \rangle$$

$$= \sum_t (-1)^{s+t} a^{j_s}_{i_t} \delta^{j_1 \dots \widehat{j_s} \dots j_k}_{i_1 \dots \widehat{i_t} \dots i_k},$$

the matrix of $L^{[k]}$ is $A^{[k]} = \left[c^{(j)}_{(i)} \right]$, where

$$c_{(i)}^{(j)} = \begin{cases} a_{i_1}^{i_1} + \dots + a_{i_k}^{i_k}, & \text{if } (i) = (j); \\ (-1)^{s+t} a_{i_t}^{j_s}, & \text{if exactly one entry } i_t \text{ of } (i) \text{ does not occur in } \\ (j) \text{ and } j_s \text{ does not occur in } (i); \\ 0, & \text{if } (i) \text{ differs from } (j) \text{ in two or more entries.} \end{cases}$$

$$(1.20)$$

Example 1.1. When $X = \mathbb{R}^3$, $\bigwedge^k \mathbb{R}^3 = \operatorname{span} \{ u^1 \wedge \cdots \wedge u^k : u^i \in \mathbb{R}^3 \} \simeq \mathbb{R}^{\binom{3}{k}}$. Let $A : \mathbb{R}^3 \to \mathbb{R}^3$ be a linear operator, which has a matrix representation $A = [a_i^j]_{3\times 3}$. Then

$$A^{(1)} = \begin{bmatrix} a_1^1 & a_1^2 & a_1^3 \\ a_2^1 & a_2^2 & a_2^3 \\ a_3^1 & a_3^2 & a_3^3 \end{bmatrix} = A, \ A^{(2)} = \begin{bmatrix} a_{12}^{12} & a_{13}^{12} & a_{12}^{23} \\ a_{13}^{12} & a_{13}^{13} & a_{13}^{23} \\ a_{23}^{12} & a_{23}^{13} & a_{23}^{23} \end{bmatrix}, \ A^{(3)} = a_{123}^{123} = \det A,$$

 $A^{[1]} = A, \quad A^{[2]} = \begin{bmatrix} a_3^2 & a_1^1 + a_3^3 & a_1^2 \\ -a_3^1 & a_2^1 & a_2^2 + a_3^3 \end{bmatrix}, \quad A^{[3]} = a_1^1 + a_2^2 + a_3^3 = \operatorname{tr} A.$

Example 1.2. Let $X = C(\Omega, \mathbb{R})$ where Ω is a measurable set in \mathbb{R}^n and $p \in C(\Omega \times \Omega \to \mathbb{R})$. Consider an integral operator $P: X \to X$,

$$(Pu)(x) = \int_{\Omega} p(x, y) u(y) \, dy.$$
 (1.21)

Let $(Qu)(x) = \int_{\Omega} q(x, y) u(y) dy$ and the multiplication of P and Q is defined by

$$(PQu)(x) = \int_{\Omega} pq(x, y) u(y) dy,$$

where $pq(x,y) = \int_{\Omega} p(x,s) q(s,y) ds$. Now the operator $P^{(k)} : \bigwedge^{k} X \to \bigwedge^{k} X$,

$$\left(P^{(k)}w\right)\left(x_1,\cdots,x_k\right) = \frac{1}{k!}\int_{\Omega^k} \det\left[p\left(x_i,y_j\right)\right]w\left(y_1,\cdots,y_k\right)dy_1\cdots dy_k,$$

satisfies the Binet-Cauchy identity $(PQ)^{(k)} = P^{(k)}Q^{(k)}$; and the operator $P^{[k]} : \bigwedge^k X \to \bigwedge^k X$,

$$(P^{[k]}w) (x_1, \cdots, x_k)$$

= $\int_{\Omega} [p(x_1, y) w(y, x_2, \cdots, x_k) + \cdots + p(x_k, y) w(x_1, \cdots, x_{k-1}, y)] dy,$

satisfies the additive identity $(P+Q)^{[k]} = P^{[k]} + Q^{[k]}$.

1. Compound Operators and Compound Equations

Example 1.3. Let X be a vector space of real-valued functions on a set Ω . Consider a linear operator $A: X \to X$ defined by

$$\langle x, Au \rangle = a(x)u(x), \quad x \in \Omega.$$
 (1.22)

Then, by Definition 1.3 and Definition 1.4, pointwise representations of the kth multiplicative and additive compounds of A satisfy

$$\langle x_1 \wedge \dots \wedge x_k, A^{(k)} w \rangle = \left(\prod_{i=1}^k a(x_i) \right) w (x_1, \dots, x_k)$$
$$\langle x_1 \wedge \dots \wedge x_k, A^{[k]} w \rangle = \sum_{i=1}^k a(x_i) w (x_1, \dots, x_k) ,$$

 and

۱

where $x_1, \dots, x_k \in \Omega$ and $w \in \bigwedge^k X$.

More generally, Example 1.3 can be included in Example 1.2 if singular kernels p(x, y) are permitted in (1.21).

In the case that X is a normed space with a basis $\{e^i\}$, this example falls under the discussion of bases at the beginning of this section. The basis representation of A is $a_j^i = \langle e^j, ae^i \rangle = a_i^j$ and thus the basis representations of $A^{(k)}$ and $A^{[k]}$ are given by (1.19) and (1.20).

1.2.2 The Laplacian Operator

Let $X = C^2(\Omega, \mathbb{R})$ and $Y = C(\Omega, \mathbb{R})$ where Ω is a bounded domain of \mathbb{R}^m . Then $X \subset Y$. By Definition 1.4, the *k*th multiplicative and additive compounds of the Laplacian $\Delta : X \to Y$, $\Delta^{(k)}$ and $\Delta^{[k]} : \bigwedge^k X \to \bigwedge^k Y$, are defined on wedges by

$$\Delta^{(k)}(u^1 \wedge \dots \wedge u^k) = (\Delta u^1) \wedge \dots \wedge (\Delta u^k),$$

$$\Delta^{[k]}(u^1 \wedge \dots \wedge u^k) = (\Delta u^1) \wedge \dots \wedge u^k + \dots + u^1 \wedge \dots \wedge (\Delta u^k)$$

and extended by linearity. For $u \in X$ and $x \in \Omega$, the pointwise representation of Δu is

$$\langle x, \Delta u \rangle = \Delta u(x),$$

the Laplacian of u with respect to x. For $w \in \bigwedge^k X$ and $x_1, \dots, x_k \in \Omega$, pointwise representations of $\Delta^{(k)}w$ and $\Delta^{[k]}w$ are

$$\langle x_1 \wedge \cdots \wedge x_k, \Delta^{(k)} w \rangle = \left(\prod_{i=1}^k \Delta_i\right) w (x_1, \cdots, x_k),$$

and

$$\left\langle x_1\wedge\cdots\wedge x_k,\Delta^{[k]}w
ight
angle =\sum_{i=1}^k\Delta_i w\left(x_1,\cdots,x_k
ight),$$

where $\Delta_i w = \langle x_i, \Delta w \rangle$ is the Laplacian of $w(x_1, \dots, x_k)$ with respect to x_i . Remark 1.4. It is emphasized that $\Delta^{(k)}$ and $\Delta^{[k]}$ are not considered as operators on $w(x_1, \dots, x_k)$ in $C^2(\Omega^k, \mathbb{R})$, but rather in $\bigwedge^k C^2(\Omega, \mathbb{R})$ and, in particular, $w(x_1, \dots, x_k)$ is antisymmetric in (x_1, \dots, x_k) .

In the more general context, $X = H^2(\Omega) = \{u \in L^2(\Omega) : \nabla u, \Delta u \in L^2(\Omega)\}$ is associated with some boundary conditions on a smooth boundary $\partial\Omega$ and $Y = L^2(\Omega); X \subset Y$. Let $\{e^i : i = 1, 2, \cdots\}$ be an orthonormal basis of X composed of eigenfunctions of the Laplacian Δ . Then the set of vectors $\{e^{i_1} \wedge \cdots \wedge e^{i_k} : 1 \leq i_1 < \cdots < i_k\}$ is an orthonormal basis of $\bigwedge^k X$ composed of eigenfunctions of $\Delta^{(k)}$ and $\Delta^{[k]}$. Since $\langle e^i, \Delta e^j \rangle = \lambda_j \delta_i^j$,

$$\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, \Delta^{(k)}(e^{j_1} \wedge \cdots \wedge e^{j_k}) \rangle = (\lambda_{j_1} \cdots \lambda_{j_k}) \, \delta^{j_1 \cdots j_k}_{i_1 \cdots i_k}$$

 and

$$\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, \Delta^{[k]}(e^{j_1} \wedge \cdots \wedge e^{j_k}) \rangle = (\lambda_{j_1} + \cdots + \lambda_{j_k}) \, \delta^{j_1 \cdots j_k}_{i_1 \cdots i_k} \,,$$

it follows that if $u \in X$, then a basis representation of Δu is

$$\langle e^i, \Delta u \rangle = \lambda_i u,$$

and if $w \in \bigwedge^k X$, then basis representations of $\Delta^{(k)}w$ and $\Delta^{[k]}w : \bigwedge^k X \to \bigwedge^k Y$ are

$$\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, \Delta^{(k)} w \rangle = (\lambda_{i_1} \cdots \lambda_{i_k}) w,$$

and

$$\langle e^{i_1} \wedge \dots \wedge e^{i_k}, \Delta^{[k]} w \rangle = (\lambda_{i_1} + \dots + \lambda_{i_k}) w.$$

1.3 Compound Differential Equations

This section deals with linear differential equations in a Banach space X.

Definition 1.5. A two-parameter family of linear operators $\{T(t,s) : 0 \le s \le t\} \subseteq \mathscr{L}(X,X)$ is said to be an *evolution operator* if

- (i) T(s,s) = I,
- (ii) T(t,s)T(s,r) = T(t,r) for $0 \le r \le s \le t$,
- (*iii*) $t \mapsto T(t, s)$ is continuous for $t \ge s$.

Let

$$\lim_{h \to 0} \frac{T(t+h,t)x - x}{h} = A(t)x,$$
(1.23)

the domain $\mathcal{D}(A(t))$ of A(t) being the set of all $x \in X$ for which the limit defined above exists. The map $A(t) : \mathcal{D}(A(t)) \subseteq X \to X$ is called the generator of T(t,s). Assume that $\mathcal{D}(A(t)) = \mathcal{D}(A)$, which is independent of t and dense in X.

If there exists a unique solution of the initial value problem

$$\frac{du}{dt} = A(t)u,$$

$$u(s) = u_s \in \mathcal{D}(A),$$
(1.24)

then there is an evolution operator given by the relation $u(t) = T(t, s)u_s$. The operators A(t) are usually unbounded. Detailed discussions of these topics may be found in [34, 42, 87, 111, 113].

Now assume that $u(t) = T(t, s)u_s, 0 \le s \le t$ solves the initial value problem (1.24). The *k*th multiplicative compound $T^{(k)}(t, s)$ of T(t, s) satisfies, from Definition 1.5,

- (i) $T^{(k)}(s,s) = I_X^{(k)} = I_{\bigwedge^k X};$
- (ii) $T^{(k)}(t,s)T^{(k)}(s,r) = T^{(k)}(t,r), 0 \le r \le s \le t$, from (ii) of Definition 1.5 and the Binet-Cauchy identity;
- (*iii*) $t \mapsto T^{(k)}(t,s)$ is continuous for $t \ge s$.

Thus $T^{(k)}(t,s)$ is also an evolution operator. Let $u^1(t), \dots, u^k(t)$ be k solutions of (1.24) with initial conditions $u^1_s, \dots, u^k_s \in \mathcal{D}(A)$. Since

$$T^{(k)}(t,s) u_s^1 \wedge \cdots \wedge u_s^k = \left(T(t,s)u_s^1\right) \wedge \cdots \wedge \left(T(t,s)u_s^k\right) = u^1(t) \wedge \cdots \wedge u^k(t)$$

and

$$\frac{d}{dt} \left[u^{1}(t) \wedge \dots \wedge u^{k}(t) \right] = \sum_{i} u^{1}(t) \wedge \dots \wedge \frac{d}{dt} u^{i}(t) \wedge \dots \wedge u^{k}(t)$$
$$= \sum_{i} u^{1}(t) \wedge \dots \wedge A(t) u^{i}(t) \wedge \dots \wedge u^{k}(t)$$
$$= A^{[k]}(t) \left[u^{1}(t) \wedge \dots \wedge u^{k}(t) \right]$$

it follows that $w(t) = u^{1}(t) \wedge \cdots \wedge u^{k}(t)$ is a solution of

$$\frac{dw}{dt} = A^{[k]}(t)w, t \ge s,$$

$$w(s) = u_s^1 \wedge \dots \wedge u_s^k.$$

This equation is called the kth compound differential equation of (1.24).

For example, if A(t) = A is a sectorial operator, then -A is the infinitesimal generator of an analytic semigroup $\{e^{-tA}\}_{t\geq 0}$ and the evolution operator $T(t,s) = e^{-(t-s)A}$ (see [87], page 20). The evolution operator of the kth compound differential equation of (1.24) is

$$T^{(k)}(t,s) = \left(e^{-(t-s)A}\right)^{(k)} = e^{-(t-s)A^{[k]}}.$$

Example 1.4. Consider a linear differential equation in \mathbb{R}^n

$$\frac{dy}{dt} = A(t)y, \tag{1.25}$$

where A(t) is a continuous real or complex matrix-valued function of t. Suppose that $y^1(t), \dots, y^k(t)$ are k solutions of (1.25). Then $w(t) = y^1(t) \wedge \dots \wedge y^k(t)$, is a solution of the *kth compound differential equation* of (1.24)

$$\frac{dw}{dt} = A^{[k]}(t)w, \qquad (1.26)$$

For the special case when k = n, it is well known that w(t) is the solution of the Abel-Liouville-Jacobi equation

$$\frac{dw}{dt} = \mathrm{tr}A(t)w.$$

If Y(t) is a fundamental matrix of (1.25), then $Y^{(k)}(t)$ is a fundamental matrix of (1.26). Here $T(t,s) = Y(t)Y^{-1}(s)$ and $T^{(k)}(t,s) = Y^{(k)}(t)(Y^{-1})^{(k)}(s)$. For example, if A is a constant matrix, then $Y(t) = e^{tA}$ and $Y^{(k)}(t) = (e^{tA})^{(k)} = e^{tA^{[k]}}$ is a fundamental matrix of (1.26). For a detailed exposition and applications in the finite dimensional situation, see [76–78].

Example 1.5. Consider a reaction diffusion equation

$$\frac{du}{dt} = (\Delta + A(t))u. \tag{1.27}$$

Suppose that (1.27) generates a semigroup on $H_0^1(0,1) = \{\phi \in L^2(0,1) : \phi' \in L^2(0,1), \phi(0) = \phi(1) = 0\}$. It is assume that the linear operator A(t) has the pointwise representation

$$\langle x, A(t)u \rangle = a(t, x)u(x). \tag{1.28}$$

This example illustrates how different choices of families of linear functionals v used to examine the exterior products of solutions of (1.27) yield different representations of the differential equation and its compound equations. Choosing first as v the pointwise evaluation functionals gives partial differential equations representing (1.27) and its compounds. Then, using eigenfunctions of the Laplacian a basis and the spectral projections as the linear operators v, equation (1.27) and its compounds are represented by infinite systems of coupled ordinary differential equations.

Case 1: A pointwise representation of the kth compound differential equation of (1.27)

For $u \in H_0^1(0,1), x \in (0,1)$, let x also denote a linear functional $u \mapsto \langle x, u \rangle = u(x)$. Then (1.27) can be written as

$$u_t = u_{xx} + a(t, x)u, \quad 0 < x < 1, \quad t > 0, u(t, 0) = u(t, 1) = 0, \quad t \ge 0.$$
 (1.29)

For $w(t) \in \bigwedge^k H_0^1(0, 1)$, a pointwise representation of w is

$$\langle x_1 \wedge \cdots \wedge x_k, w(t) \rangle = w(t, x_1, \cdots, x_k).$$

A pointwise representation of the kth compound operator $(\Delta + A(t))^{[k]} = \Delta^{[k]} + A(t)^{[k]} : \bigwedge^k H_0^1(0,1) \to \bigwedge^k L^2(0,1)$ follows from Example 1.3 and Section 1.2.2 and thus, a pointwise representation of the kth compound differential equation of (1.27) in $\bigwedge^k H_0^1(0,1)$ is

$$w_t = \sum_{i=1}^k w_{x_i x_i} + \sum_{i=1}^k a(t, x_i) w.$$
 (1.30)

If $u^1(t), \dots, u^k(t)$ are solutions of (1.29), then

$$\langle x_1 \wedge \cdots \wedge x_k, u^1(t) \wedge \cdots \wedge u^k(t) \rangle = \det[u^j(t, x_i)].$$

It can be checked by differentiating the above determinant that $u^1(t) \wedge \cdots \wedge u^k(t)$ satisfies the *k*th compound differential equation (1.30) and Dirichlet boundary conditions of u^i implies that

$$\langle x_1 \wedge \cdots \wedge x_k, u^1(t) \wedge \cdots \wedge u^k(t) \rangle \Big|_{x_i=0,1} = 0, \qquad i = 1, \cdots, k.$$

Again, (1.30) is not considered as an equation for $w(t, x_1, \dots, x_k)$ in $H_0^1((0,1)^k)$, but rather in $\bigwedge^k H_0^1(0,1)$.

Case 2: A basis representation of the kth compound differential equation of (1.27)

Let $\{e^j(x) = \sqrt{2} \sin j\pi x : j = 1, 2, \dots\}$ be an orthonormal basis of $H_0^1(0, 1)$. Define another linear map $e^j : u \mapsto \langle e^j, u \rangle$. Then $u(t) = \sum_j u_j(t)e^j$ and $u_j = \langle e^j, u \rangle$ satisfies

$$\frac{du_j}{dt} = [\lambda_j + a_j^j(t)]u_i + \sum_{i \neq j} a_j^i(t)u_j,$$
(1.31)

where $\lambda_j = -(j\pi)^2$ and $a_j^i = \langle e^i, ae^j \rangle = a_i^j$. For $w(t) \in \bigwedge^k H_0^1(0, 1)$, a basis representation of w is

$$\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, w(t) \rangle := w_{(i)}(t) = w_{(i_1 \cdots i_k)}(t).$$

A basis representation of the *k*th compound operator $(\Delta + A(t))^{[k]} = \Delta^{[k]} + A(t)^{[k]}$ follows from Example 1.3 and Section 1.2.2 and thus, a basis representation of the *k*th compound differential equation of (1.27) is

$$\frac{dw_{(i_1\cdots i_k)}}{dt} = \left(\lambda_{i_1} + \cdots + \lambda_{i_k} + a_{i_1}^{i_1}(t) + \cdots + a_{i_k}^{i_k}(t)\right) w_{(i_1\cdots i_k)}
+ \sum_{j\notin(i_1\cdots i_k)} \left(a_{i_1}^j(t)w_{(ji_2\cdots i_k)} + \cdots + a_{i_k}^j(t)w_{(i_1\cdots i_{k-1}j)}\right).$$
(1.32)

If $u^1(t), \dots, u^k(t)$ are solutions of (1.29), then

$$\langle e^{i_1} \wedge \dots \wedge e^{i_k}, u^1(t) \wedge \dots \wedge u^k(t) \rangle = \det \left[\langle e^{i_r}, u^p(t) \rangle \right] = u^{1 \dots k}_{i_1 \dots i_k}(t).$$

It can be checked by differentiating the above determinants that $u^1(t) \wedge \cdots \wedge u^k(t)$ satisfies the *k*th compound differential equation (1.32).

Chapter 2 Dimension Problems

This chapter addresses the problem of estimation of the codimension of the asymptotically stable solution subspace of a linear differential equation. This question arises not only for its intrinsic interest as a singular linear boundary value problem but also for its importance in investigations of the dimension of stable manifolds of equilibria and periodic orbits of nonlinear equations by linearization techniques.

The approach is motivated by results on the dimension of the asymptotically stable solution subspace of a differential equation

$$\frac{du}{dt} = A(t)u, \qquad u(\cdot) \in \mathbb{R}^n.$$
(2.1)

When n = 2, Milloux [74] shows that if 0 < a(t) is a nondecreasing function, then the scalar equation u'' + a(t)u = 0 has a nontrivial solution $u = u_0(t)$ such that

 $\lim_{t\to\infty} u_0(t) = 0 \quad \text{if and only if} \quad \lim_{t\to\infty} a(t) = \infty.$

Higher dimensional equations have been studied by Hartman [40], Coppel [19], Macki and Muldowney [65] and Muldowney [76–78]. Hartman [40] ([41], page 501) shows that if $0 \leq \lim_{t\to\infty} ||u(t)|| < \infty$ exists for all solutions u, and $|| \cdot ||$ is the Euclidean norm, then there exists a nontrivial solution such that $\lim_{t\to\infty} ||u_0(t)|| = 0$ if and only if $\lim_{t\to\infty} \int_0^t \operatorname{Re} \operatorname{tr} A = -\infty$. Hartman's proof relies heavily on properties of the norm. Coppel [19], page 60, shows that the result holds for any norm. Macki and Muldowney [65], by an entirely different approach, weaken the restriction that ||u(t)|| tends to a limit to a stability requirement on u(t). Muldowney [76–78] extends these results by

means of a sequence of compound matrices $A^{[k]}(t), k = 1, \dots, n$ of A(t) such that, under the same stability assumptions as in [65], the given system has an (n - k + 1)-dimensional asymptotically stable solution subspace if and only if all nontrivial solutions of the system

$$\frac{dw}{dt} = A^{[k]}(t)u$$

tend to zero.

The applications of compound matrices to differential equations are further developed in [51, 56, 58, 76–78]. In [78] a general result on the existence of an asymptotically stable subspace in a linear space of \mathbb{R}^n -valued functions is obtained. This chapter extends that result to general vector spaces. Nonlinear applications include estimation of the codimension of stable manifolds in Section 2.2. It also provides an important step in the proof of Theorem 4, Section 4.2.1, on the existence of stable periodic solutions.

2.1 Asymptotically stable subspaces

This section extends a result of Muldowney [78] on the existence of an asymptotically stable subspace in a set of vector space valued functions on $[0,\infty)$.

Let X be a vector space and X^* be its (algebraic) dual space. Let \mathcal{U} be a vector space of maps $t \mapsto u(t)$ from $[0, \infty)$ to X and \mathcal{V} be a family of maps $t \mapsto v(t)$ from $[0, \infty)$ to X^* .

Condition L: The pair $\{\mathcal{U}, \mathcal{V}\}$ satisfies Condition L if for each $u \in \mathcal{U}$, the following two assumptions hold:

- (i) $\limsup_{t\to\infty} |\langle v,u\rangle(t)| < \infty \text{ for every } v \in \mathcal{V};$
- (ii) $\liminf_{\substack{t \to \infty \\ \text{for every } v \in \mathcal{V}.}} |\langle v, u \rangle (t)| = 0 \text{ for every } v \in \mathcal{V} \text{ implies that } \lim_{t \to \infty} \langle v, u \rangle (t) = 0$

Here $\langle v, u \rangle (t) = \langle v(t), u(t) \rangle$.

Remark 2.1. For example, suppose that X is a normed space and that \mathcal{U} is a linear space of functions $t \mapsto u(t)$ from $[0, \infty)$ to X that satisfy in some
sense a linear differential equation of the form

$$\frac{du}{dt} = A(t)u, \qquad u(\cdot) \in X.$$
(2.2)

If \mathcal{V} is the set of all functions $t \mapsto v(t)$ from $[0, \infty)$ to X', the continuous dual space of X, such that ||v(t)|| = 1, then the pair $\{\mathcal{U}, \mathcal{V}\}$ satisfies Condition L provided that (i) and (ii) hold:

- (i) ||u(t)|| is bounded, $0 \le t < \infty$;
- (ii) $\liminf_{t\to\infty} \|u(t)\| = 0$ implies $\lim_{t\to\infty} \|u(t)\| = 0$.

The above conditions (i) and (ii) are satisfied if $\lim_{t\to\infty} ||u(t)|| < \infty$ exists for all $u \in \mathcal{U}$. Conditions (i) and (ii) are also satisfied if there exists a constant C such that

$$||u(t)|| \le C ||u(s)||, \ 0 \le s \le t < \infty.$$
(2.3)

Thus the solution space of a linear differential equation satisfies Condition L if the equation is uniformly stable. The condition (2.3) can be weakened to

$$||u(t)|| \le C_u ||u(s)||, \ 0 \le s \le t < \infty,$$

where the constant C_u depends on the solution u.

Let

$$\mathcal{U}_{0} = \left\{ u \in \mathcal{U} : \lim_{t \to \infty} \langle v, u \rangle (t) = 0 \text{ for every } v \in \mathcal{V} \right\}.$$
(2.4)

In the following proposition codim $\mathcal{U}_0 < k$ means that any subspace of \mathcal{U} that has dimension k or greater must intersect \mathcal{U}_0 nontrivially.

Proposition 2.1. Let \mathcal{U} be a vector space of maps $t \mapsto u(t)$ from $[0, \infty)$ to X and \mathcal{V} be a family of maps $t \mapsto v(t)$ from $[0, \infty)$ to X^* . Suppose that the pair $\{\mathcal{U}, \mathcal{V}\}$ satisfies Condition L. Then

codim
$$\mathcal{U}_0 < k$$

if and only if

$$\lim_{t \to \infty} \left\langle v_1 \wedge \dots \wedge v_k, u^1 \wedge \dots \wedge u^k \right\rangle(t) = 0$$
(2.5)

is satisfied for all $u^1, \cdots, u^k \in \mathcal{U}$ and $v_1, \cdots, v_k \in \mathcal{V}$.

Proof. Suppose that codim $\mathcal{U}_0 < k$. Then for every k linearly independent functions $u^1, \dots, u^k \in \mathcal{U}$, there exists a nontrivial linear combination $u_0 = c_1 u^1 + \dots + c_k u^k$ such that

$$\lim_{t \to \infty} \langle v, u_0 \rangle(t) = 0 \quad \text{for all } v \in \mathcal{V}.$$
(2.6)

Without loss of generality, it is assumed that $c_1 \neq 0$. Then, it follows from Condition L(i) and (2.6) that for every $v_1, \dots, v_k \in \mathcal{V}$,

$$\lim_{t \to \infty} \left\langle v_1 \wedge \dots \wedge v_k, u^1 \wedge \dots \wedge u^k \right\rangle (t)$$

= $\frac{1}{c_1} \lim_{t \to \infty} \left\langle v_1 \wedge \dots \wedge v_k, u_0^1 \wedge \dots \wedge u^k \right\rangle (t)$ (2.7)
= 0.

Conversely, suppose that (2.5) is satisfied. If $u^1, \dots, u^k \in \mathcal{U}$ are linearly independent, it is to be proved that $\operatorname{span}\{u^1, \dots, u^k\}$ intersects \mathcal{U}_0 nontrivially. The proof is by mathematical induction on k. It is evidently true when k = 1 since, in that case,

$$\left\langle v_{1}, u^{1} \right\rangle(t) \underset{t \to \infty}{\longrightarrow} 0$$

for all $v_1 \in \mathcal{V}$ and hence $u^1 \in \mathcal{U}_0$. Suppose that the proposition is true if $1 \leq k < h$. Let $v_1, \dots, v_h \in \mathcal{V}$. From Condition L(i) any sequence $t_n \xrightarrow[n \to \infty]{} \infty$ has a subsequence, which will also be denoted t_n , such that

$$\lim_{n \to \infty} \left\langle v_i, u^j \right\rangle (t_n) = \xi_i^j \tag{2.8}$$

exists $i, j = 1, \dots, h$. Let $\xi_{12\dots k}^{12\dots k}$ denote det $[\xi_i^j]$, $i, j = 1, \dots, k$. Now (2.5), k = h, implies $\xi_{12\dots h}^{12\dots h} = 0$. It may be assumed that t_n and $v_1, \dots, v_{h-1} \in \mathcal{V}$ can be chosen so that $\xi_{12\dots h-1}^{12\dots h-1} \neq 0$ since otherwise $\lim_{t\to\infty} \langle v_1 \wedge \dots \wedge v_{h-1}, u^1 \wedge \dots \wedge u^{h-1} \rangle$ (t) = 0 for all $v_1, \dots, v_{h-1} \in \mathcal{V}$ and the induction hypothesis implies that there is a nontrivial $u_0 = \sum_{j=1}^{h-1} c_j u^j \in \mathcal{U}_0$. Thus, with

$$\xi_{12\cdots h-1}^{12\cdots h-1} \neq 0, \qquad \xi_{12\cdots h}^{12\cdots h} = 0$$

expanding the determinant $\xi_{12\cdots h}^{12\cdots h}$ along the last row gives

$$0 = \xi_{12\cdots h}^{12\cdots h} = \sum_{j=1}^{h} c_j \xi_h^j$$
(2.9)

where c_j is the cofactor of ξ_h^j , $j = 1, \dots, h$. In particular, $c_h = \xi_{12\dots h-1}^{12\dots h-1} \neq 0$. Then (2.9) implies

$$0 = \sum_{j=1}^{h} c_j \xi_h^j = \lim_{n \to \infty} \sum_{j=1}^{h} c_j \left\langle v_h, u^j \right\rangle (t_n)$$

and thus $u_0 = \sum_{j=1}^h c_j u^j$ satisfies

$$\lim_{n \to \infty} \langle v_h, u_0 \rangle (t_n) = 0.$$
(2.10)

Furthermore, for any choice of $v_h \in \mathcal{V}$, a subsequence of t_n , also denoted t_n , may be chosen so that (2.8) is still satisfied and the preceding analysis shows that each choice of v_h satisfies (2.10) with the same constants c_j , which do not depend on v_h , for some sequence $t_n \xrightarrow[n \to \infty]{} \infty$. Therefore $\liminf_{t \to \infty} \langle v, u_0 \rangle (t) = 0$ for all $v \in \mathcal{V}$ which, by Condition L(ii), implies

$$\lim_{t \to \infty} \left\langle v, u_0 \right\rangle(t) = 0$$

and so

$$u_0 = \sum_{j=1}^h c_j u^j \in \mathcal{U}_0.$$

The proposition is thus true for k = h also and hence for all k by mathematical induction.

2.1.1 Function Space

Let X be a vector space of real-valued functions on a finite, countable or uncountable set Ω . For $x \in \Omega$, let x also denote the linear functional $u \mapsto \langle x, u \rangle := u(x), u \in X$. The pointwise representation of the exterior products discussed in Section 1.1.3 is $u^1 \wedge u^2 \wedge \cdots \wedge u^k$: $\bigwedge^k X \longrightarrow \mathbb{R}$,

$$\langle x_1 \wedge \cdots \wedge x_k, u^1 \wedge \cdots \wedge u^k \rangle = u^1 \wedge \cdots \wedge u^k (x_1, \cdots, x_k) = \det \left[u^i(x_j) \right].$$

Let \mathcal{U} be a linear space of maps $t \mapsto u(t, \cdot)$ and \mathcal{V} be the set of constant maps $\{t \mapsto v(t) = x : x \in X^*\}.$

The following corollary is a particular case of Proposition 2.1.

Corollary 2.2. Suppose that, for every $u \in U$, the following two conditions hold:

(i) $\limsup_{t \to \infty} |u(t,x)| < \infty$ for all $x \in \Omega$;

(ii) $\liminf_{t\to\infty} |u(t,x)| = 0$ for all $x \in \Omega$ implies $\lim_{t\to\infty} u(t,x) = 0$ for all $x \in \Omega$.

Then

$$codim\left\{ u \in \mathcal{U} : \lim_{t \to \infty} |u(t, x)| = 0 \quad for \ all \ x \in \Omega \right\} < k$$
(2.11)

if and only if, for each $u^1, \cdots, u^k \in \mathcal{U}$ and all $x_1, \cdots, x_k \in \Omega$,

$$\lim_{t \to \infty} \det \left[u^i(t, x_j) \right] = 0.$$

In particular, conditions (i) and (ii) are satisfied if, for every $u \in \mathcal{U}$, $\lim_{t \to \infty} |u(t,x)| < \infty$ exists for all $x \in \Omega$.

2.1.2 Normed Vector Space

Let X be a normed vector space, \mathcal{U} a linear space of functions $t \mapsto u(t)$ from $[0,\infty)$ to X and \mathcal{V} the set of functions $t \mapsto v(t)$ from $[0,\infty)$ to X', $\|v(t)\|_{X'} = 1$. Proposition 2.1 has the following corollary.

Corollary 2.3. Suppose that for every $u \in U$, the following two conditions are satisfied:

- (i) $\limsup_{t\to\infty} \|u(t)\|_X < \infty;$
- (*ii*) $\liminf_{t \to \infty} \|u(t)\|_X = 0$ implies $\lim_{t \to \infty} \|u(t)\|_X = 0$.

Then

$$codim\left\{ u \in \mathcal{U} : \lim_{t \to \infty} \left\| u\left(t\right) \right\|_X = 0 \right\} < k$$

if and only if, for all $u^1, \cdots, u^k \in \mathcal{U}$,

$$\lim_{t\to\infty}\left\|u^1\wedge\cdots\wedge u^k\left(t\right)\right\|_{\bigwedge^k X}=0.$$

In particular, conditions (i) and (ii) are satisfied if, for every $u \in \mathcal{U}$, $\lim_{t \to \infty} \|u(t)\|_X < \infty$ exists.

Consider X, a vector space of real-valued functions on Ω . The following table gives examples of the norms $||u(t)||_X$ and $||u^1 \wedge \cdots \wedge u^k(t)||_{\wedge^k X}$ in

$\left\ u\left(t ight) ight\ _{X}$	$\left\ u^{1} \wedge \cdots \wedge u^{k} \left(t \right) \right\ _{\wedge^{k} X}$
$\left(\int_{\Omega}\left u\left(t,x\right)\right ^{p}dx\right)^{\frac{1}{p}}$	$\left(rac{1}{k!}\int_{\Omega^k}\left \det\left[u^j\left(t,x_i\right) ight]\right ^pdx_1\cdots dx_k ight)^{rac{1}{p}}$
$\sup_{x\in\Omega}\left u\left(t,x\right)\right $	$\sup_{x_{1},\cdot\cdot,x_{k}\in\Omega}\left \det\left[u^{j}\left(t,x_{i}\right)\right]\right $
$\left(\sum_{i}\left u_{i}\left(t\right)\right ^{p}\right)^{\frac{1}{p}}$	$\left(\sum_{(i)}\left u_{i_{1}\cdots i_{k}}^{1\cdots k}\left(t\right)\right ^{p}\right)^{\frac{1}{p}}$
$\sup_{i}\left u_{i}\left(t\right)\right $	$\sup_{(i)}\left u_{i_{1}\cdots i_{k}}^{1\cdots k}\left(t\right)\right $

Table 2.1: Norms of kth exterior products

Corollary 2.3 for a set \mathcal{U} of functions $t \mapsto u(t)$ in this space. In the first two examples Ω is a measurable set in \mathbb{R}^n and in the last two examples Ω is a subset of the set \mathbb{N} of natural numbers. In each case, it is assumed that Xis the space of functions for which the expression defining the norm is finite. Notation in the last two examples is defined in Section 1.1.2.

The examples may be related, for example, through an orthonormal basis $\{e^i\}$ with respect to an inner product $\langle u, v \rangle = \int_{\Omega} uv$ and

$$egin{array}{rcl} u\left(t,x
ight) &\sim& \displaystyle{\sum_{i}u_{i}\left(t
ight)e^{i}(x),} \ u_{i}\left(t
ight) &=& \displaystyle{\int_{\Omega}u\left(t,x
ight)e^{i}\left(x
ight)dx} \end{array}$$

and, then

$$u^{1} \wedge \cdots \wedge u^{k}(t, x_{1}, \cdots, x_{k}) \sim \sum_{(i)} u^{1 \cdots k}_{i_{1} \cdots i_{k}}(t) e^{i_{1}} \wedge \cdots \wedge e^{i_{k}}(x_{1}, \cdots, x_{k}).$$

Remark 2.2. When X is a Hilbert space and p = 2, the second and fourth lines in the table above give the same condition in Corollary 2.3.

2.1.3 Differential Equations

In Remark 2.1, Condition L is discussed in the context of the solution space \mathcal{U} of a linear differential equation. This discussion is continued with

applications of Proposition 2.1 to estimation of the codimension of the asymptotically stable subspace \mathcal{U}_0 .

Theorem 2.4. Suppose that X is a normed space and that the differential equation

$$\frac{du}{dt} = A\left(t\right)u$$

has a solution space \mathcal{U} that satisfies, if $u \in \mathcal{U}$,

(i) $\limsup_{t \to \infty} \|u(t)\|_X < \infty;$

(ii) $\liminf_{t \to \infty} \|u(t)\|_{X} = 0 \text{ implies } \lim_{t \to \infty} \|u(t)\|_{X} = 0.$

Then

$$codim\left\{ u \in \mathcal{U} : \lim_{t \to \infty} \|u(t)\|_X = 0 \right\} < k$$

if and only if all solutions $w = u^1 \wedge \cdots \wedge u^k$, $u^i \in \mathcal{U}$, of the kth compound equation

$$\frac{dw}{dt} = A^{\left[k\right]}\left(t\right)w$$

satisfy

$$\lim_{t\to\infty}\left\|w\left(t\right)\right\|_{\bigwedge^{k}X}=0.$$

Remark 2.3. In particular conditions (i) and (ii) of Theorem 2.4 are satisfied if, for every $u \in \mathcal{U}$, $\lim_{t \to \infty} ||u(t)||_X < \infty$ exists or if the equation $\frac{du}{dt} = A(t)u$ is uniformly stable. The finite dimensional results discussed in the introduction to this chapter are all special cases of Theorem 2.4.

Consider a scalar reaction diffusion equation

$$\frac{du}{dt} = (\Delta + A(t))u, \quad u(\cdot) \in H_0^1(0, 1),$$
(2.12)

where the linear operator A(t) has the pointwise representation

$$\langle x, A(t)u \rangle = a(t, x)u(x).$$

In Example 1.5 of Section 1.3, two representations of the kth compound differential equation of (2.12) are considered.

Case 1: The pointwise representation of the *k*th compound differential equation of (2.12) on $\bigwedge^k H_0^1(0,1)$ is

$$w_t = \sum_{i=1}^k w_{x_i x_i} + \sum_{i=1}^k a(t, x_i) w.$$
(2.13)

Case 2: The basis representation of the *k*th compound differential equation of (2.12) on $\bigwedge^k H_0^1(0,1)$ is

$$\frac{dw_{(i_{1}\cdots i_{k})}}{dt} = (\lambda_{i_{1}} + \dots + \lambda_{i_{k}} + a_{i_{1}i_{1}}(t) + \dots + a_{i_{k}i_{k}}(t)) w_{(i_{1}\cdots i_{k})}
+ \sum_{j \notin (i_{1}\cdots i_{k})} (a_{i_{1}j}(t)w_{(ji_{2}\cdots i_{k})} + \dots + a_{i_{k}j}(t)w_{(i_{1}\cdots i_{k-1}j)})$$
(2.14)

where $\langle e^{i_1} \wedge \cdots \wedge e^{i_k}, w(t) \rangle := w_{(i)}(t) = w_{(i_1 \cdots i_k)}(t), \ e^j(x) = \sqrt{2} \sin j\pi x,$ $\lambda_j = -(j\pi)^2 \text{ and } a_{ij} = a_{ji} = \langle e^i, ae^j \rangle.$

To compare different choices of the functionals in X', Theorem 2.4 implies the following two corollaries. Corollary 2.5 discusses the problem in $L^2(0,1)$, $||u||_{L^2}^2 = \int_0^1 u(t,x)^2 dx$, while Corollary 2.6 considers the l_{∞} norm, $||u||_{\infty} = \sup_i |u_i(t)|$.

Corollary 2.5. Suppose that there exists a constant M such that

$$\int_{s}^{t} \left[\lambda_{1} + \max_{x} a(\tau, x) \right] d\tau \le M, \quad 0 \le s \le t < \infty,$$
(2.15)

where the maximum is taken over $0 \le x \le 1$ and $\lambda_j = -(j\pi)^2$, $j = 1, 2, \cdots$. Then

codim
$$\left\{ u(\cdot) \text{ satisfies (2.12) and } \lim_{t \to \infty} \|u(t)\|_{L^2} = 0 \right\} < k$$

if

$$\int_0^\infty \left[\lambda_1 + \dots + \lambda_k + k \max_x a(t, x) \right] dt = -\infty.$$
 (2.16)

Corollary 2.6. Suppose that there exists a constant M such that

$$\int_{s}^{t} \sup_{i} \left\{ \lambda_{i} + a_{ii}(\tau) + \sum_{j \neq i} |a_{ij}(\tau)| \right\} d\tau \leq M, \quad 0 \leq s \leq t < \infty, \quad (2.17)$$

where $a_{ij} = a_{ji} = \langle e^i, ae^j \rangle$ and $\lambda_j = -(j\pi)^2$, $j = 1, 2, \cdots$. Then $codim \left\{ u(\cdot) \text{ satisfies (2.12) and } \lim_{t \to \infty} \|u(t)\|_{\infty} = 0 \right\} < k$

if

$$\int_{0}^{\infty} \sup_{(i)} \left\{ \sum_{s=1}^{k} \left(\lambda_{i_s} + a_{i_s i_s}(\tau) \right) + \sum_{j \notin (i)} \left(|a_{i_1 j}(\tau)| + \dots + |a_{i_k j}(\tau)| \right) \right\} d\tau = -\infty$$
(2.18)

where $(i) = (i_1, \dots, i_k), i_1 < i_2 < \dots < i_k$.

Remark 2.4. In Corollary 2.5, condition (2.15) implies that the reaction diffusion equation (2.12) is uniformly stable in $L^2(0,1)$ by considering $||u||_{L^2}$ as a Lyapunov function. The asymptotic stability of the *k*th compound differential equation (2.13) in $\bigwedge^k L^2(0,1)$ is obtained from condition (2.16) by considering $||w||_{\bigwedge^k L^2}$. Similar arguments using conditions (2.17) and (2.18) with $||u||_{\infty}$ and $||w||_{\bigwedge^k l_{\infty}}$ prove Corollary 2.6 respectively.

2.2 Dimension Problems for Steady State Solutions

This section derives results on the stability of steady state solutions of differential equations. For an autonomous ordinary differential equation

$$\frac{du}{dt} = f(u), \qquad u(\cdot) \in \mathbb{R}^n,$$

a necessary and sufficient condition for an equilibrium $u = u^*$ to be stable hyperbolic is that the linear variational equation at $u = u^*$,

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(u^*)v,$$

is uniformly asymptotically stable. General results on the stability of steady state solutions of differential equation in Banach space can be found in Smoller [103], Theorem 11.20, page 120 and Henry [42], Theorem 5.1.1, page 98.

Let A be a sectorial operator in a Banach space X and f be continuously differentiable from X^{α} into X where $0 \leq \alpha < 1$. Let $\mathcal{D}(A)$ denote the domain of A. Consider an autonomous differential equation

$$\frac{du}{dt} + Au = f(u), \qquad u(\cdot) \in X.$$
(2.19)

Equation (2.19) has been discussed in Henry [42]. Definitions and properties of sectorial operators and the space X^{α} are listed in Appendix B.

Definition 2.1. A solution of the initial value problem

$$\frac{du}{dt} + Au = f(u),$$
$$u(0) = u_0,$$

on (0,T) is a continuous function $u : [0,T) \to X$ such that $u(0) = u_0$ and on (0,T), $(t,u(t)) \in \mathbb{R} \times X^{\alpha}$, $u(t) \in \mathcal{D}(A)$, $\frac{du}{dt}(t)$ exists and the differential equation (3.1) is satisfied on (0,T).

Lemma 2.7. Assume that A is a sectorial operator, $0 \le \alpha < 1$, and that $f: X^{\alpha} \to X$ is locally Lipschitz continuous in u. Then for any $u_0 \in X^{\alpha}$, there exists $T = T(u_0) > 0$ such that (3.1) has a unique solution u on (0,T) with initial value $u(0) = u_0$.

Definition 2.2. A solution is a steady state solution or (an equilibrium) if $u = u^* \in \mathcal{D}(A)$ and $Au^* = f(u^*)$. The steady state solution u^* is stable hyperbolic if the spectrum of $A - \frac{\partial f}{\partial u}(u^*)$ lies in $\{\operatorname{Re} \lambda > \beta\}$ for some $\beta > 0$.

Theorem 2.8. A steady state solution u^* is stable hyperbolic if and only if the linear variational equation at $u(t) = u^*$,

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u^*)v, \qquad (2.20)$$

is uniformly asymptotically stable.

Remark 2.5. Definition 2.2 is equivalent to Definition 11.19 in Smoller [103], page 120. Theorem 11.20 in Smoller [103], page 120, shows that the hyperbolic stability of u^* is equivalent to the uniform asymptotic stability of (2.20). Theorem 5.1.1 in Henry [42] proves that if f = f(t, u), then $u = u^*$ is stable hyperbolic if

$$\frac{dv}{dt} + Av = Bv$$

is uniformly asymptotically stable where B is a bounded linear operator from X^{α} to X such that

$$f(t, u^* + z) = f(t, u^*) + Bz + g(t, z)$$

and $||g(t,z)|| = o(||z||_{\alpha})$ as $||z||_{\alpha} \to 0$.

The effect of adding diffusion to the dynamics of an ordinary differential equation in \mathbb{R}^n has been studied by [16, 17, 57, 62, 63, 81, 83, 85, 103, 105]. Turing [105] is the first to demonstrate that different diffusion coefficients can cause a stable equilibrium of the ordinary differential equation to cease to be stable for the reaction diffusion equation. The diffusion-driven instability has become an important mechanism for the occurrence of interesting patterns in many model systems. Turing's idea has been explored by many authors; see [17, 81, 83, 85]. With the help of compound matrices, for a stable matrix A with real entries, Wang and Li [57] derive sufficient and necessary conditions for A - D to be stable for any nonnegative diagonal matrix D. These conditions can be used to study the stability and instability of constant steady state solutions to reaction diffusion equations.

In the following, the stability of steady state solutions of scalar reaction diffusion equations is studied. Theorem 2.4 gives an estimate of the codimension of the asymptotically stable solution subspace, which is related to the stability of a compound differential equation. Rather than use Theorem 2.4 directly, estimates on the spectrum of an operator A from the spectrum of its compound $A^{[k]}$ are derived.

Consider a nonlinear scalar reaction diffusion equation

$$u_t = u_{xx} + f(x, u), \quad a < x < b, \quad t > 0, u(t, a) = u(t, b) = 0, \quad t > 0.$$
 (2.21)

Assume that (2.21) generates a semiflow on $H_0^1(a, b)$ and that u(t, x) exists for all t. Suppose that $u = u^*(x)$ is a steady state solution of (2.21). The linear variational equation at $u = u^*(x)$ is

$$v_t = v_{xx} + q(x)v, \qquad a < x < b, \quad t > 0, v(t, a) = v(t, b) = 0, \qquad t > 0,$$
(2.22)

where

$$q(x) := f_u(x, u^*(x)).$$
(2.23)

Assume that (2.22) also generates a semiflow on $H_0^1(a, b)$ and that v(t, x) exists for all t. Theorem 2.8 states that the steady state solution $u = u^*(x)$ of (2.21) is stable hyperbolic if the linear variational equation (2.22) is uniformly asymptotically stable and thus, if the eigenvalues of the following eigenvalue problem

$$\begin{aligned} \lambda \phi &= \phi'' + q(x)\phi, \\ \phi(a) &= \phi(b) = 0, \end{aligned} \tag{2.24}$$

satisfy $\lambda < 0$, the steady state solution $u = u^*(x)$ is stable hyperbolic. The eigenvalue problem (2.24) is a Sturm-Liouville boundary value problem; see Hartman [41], page 337-344 for details.

Let $L\phi = \phi'' + q(x)\phi$. Then L is a self-adjoint operator on $H_0^1(a, b)$. The standard analysis in [2, 20, 21, 103] on the eigenvalue problem of a self-adjoint second order differential equation implies that the principal eigenvalue of (2.24) is

$$\lambda_1 = \sup_{\phi \in H_0^1} \int_a^b \left[-\left(\phi'\right)^2 + q\phi^2 \right] \left/ \int_a^b \phi^2$$

where the supremum is taken over the functions $\phi \in H_0^1(a, b)$ with $\int_a^b \phi^2 \neq 0$ and the maximizing function ϕ_1 is the corresponding eigenfunction. In particular, if $\mu_1 \geq \mu_2 \geq \cdots$ and e_1, e_2, \cdots are the eigenvalues and orthonormal eigenfunctions of the Laplacian in $H_0^1(a, b)$,

$$\mu_1 = \sup_{\phi \in H_0^1} \int_a^b - (\phi')^2 \left/ \int_a^b \phi^2 = - \int_a^b (e_1')^2 \right.$$

Therefore

$$\mu_{1} + \int_{a}^{b} q(e_{1})^{2} \leq \lambda_{1} \leq \mu_{1} + \max_{a < x < b} q(x).$$

Since L is self-adjoint, the kth additive compound operator $L^{[k]}$ of L is also self-adjoint on $\bigwedge^k H_0^1(a, b)$. Let $\lambda_1 \geq \lambda_2 \geq \cdots$ be the eigenvalues of L, counting multiplicities, and ϕ_1, ϕ_2, \cdots be the corresponding eigenfunctions. Then $\{\phi_{i_1} \wedge \cdots \wedge \phi_{i_k} : 1 \leq i_1 < \cdots < i_k\}$ forms a basis of $\bigwedge^k H_0^1(a, b)$ and the principal eigenvalue of $L^{[k]}$ is $\lambda_1 + \cdots + \lambda_k$ satisfying

$$\lambda_1 + \dots + \lambda_k = \sup_{w \in \bigwedge^k H_0^1} \int_{(a,b)^k} \sum_{i=1}^k \left[-(w_{x_i})^2 + q(x_i) w^2 \right] \bigg/ \int_{(a,b)^k} w^2 ,$$

where the supremum is taken over the functions $w \in \bigwedge^k H_0^1(a, b)$ with $\int_{(a,b)^k} w^2 \neq 0$. Therefore

$$\mu_1 + \dots + \mu_k + \int_{(a,b)^k} \sum_{i=1}^k q(x_i) (e_1 \wedge \dots \wedge e_k)^2 \left/ \int_{(a,b)^k} (e_1 \wedge \dots \wedge e_k)^2 \right|^2$$

$$\leq \lambda_1 + \dots + \lambda_k \leq \mu_1 + \dots + \mu_k + k \max_x q(x).$$

It follows from

$$\begin{split} S^{j_1 \cdots j_r}_{i_1 \cdots i_r} &= \langle e_{i_1} \wedge \cdots \wedge e_{i_r}, e_{j_1} \wedge \cdots \wedge e_{j_r} \rangle \\ &= \frac{1}{r!} \int_{(a,b)^r} \left(e_{i_1} \wedge \cdots \wedge e_{i_r} \right) \left(e_{j_1} \wedge \cdots \wedge e_{j_r} \right), \end{split}$$

where $r \geq 1$, that

$$\int_{(a,b)^k} \left(e_1 \wedge \dots \wedge e_k \right)^2 = k! \tag{2.25}$$

and

$$\int_{(a,b)^{k}} \sum_{i=1}^{k} q(x_{i}) (e_{1} \wedge \dots \wedge e_{k})^{2} = k! \sum_{i=1}^{k} \int_{a}^{b} q(e_{i})^{2}$$

which implies the following proposition.

Proposition 2.9. Let $\mu_1 \geq \mu_2 \geq \cdots$ and e_1, e_2, \cdots be the eigenvalues and orthonormal eigenfunctions of the Laplacian in $H_0^1(a,b)$. Then the eigenvalues $\lambda_1 \geq \lambda_2 \geq \cdots$ of the eigenvalue problem (2.24) satisfy

$$\mu_1 + \dots + \mu_k + \sum_{i=1}^k \int_a^b q(e_i)^2$$

$$\leq \lambda_1 + \dots + \lambda_k \leq \mu_1 + \dots + \mu_k + k \max_{a < x < b} q(x).$$
(2.26)

Remark 2.6. The inequalities (2.26) are sharp in the sense that both can be replaced by equality when q(x) is a constant. The above techniques can be applied to a more general eigenvalue problem, for example,

$$\lambda \phi = \Delta \phi + a(x)\phi,$$

with homogeneous boundary conditions on $\partial\Omega$,

$$u(\partial \Omega) = 0$$
, or $\frac{du}{dn} + b(x)u = 0$,

where Δ is the Laplacian on Ω , a is bounded and b is a piecewise continuous function on $\partial\Omega$. Here Ω is a bounded domain of \mathbb{R}^n and $\partial\Omega$ is smooth.

The estimates obtained in Proposition 2.9 on the eigenvalues for the regular Sturm-Liouville boundary value problem (2.24) may be of independent interest and a particular case is noted separately in the following proposition with a = 0, b = 1. In this case, $\mu_j = -(j\pi)^2$, $e_j = \sqrt{2}\sin(j\pi x)$, $j = 1, 2, \cdots$.

Proposition 2.10. The eigenvalues $\lambda_1 \geq \lambda_2 \geq \cdots$ of the Sturm-Liouville boundary value problem

$$\begin{aligned} \lambda \phi &= \phi'' + q(x)\phi, \\ \phi(0) &= \phi(1) = 0, \end{aligned}$$

satisfy

$$-\frac{1}{6}k(k+1)(2k+1)\pi^{2} + 2\sum_{j=1}^{k}\int_{0}^{1}q(x)\sin^{2}(j\pi x) dx$$

$$\leq \sum_{j=1}^{k}\lambda_{j} \leq -\frac{1}{6}k(k+1)(2k+1)\pi^{2} + k\max_{0\leq x\leq 1}q(x).$$
(2.27)

Stability of the steady state solution $u = u^*$ of the scalar reaction diffusion equation (2.21) follows from Theorem 2.8 and Proposition 2.9.

Corollary 2.11. If

$$\mu_1 + \max_{a < x < b} q(x) < 0, \tag{2.28}$$

then $u = u^*(x)$ is stable hyperbolic.

Corollary 2.12. If

$$\mu_1 + \dots + \mu_k + k \max_{a < x < b} q(x) < 0, \qquad (2.29)$$

then $u = u^*(x)$ has a stable manifold with codimension at most k - 1.

The above procedure also applies to a more general reaction diffusion equation $u_t = u_{xx} + f(x, u, u_x)$. However, the eigenvalue problems of its linear variational equation and compound differential equations need to be transformed to the self-adjoint form. An example of $f = f(u, u_x)$ is explained as follows.

Example 2.1. Consider a scalar reaction diffusion equation

$$u_t = u_{xx} + u + \varepsilon u(1 - u^2 - u_x^2), \quad 0 < x < 2\pi, \quad t > 0, u(t, 0) = u(t, 2\pi), \quad u_x(t, 0) = u_x(t, 2\pi), \quad t > 0.$$
(2.30)

When $\varepsilon \geq 0$, since

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_{0}^{2\pi} u^{2} dx &= \int_{0}^{2\pi} u_{t} u dx \\ &= \int_{0}^{2\pi} [-u_{x}^{2} + (1+\varepsilon)u^{2} - \varepsilon (u^{4} + u_{x}^{2}u^{2})] dx \\ &\leq (1+\varepsilon) \int_{0}^{2\pi} u^{2} \end{aligned}$$

and

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_{0}^{2\pi} u_{x}^{2} dx &= -\int_{0}^{2\pi} u_{t} u_{xx} \\ &= \int_{0}^{2\pi} \left[-u_{xx}^{2} + (1+\varepsilon)u_{x}^{2} + \varepsilon(u^{3} + uu_{x}^{2})u_{xx} \right] dx \\ &= \int_{0}^{2\pi} \left[-u_{xx}^{2} + (1+\varepsilon)u_{x}^{2} - \varepsilon \left(3u^{2}u_{x}^{2} + \frac{u_{x}^{4}}{3} \right) \right] dx \\ &\leq (1+\varepsilon) \int_{0}^{2\pi} u_{x}^{2} dx, \end{aligned}$$

the solution u(t,x) of (2.30) exists in $X = \{u \in H^1(0,2\pi) : u(0) = u(2\pi), u_x(0) = u_x(2\pi)\}$ for all $t \ge 0$. Each solution of (2.30) can be written as $u(t,x) = a_0(t) + \sum_{j=1}^{\infty} (a_j(t)\cos(jx) + b_j(t)\sin(jx))$. It can be shown that the subspaces $S_1 = \text{span}\{1\}$ and $S_2 = \text{span}\{\cos(x), \sin(x)\}$ are invariant with respect to (2.30). In fact, if $u = a_0(t) + a_1(t)\cos(x) + b_1(t)\sin(x)$, then

$$\begin{aligned} a_0'(t) + a_1'(t)\cos(x) + b_1'(t)\sin(x) \\ &= -a_1(t)\cos(x) - b_1(t)\sin(x) + a_0(t) + a_1(t)\cos(x) + b_1(t)\sin(x) \\ &+ \varepsilon(a_0(t) + a_1(t)\cos(x) + b_1(t)\sin(x)) \\ &\cdot (1 - a_0^2(t) - a_1^2(t) - b_1^2(t) - 2a_0(t)a_1(t)\cos(x) - 2a_0(t)b_1(t)\sin(x)) \end{aligned}$$

Thus, if $u = a_0(t)$, then

$$a_0' = a_0 + \varepsilon a_0 (1 - a_0^2);$$

if $u = a_1(t)\cos(x) + b_1(t)\sin(x)$, then

$$\begin{aligned}
a_1' &= \varepsilon a_1 (1 - a_1^2 - b_1^2), \\
b_1' &= \varepsilon a_1 (1 - a_1^2 - b_1^2).
\end{aligned}$$
(2.31)

When $\varepsilon > 0$, the reaction diffusion equation (2.30) has steady states solutions u = 0, $u = \pm \sqrt{\frac{1+\varepsilon}{\varepsilon}}$ and $u = \cos(x + \alpha)$, where $\alpha \in \mathbb{R}$. The stability of these



Figure 2.1: Stability of steady state solutions in the invariant subspace S_1 and S_2 .

three steady state solutions in the invariant space S_1 and S_2 is illustrated by Figure 2.1.

In the following the stability of these steady states solutions in the whole space X will be studied.

Steady state Solution u = 0:

The linear variational equation at u = 0 is

$$egin{aligned} & v_t = v_{xx} + (1+arepsilon)v, & 0 < x < 2\pi, \quad t > 0, \ & v(t,0) = v(t,2\pi), & v_x(t,0) = v_x(t,2\pi), & t > 0 \end{aligned}$$

and its eigenvalue problem is

$$\lambda \phi = \phi'' + (1 + \varepsilon)\phi, \phi(0) = \phi(2\pi), \quad \phi'(0) = \phi'(2\pi),$$
(2.32)

for some λ , which has eigenvalues

$$\lambda_1 = 1 + \varepsilon > 0, \quad \lambda_2 = \lambda_3 = \varepsilon > 0, \quad \lambda_4 = \lambda_5 = -2^2 + 1 + \varepsilon, \quad \cdots,$$

and corresponding eigenfunctions

 $\phi_1 = 1, \quad \phi_2 = \cos(x), \quad \phi_3 = \sin(x), \quad \phi_4 = \cos(2x), \quad \phi_5 = \sin(2x), \quad \cdots,$ respectively.

Case 1: $n^2 - 1 < \varepsilon < (n+1)^2 - 1, n \ge 2$

u = 0 has a (2n + 1)-dimensional unstable manifold and a stable manifold with codimension (2n + 1).

Case 2: $\varepsilon = n^2 - 1, n \ge 2$

u = 0 has a (2n+1)-dimensional unstable manifold, a 2-dimensional center manifold, a stable manifold with codimension (2n+3).

Steady state solution $u = \pm \sqrt{\frac{1+\varepsilon}{\varepsilon}}$:

The linear variational equation at $u = \pm \sqrt{\frac{1+\varepsilon}{\varepsilon}}$ is

$$\begin{aligned} & v_t = v_{xx} - 2(1+\varepsilon)v, \quad 0 < x < 2\pi, \quad t > 0, \\ & v(t,0) = v(t,2\pi), \quad v_x(t,0) = v_x(t,2\pi), \quad t > 0 \end{aligned}$$

and its eigenvalue problem is

$$\begin{aligned} \lambda \phi &= \phi'' - 2(1+\varepsilon)\phi, \\ \phi(0) &= \phi(2\pi), \quad \phi'(0) &= \phi'(2\pi), \end{aligned} \tag{2.33}$$

for some λ , which has eigenvalues

$$\lambda_1 = -2(1+\varepsilon) < 0, \quad \lambda_2 = \lambda_3 = -1 - 2(1+\varepsilon) < 0,$$
$$\lambda_4 = \lambda_5 = -2^2 - 2(1+\varepsilon) < 0, \quad \cdots,$$

and corresponding eigenfunctions

 $\phi_1 = 1$, $\phi_2 = \cos(x)$, $\phi_3 = \sin(x)$, $\phi_4 = \cos(2x)$, $\phi_5 = \sin(2x)$, \cdots , respectively. Therefore $u = \pm \sqrt{\frac{1+\varepsilon}{\varepsilon}}$ is stable.

Steady state solution $u = \cos(x + \alpha)$:

Only the case u = cos(x) will be discussed and other cases are similar. The linear variational equation at u = cos(x) is

$$v_t = v_{xx} + \varepsilon \sin(2x)v_x + (1 - \varepsilon - \varepsilon \cos(2x))v, \quad 0 < x < 2\pi, \quad t > 0, v(t,0) = v(t,2\pi), \quad v_x(t,0) = v_x(t,2\pi), \quad t > 0$$
(2.34)

and its eigenvalue problem is

$$\lambda \phi = \phi'' + \varepsilon \sin(2x)\phi' + (1 - \varepsilon - \varepsilon \cos(2x))\phi,$$

$$\phi(0) = \phi(2\pi), \quad \phi'(0) = \phi'(2\pi),$$
(2.35)

for some λ , which has solutions

$$\phi_1 = \sin(x), \quad \text{when } \lambda_1 = 0,$$

and

$$\phi_2 = \cos(x), \quad \text{when } \lambda_2 = -2\varepsilon < 0.$$

Rewrite (2.35) as a self-adjoint second order differential equation

$$L\phi - \lambda p\phi = (p\phi')' + q\phi - \lambda p\phi = 0, \qquad (2.36)$$

with periodic boundary condition, where

$$p(x) = \exp\left(-\frac{\varepsilon}{2}\cos(2x)\right) > 0, \quad q(x) = (1 - \varepsilon - \varepsilon\cos(2x))p(x).$$

Then the first principal eigenvalue λ_p of (2.36), and thus of (2.35), is given by

$$\lambda_p = \sup_{\phi \in H_{per}^1} \int_0^{2\pi} \left[-p(\phi')^2 + q\phi^2 \right] \left/ \int_0^{2\pi} p\phi^2 \right.$$
(2.37)

where the supremum is taken over $\phi \in H^1_{per}(0, 2\pi) = \{\phi \in L^2(0, 2\pi) : \phi' \in L^2(0, 2\pi), \phi(0) = \phi(2\pi), \phi'(0) = \phi'(2\pi)\}$ with $\int_0^{2\pi} p\phi^2 \neq 0$. Substitute $\phi = \frac{1}{\sqrt{p(x)}}$ into the right-hand side of (2.37) to obtain

$$\lambda_p \ge 1 - \varepsilon - \frac{\varepsilon^2}{8}$$

Moreover,

 $\lambda_p \leq 1.$

For $\varepsilon > 0$, $1 - \varepsilon - \frac{\varepsilon^2}{8} = 0$ when $\varepsilon = 2(\sqrt{6} - 2) \approx 0.898979$ (see Figure 2.2). If $0 < \varepsilon < 2(\sqrt{6} - 2) \approx 0.898979$, then $\lambda_p > 0$ and thus λ_1, λ_2 and λ_p are three different eigenvalues of (2.36).

If v^1, v^2, v^3, v^4 are solutions of (2.34), then $w(t) = (v^1 \wedge v^2 \wedge v^3 \wedge v^4)(t) \in \bigwedge^4 X$ has a pointwise representation

$$w(t,x_1,x_2,x_3,x_4) = \det \left[v^i(t,x_j)
ight],$$

which satisfies the 4th compound differential equation of the linear variational equation (2.34)

$$w_t = \sum_{i=1}^4 w_{x_i x_i} + \varepsilon \sum_{i=1}^4 \sin(2x_i) w_{x_i} + 4(1-\varepsilon) w - \varepsilon \sum_{i=1}^4 \cos(2x_i) w. \quad (2.38)$$

Consider a Lyapunov function

$$V(t) = \int_{(0,2\pi)^4} w^2 = \int_{(0,2\pi)^4} w^2(t, x_1, x_2, x_3, x_4) dx_1 dx_2 dx_3 dx_4.$$

Then

$$\begin{aligned} \frac{d^+V}{dt} &= \int_{(0,2\pi)^4} ww_t \\ &= \int_{(0,2\pi)^4} \left(w \sum_{j=1}^4 w_{x_j x_j} + 4(1-\varepsilon) w^2 - \varepsilon \sum_{j=1}^4 \cos(2x_j) w^2 \right) \\ &+ \int_{(0,2\pi)^4} \sum_{j=1}^4 \varepsilon \sin(2x_j) ww_{x_j} \\ &\leq (0-1^2-1^2-2^2+4(1-\varepsilon)) \int_{(0,2\pi)^4} w^2 \\ &- 2\varepsilon \int_{(0,2\pi)^4} \sum_{j=1}^4 \cos(2x_j+2ct) w^2 \\ &\leq (-2-4\varepsilon+2\varepsilon \cdot 4) \int_{(0,2\pi)^4} w^2 = 2(-2+4\varepsilon) V. \end{aligned}$$

Thus, for any four eigenvalues $\lambda_{i_1}, \lambda_{i_2}, \lambda_{i_3}, \lambda_{i_4}$ of (2.35), by choosing $w = \phi_{i_1} \wedge \cdots \wedge \phi_{i_4}$ where $\phi_{i_s}, s = 1, \cdots, 4$ are the corresponding eigenfunctions, it follows that

$$\lambda_{i_1} + \lambda_{i_2} + \lambda_{i_3} + \lambda_{i_4} \le -2 + 4\varepsilon.$$

In particular,

$$\lambda_p + \lambda_1 + \lambda_2 + \lambda_3 \le -2 + 4\varepsilon.$$

Therefore,

$$\lambda_3 \le -3 + 7\varepsilon + \frac{\varepsilon^2}{8}.$$

For $\varepsilon > 0$, $-3 + 7\varepsilon + \frac{\varepsilon^2}{8} = 0$ when $\varepsilon = 2(\sqrt{202} - 14) \approx 0.425341$ and $1 - \varepsilon - \frac{\varepsilon^2}{8} = -3 + 7\varepsilon + \frac{\varepsilon^2}{8}$ when $\varepsilon = 2(13\sqrt{2} - 18) \approx 0.769553$ (See Figure 2.2).

Corollary 2.13. If $0 < \varepsilon < 2(\sqrt{202} - 14)$, then $u = \cos(x)$ has a 1-dimensional center manifold, a 1-dimensional unstable manifold and a stable manifold with codimension 2.

Corollary 2.14. If $2(\sqrt{202} - 14) \le \varepsilon < 2(13\sqrt{2} - 18)$, then $u = \cos(x)$ has a stable manifold with codimension at most 3.



Figure 2.2: Graphs of $y = 1 - \varepsilon - \frac{\varepsilon^2}{8}$ and $y = -3 + 7\varepsilon + \frac{\varepsilon^2}{8}$.

Chapter 3

On Poincaré's Stability Criterion for Periodic Orbits

This chapter deals with the stability of periodic solutions of differential equations. A fundamental result on this topic is that a nonconstant periodic solution is orbitally asymptotically stable with asymptotic phase if, with the exception of a single characteristic multiplier which equals 1, the moduli of all characteristic multipliers of the periodic solution are less than 1. For a 2-dimensional autonomous ordinary differential equation, this is equivalent to Poincaré's stability criterion: a nonconstant ω -periodic solution $u = \phi(t)$ of

$$\frac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^2,$$

is orbitally asymptotically stable with asymptotic phase if

$$\int_0^{\omega} \operatorname{div} \, f(\phi(t)) dt < 0.$$

This is the requirement that the Liouville equation

$$rac{dw}{dt} = {
m div}\; f(\phi(t))w$$

be uniformly asymptotically stable, which means that 2-dimensional areas near the orbit of $\phi(t)$ decrease exponentially under the flow of the differential equation and, as a consequence, nearby orbits are attracted to the orbit of $\phi(t)$.

A higher dimensional generalization of Poincaré's stability criterion is obtained by Muldowney [78]: a nonconstant ω -periodic solution $u = \phi(t)$ of

$$\frac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^n,$$

is orbitally asymptotically stable with asymptotic phase if

$$rac{dw}{dt} = rac{\partial f}{\partial u}^{[2]}(\phi(t))w$$

is uniformly asymptotically stable. This equation is called the second compound differential equation of the linear variational equation

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(\phi(t))v,$$

which reduces to the Liouville equation when n = 2. The uniform asymptotic stability of the second compound differential equation means that, under the flow of the nonlinear equation, 2-dimensional areas near the orbit of $\phi(t)$ diminish exponentially, which is the reason that the characteristic multipliers of the periodic solution $\phi(t)$, with the exception of a single characteristic multiplier which equals one, all have moduli less than 1.

When, with the exception of a single characteristic multiplier which equals 1, the characteristic multipliers have moduli less than 1, the periodic solution is said to be *stable hyperbolic*. It was shown by Muldowney [78] that the periodic solution is stable hyperbolic if and only if the second compound differential equation is uniformly asymptotically stable.

In this chapter the property of the diminishing area being a sufficient condition for a nonconstant periodic solution to be stable hyperbolic is extended to autonomous differential equations in general Banach spaces X with particular applications to reaction diffusion equations. As in the finite dimensional situation, this property is characterized by the asymptotic stability of an associated compound differential equation. This allows the possibility of applying standard asymptotic stability techniques such as Lyapunov theory to the difficult problem of estimation of the characteristic multipliers.

3.1 Orbital Stability of Periodic Solutions

The following notation, definitions, and lemmas are taken from Henry [42], page 53-54, page 197-202 and page 247-261.

Let A be a sectorial operator in a Banach space X and f be continuously differentiable from X^{α} into X where $0 \leq \alpha < 1$. Let $\mathcal{D}(A)$ denote the domain of A. Consider an autonomous differential equation

$$\frac{du}{dt} + Au = f(u), \quad u(\cdot) \in X.$$
(3.1)

In the following, let $u(t) = u(t, u_0)$ denote the solution of (3.1) with initial value $u(0) = u_0$. A nonconstant periodic solution of the autonomous differential equation (3.1) is $u = \phi(t)$ if there exists a least $\omega > 0$ such that $\phi(t + \omega) = \phi(t)$. Let $\gamma = \{\phi(t), 0 \le t \le \omega\}$ denote the orbit of the ω -periodic solution $\phi(t)$.

Definition 3.1. An ω -periodic solution $\phi(t)$ of (3.1) is said to be *orbitally* stable if the orbit γ is stable, that is if for any neighbourhood U of γ , there exists a neighbourhood N of γ such that $u_0 \in N$ implies that the solution of (3.1) $u(t, u_0) \in U$ for all t > 0. An ω -periodic solution $\phi(t)$ is said to be *orbitally asymptotically stable* if the orbit γ is asymptotically stable, that is, the orbit γ is stable and there exists a δ such that $\operatorname{dist}_{X^{\alpha}}(u_0, \phi(0)) < \delta$ implies that $\operatorname{dist}_{X^{\alpha}}(u(t, u_0), \phi(t)) \to 0$ as $t \to \infty$. An ω -periodic solution $\phi(t)$ is said to be *orbitally asymptotically stable with asymptotic phase* if there exist $M, \rho, \beta > 0$ such that

$$\operatorname{dist}_{X^{\alpha}}\{u_0,\gamma\} = \min_{0 \le t \le p} \|u_0 - \phi(t)\|_{\alpha} < \rho$$

implies that there exists a real constant $h = h(u_0)$ such that the solution $u(t, u_0)$ of (3.1) satisfies

$$||u(t, u_0) - \phi(t - h)||_{\alpha} < Me^{-\beta t}, \text{ for } t \ge 0.$$

An ω -periodic solution $\phi(t)$ of (3.1) is orbitally unstable if it is not orbitally stable.

The linear variational equation of (3.1) at the solution $\phi(t)$ is

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(\phi(t))v, \qquad (3.2)$$

whose solutions satisfy

$$v(t) = T(t,s)v(s), \quad t \ge s,$$

where T(t, s) is the evolution operator generated by (3.2).

Definition 3.2. The period map of (3.2) is

$$U(t) = T(t + \omega, t). \tag{3.3}$$

The nonzero eigenvalues of U(t) are called *characteristic multipliers* of the periodic solution $\phi(t)$ or *characteristic multipliers* of the linear variational equation (3.2).

Remark 3.1. For an ordinary differential equation, this map is called the *monodromy matrix*, whose eigenvalues are also the eigenvalues of the Jacobian matrix of the Poincaré map.

Lemma 3.1. $U(t + \omega) = U(t)$ for all t. The characteristic multipliers are independent of t, that is, the nonzero eigenvalues of U(t) coincide with those of U(s). In fact, $\sigma(U(t)) \setminus \{0\}$ is independent of t. If A has compact resolvent, then U(t) is compact and $\sigma(U(t)) \setminus \{0\}$ consists entirely of characteristic multipliers.

Lemma 3.2. Suppose that σ_1 is a spectral subset of $\sigma(U(t))$ for all t; the usual case is when σ_1 is a finite collection of isolated eigenvalues or the complement of such a set. Then for each t, the space X may be decomposed as $X = X_1(t) \oplus X_2(t)$, the direct sum of closed subspaces invariant under U(t), $\sigma(U(t)|_{X_1(t)}) = \sigma_1, \sigma(U(t)|_{X_2(t)}) = \sigma(U(t)) \setminus \sigma_1$. If $t \ge s, T(t, s)$ maps $X_1(s)$ into $X_1(t)$, and is a one-to-one map onto $X_1(t)$ if $0 \notin \sigma_1$.

Let $e^{\beta\omega} = \sup\{|\lambda|, \lambda \in \sigma_1\}$. Then for each $\varepsilon > 0$, there exists $M_{\varepsilon} > 0$ such that

$$||T(t,s)u|| \le M_{\varepsilon} e^{(\beta+\varepsilon)(t-s)} ||u||$$

for $t \geq s$ and $u \in X_1(s)$.

Suppose $0 \notin \sigma_1$ and let $e^{\alpha \omega} = \inf\{|\lambda|, \lambda \in \sigma_1\} > 0$. Then $T(t, s)u, u \in X_1(s)$, may be defined also for $t \leq s$, and for small $\varepsilon > 0$, there exists $M_{\varepsilon} > 0$ such that

$$||T(t,s)u|| \le M_{\varepsilon} e^{(\alpha-\varepsilon)(t-s)} ||u||$$

for $t \leq s$ and $u \in X_1(s)$.

Assume that $\phi(t)$ is a nonconstant ω -periodic solution of (3.1). Let S be a C^1 manifold (in X^{α}) of codimension 1 such that $\phi(0) \in S$ and $\frac{d\phi}{dt}(0)$ is not tangent to S at $\phi(0)$. For $\xi \in S$ near $\phi(t_0)$, define $\Phi(\xi) \in S$ by

$$\Phi(\xi) = u(\omega(\xi), \xi), \tag{3.4}$$

where $\omega(\xi) = \omega + O(||\xi - \phi(0)||_{\alpha})$ is chosen to ensure $\Phi(\xi) \in S$. Let $\Phi'(\xi) = \frac{d\Phi(\xi)}{d\xi}$.

Definition 3.3. The map Φ defined in (3.4) is called the *Poincaré map*.

It is shown in Henry [42], page 258-260, that Φ is a well-defined C^1 function on S near $\phi(0)$ and

$$\Phi'(\phi(0)) = U(0)$$

where T(t, s) is the evolution operator generated by (3.2). Notice that any fixed point of Φ yields an ω -periodic solution of (3.1).

Without loss of generality, assume that $\phi(0) = 0, \frac{d\phi}{dt}(0) \neq 0$ and S is represented near 0 in the form x = h(y) for $y \in Y$ where Y is the tangent space to S at 0. Thus $||h(y) - y||_{\alpha} = o(||y||_{\alpha})$ as $y \to 0$ in Y. Now $X^{\alpha} = \operatorname{span} \left\{ \frac{d\phi}{dt}(0), Y \right\}$. Let $P_Y : X^{\alpha} \to Y$ be the projection onto Y where $P_Y \frac{d\phi}{dt}(0) = 0, P_Y y = y$ for $y \in Y$.

If {1} is an isolated eigenvalue of U(0) and $X^{\alpha} = X_1^{\alpha} \oplus X_2^{\alpha}$ is the corresponding decomposition with

$$\begin{aligned} \sigma(U(0)|_{X_1^{\alpha}}) &= \{1\}, \\ \sigma(U(0)|_{X_2^{\alpha}}) &= \sigma(U(0)) \setminus \{1\}, \end{aligned}$$

then $Y = Y_1 \oplus Y_2$ where $Y_j = P_Y X_j^{\alpha}, j = 1, 2$ are $\Phi'(0)$ -invariant subspaces and $(Z_j(\alpha)) = Z_j(\alpha)$

$$\sigma(\Phi'(0)|_{Y_1}) \subseteq \{1\},$$

$$\sigma(\Phi'(0)|_{Y_2}) = \sigma(U(t_0)) \setminus \{1\}$$

In particular, if $\{1\}$ is an isolated simple eigenvalue of U(0), then $Y_1 = \{0\}$ and $1 \notin \sigma(\Phi'(0))$. If $\{1\}$ is an eigenvalue of multiplicity m, then dim $Y_1 = m - 1$. Suppose that $Y_2 = Y_S \oplus Y_U$ where

$$\sigma(\Phi'(0)|_{Y_S}) = \sigma(U(0)) \cap \{|\lambda| < 1, \lambda \in \sigma(U(0))\},$$

$$\sigma(\Phi'(0)|_{Y_U}) = \sigma(U(0)) \cap \{|\lambda| > 1, \lambda \in \sigma(U(0))\}.$$

Let $e^{\beta\omega} = \sup\{|\lambda|, \lambda \in \sigma(\Phi'(0)|_{Y_S})\}$ and $e^{\alpha\omega} = \inf\{|\lambda|, \lambda \in \sigma(\Phi'(0)|_{Y_U})\}$. Assume that either $\sigma(\Phi'(0)|_{Y_S})$ or $\sigma(\Phi'(0)|_{Y_U})$ is a finite collection of isolated eigenvalues and $\beta < 0, \alpha > 0$. Then Lemma 3.2 implies that there exist constants M, N > 0 such that

$$||T(t,0)u_0||_{\alpha} \le M e^{\frac{D}{2}t} ||u_0||_{\alpha}, \text{ for } t \ge 0 \text{ and } u_0 \in Y_S,$$

and $T(t, s)u, u \in Y_U$, may be defined also for $t \leq s$, and

$$||T(t,0)u_0||_{\alpha} \le N e^{\frac{\alpha}{2}t} ||u_0||_{\alpha}, \text{ for } t \le 0 \text{ and } u_0 \in Y_U.$$

Definition 3.4. The dimension of Y_S is called the *dimension of the (local)* stable manifold of $\phi(t)$. The dimension of Y_U is called the *dimension of* the (local) unstable manifold of $\phi(t)$. The dimension of span{ $\phi'(0), Y_1$ } is called the *dimension of the (local) center manifold* of $\phi(t)$. The sum of the dimension of the (local) center manifold and the dimension of the (local) unstable manifold is called the *codimension of the (local) stable manifold* of $\phi(t)$.

Lemma 3.3. If A has compact resolvent, then the zero solution of (3.2) is uniformly asymptotically stable if and only if all nonzero eigenvalues of U(t)have moduli less than 1.

Lemma 3.4. Let f be continuously differentiable from X^{α} into X in a neighbourhood of a nonconstant periodic solution $\phi(t)$ of (3.1). Suppose that $t \mapsto \frac{\partial f}{\partial x}(\phi(t)) \in \mathscr{L}(X^{\alpha}, X)$ is Hölder continuous. Then the linear variational equation $\frac{\partial v}{\partial x} = \frac{\partial f}{\partial x}(\psi(t))$

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(\phi(t))v$$

has 1 as a characteristic multiplier.

Lemma 3.5. Suppose that the assumptions in Lemma 3.4 are satisfied. If $\sigma(U(t)) \cap \{|\lambda| > 1\}$ is a nonempty spectral set, then $\phi(t)$ is orbitally unstable.

Lemma 3.6. Suppose that the assumptions in Lemma 3.4 are satisfied. If 1 is an isolated simple characteristic multiplier of the periodic solution $\phi(t)$, and the remainder of the spectrum lies in $\{|\lambda| < e^{-\beta\omega}\}$ for some $\beta > 0$, then $\phi(t)$ is orbitally asymptotically stable with asymptotic phase.

These theoretical results from Henry [42] furnish a context in which the compound differential equation approach to orbital stability may be developed. The next theorem uses the compound differential equation introduced in Chapter 1 to provide a sufficient condition for the remainder of the spectrum of the period map lying in $\{|\lambda| < e^{-\beta\omega}\}$ for some $\beta > 0$.

Theorem 3.7. Suppose that $\phi(t)$ is a nonconstant ω -periodic solution of (3.1). If the second compound differential equation

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f}{\partial u}^{[2]}(\phi(t))w$$

of the linear variational equation (3.2) is uniformly asymptotically stable in $\bigwedge^2 X^{\alpha}$, then $\phi(t)$ is orbitally asymptotically stable with asymptotic phase.

Theorem 3.7 will follow from Theorem 4.12 in Section 4.2. No proof is given here. A particular case when A has compact resolvent is discussed in detail as follows.

Theorem 3.8. Suppose that A has compact resolvent and $\phi(t)$ is a nonconstant ω -periodic solution of (3.1). If the kth compound differential equation

$$\frac{dw}{dt} + A^{[k]}w = \frac{\partial f^{[k]}}{\partial u}(\phi(t))w$$
(3.5)

of the linear variational equation (3.2) is uniformly asymptotically stable in $\bigwedge^k X^{\alpha}$, then at most k-2 characteristic multipliers of the periodic solution $\phi(t)$ have moduli greater than or equal to 1.

Proof. Let T(t,s) be the evolution operator generated by the linear variational equation

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(\phi(t))v.$$

Let $U(t) = T(t + \omega, t)$ denote the period map. Since $\phi(t)$ is a nonconstant ω -periodic solution of (3.1), Lemma 3.4 implies that one of the characteristic multipliers of the periodic solution $\phi(t)$ is 1. From Section 1.3, $w(t) = T^{(k)}(t,s)w(s)$ is a solution of (3.5), and thus the period map of (3.5) is $U^{(k)}(t) = T^{(k)}(t + \omega, t)$. Since (3.5) is uniformly asymptotically stable and A has compact resolvent, Lemma 3.3 implies that the nonzero eigenvalues of $U^{(k)}(t)$ have moduli less than 1. Let $\{\lambda_i\}$ be the eigenvalues, counting multiplicities, of U(t). Then all possible products of k eigenvalues of U(t) in the form $\lambda_{i_1}\lambda_{i_2}\cdots\lambda_{i_k}, i_1 < i_2 < \cdots < i_k$, counting multiplicities, are the

eigenvalues of $U^{(k)}(t)$. Since one of the characteristic multipliers of $\phi(t)$ is 1, it can be concluded that at most k-2 characteristic multipliers of $\phi(t)$ have moduli greater than or equal to 1.

Definition 3.5. A periodic solution is *stable hyperbolic* if, with the exception of a single characteristic multiplier which equals 1, the moduli of the characteristic multipliers of the periodic solution are less than 1.

Corollary 3.9. Suppose that A has compact resolvent and $\phi(t)$ is a nonconstant periodic solution of (3.1). If $\phi(t)$ is stable hyperbolic, then $\phi(t)$ is orbitally asymptotically stable with asymptotic phase.

Corollary 3.10. Suppose that A has compact resolvent and $\phi(t)$ is a nonconstant ω -periodic solution of (3.1). Then $\phi(t)$ is stable hyperbolic if the second compound differential equation

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f^{[2]}}{\partial u}(\phi(t))w$$
(3.6)

of the linear variational equation (3.2) is uniformly asymptotically stable in $\bigwedge^2 X^{\alpha}$.

Remark 3.2. Let U(t) be the periodic map of the linear variational equation of (3.1) at the solution $\phi(t)$

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(\phi(t))v.$$
(3.7)

Suppose that $\{\lambda_i\}$ are the eigenvalues, counting multiplicities, of U(t). If the corresponding eigenfunctions of U(t) form a basis of X, then the eigenvalues of $U^{(k)}(t)$, counting multiplicities, are given by all possible products of k eigenvalues of U(t) in the form $\lambda_{i_1}\lambda_{i_2}\cdots\lambda_{i_k}$, $i_1 < i_2 < \cdots < i_k$. In this case, the uniform asymptotic stability of the second compound differential equation (3.5) is equivalent to the hyperbolic stability of the periodic solution $\phi(t)$.

Remark 3.3. The finite linear combinations of expression of the form $v^1 \wedge v^2$ are solutions of the second compound differential equation (3.6), where $v^i, i = 1, 2$ are solutions of the linear variational equation (3.7) of (3.1) at the solution $\phi(t)$. If the second compound differential equation (3.6) is uniformly asymptotically stable in $\bigwedge^2 X^{\alpha}$, then there exists a $\sigma > 0$, such that

$$\|(v^{1} \wedge v^{2})(w)\|_{\bigwedge X^{\alpha}} \le e^{-\sigma w} \|(v^{1} \wedge v^{2})(0)\|_{\bigwedge X^{\alpha}}.$$
(3.8)

Hence

$$\|U^{(2)}(0)(v^1 \wedge v^2)(0)\|_{\bigwedge X^{\alpha}} < \|(v^1 \wedge v^2)(0)\|_{\bigwedge X^{\alpha}},$$

where $U^{(2)}(t)$ is defined in the proof of Theorem 3.8. Let $\{\lambda_i\}$ be the eigenvalues, counting multiplicities, of U(0). Then all possible products of two eigenvalues of U(0) in the form $\lambda_{i_1}\lambda_{i_2}$, $i_1 < i_2$, counting multiplicities, are the eigenvalues of $U^{(2)}(0)$. Since one of the characteristic multipliers of $\phi(t)$ is 1, it can be concluded from that (3.8) that all the other characteristic multipliers of $\phi(t)$ have moduli less than 1 and thus $\phi(t)$ is stable hyperbolic.

3.2 Reaction Diffusion Systems

In this section, Corollary 3.10 is applied to study the stability of periodic solutions of reaction diffusion systems.

Let $\Omega \subset \mathbb{R}^m$ be a bounded domain with smooth boundary and $D = \text{diag}(d_1, \dots, d_n), d_i \geq 0$. Let $f : \Omega \times \mathbb{R}^n \to \mathbb{R}^n$ be locally Lipschitz continuous in u, uniformly in x. Suppose that $\phi(t, x)$ is an ω -periodic solution of

$$\frac{\partial u}{\partial t} = D\Delta u + f(x, u),$$
(3.9)

with Dirichlet boundary condition

$$u|_{\partial\Omega} = 0, \tag{3.10}$$

or Neumann boundary condition

$$\left. \frac{\partial u}{\partial n} \right|_{\partial \Omega} = 0, \tag{3.11}$$

or a more general boundary condition of the form

$$Bu|_{\partial\Omega} = 0. \tag{3.12}$$

In the following, the case with Dirichlet boundary condition is discussed and similar arguments can be applied to other boundary conditions. Let $X = L^2(\Omega, \mathbb{R}^n)$ and $A : \mathcal{D}(A) \subset X \to X$ be the linear unbounded operator $Au = -\Delta u$, where

$$\mathcal{D}(A) = \left\{ u \in H^2(\Omega, \mathbb{R}^n), \, \, u|_{\partial\Omega} = 0
ight\}.$$

The initial value problem of (3.9) is well-posed in X^{α} , $0 \leq \alpha < 1$. The linear variational equation of (3.9) at $u = \phi(t, x)$ is

$$\begin{aligned} \frac{\partial v}{\partial t} &= D\Delta v + \frac{\partial f}{\partial u}(x, \phi(t, x))v, \\ v|_{\partial\Omega} &= 0. \end{aligned}$$
(3.13)

Let $\lambda_i, i = 1, 2, \cdots$ be the eigenvalues of the Laplace equation

$$\Delta u = -\lambda_i u, \tag{3.14}$$

with Dirichlet boundary condition (3.10) and e^1, e^2, \cdots be its corresponding orthonormal eigenfunctions. It is assumed that $0 < \lambda_1 \leq \lambda_2 \leq \cdots$.

Remark 3.4. If $\phi(t, x)$ is an ω -periodic solution of (3.9) with Neumann boundary condition, then $0 = \lambda_1 < \lambda_2 \leq \lambda_3 \leq \cdots$ is assumed. For other boundary conditions, λ_i are the eigenvalues of the Laplace equation (3.14) and $\lambda_1 \leq \lambda_2 \leq \lambda_3 \leq \cdots$ is assumed.

Let $v^i = (v_1^i, v_2^i, \dots, v_n^i)^T \in X^{\alpha}, i = 1, 2$ be solutions of (3.13) and $f(x, u) = (f_1(x, u), f_2(x, u), \dots, f_n(x, u))^T$. In the following, two representations of the second compound differential equation of (3.13) are discussed.

Case 1: A pointwise representation of the second compound differential equation of (3.13)

For $u \in X^{\alpha}$, $x \in \Omega$, $i = 1, \dots, n$, define a linear functional $(i, x) : u \mapsto \langle (i, x), u \rangle = u_i(x)$. Then (3.13) can be written as

$$\frac{\partial v_j}{\partial t} = d_j \Delta v_j + \sum_{k=1}^n a_{jk}(t, x) v_k,$$

where

$$a_{jk}(t,x) = \frac{\partial f_j}{\partial u_k}(x,\phi(t,x)).$$

. .

For $w(t) \in \bigwedge^2 X^{\alpha}$, $x_1, x_2 \in \Omega$, $i, j = 1, \dots, n$, a pointwise representation of w(t) is

$$\langle (i, x_1) \wedge (j, x_2), w(t) \rangle := w_{ij}(t, x_1, x_2)$$

and a pointwise representation of the operator $\left(D\Delta + \frac{\partial f}{\partial u} \right)^{[2]}$ is
 $\left\langle (i, x_1) \wedge (j, x_2), \left(D\Delta + \frac{\partial f}{\partial u} \right)^{[2]} w \right\rangle$

$$= d_i \Delta_1 w_{ij}(t, x_1, x_2) + d_j \Delta_2 w_{ij} + (a_{ii}(t, x_1) + a_{jj}(t, x_2)) w_{ij}(t, x_1, x_2) \\ + \sum_{k \neq i} a_{ik}(t, x_1) w_{kj} + \sum_{k \neq j} a_{jk}(t, x_2) w_{ik},$$

where $\Delta_1 w_{ij}$ and $\Delta_2 w_{ij}$ are the Laplacian of $w_{ij}(x_1, x_2)$ with respect to x_1 and x_2 . Since w_{ij} satisfies

$$w_{ij}(t, x_1, x_2) = -w_{ji}(t, x_2, x_1),$$

there are $k + (k-1) + (k-2) + \dots + 1 = \binom{k+1}{2}$ different types of determinants that will need to be discussed. Thus a pointwise representation of the second compound differential equation of (3.13) in $\bigwedge^2 X^{\alpha}$ is

$$\frac{\partial w_{ii}}{\partial t} = d_i (\Delta_1 w_{ii} + \Delta_2 w_{ii}) + (a_{ii}(t, x_1) + a_{ii}(t, x_2)) w_{ii}
+ \sum_{k \neq i} (a_{ik}(t, x_1) w_{ki} + a_{ik}(t, x_2) w_{ik}), \quad i = 1, 2, \cdots n,
\frac{\partial w_{ij}}{\partial w_{ij}} = d \Delta_1 w_{ij} + d \Delta_2 w_{ij} + (a_i(t, x_1) + a_i(t, x_2)) w_{ij}$$
(3.15)

$$\frac{w_{ij}}{\partial t} = d_i \Delta_1 w_{ij} + d_j \Delta_2 w_{ij} + (a_{ii}(t, x_1) + a_{jj}(t, x_2)) w_{ij}$$

$$+ \sum_{k \neq i} a_{ik}(t, x_1) w_{kj} + \sum_{k \neq j} a_{jk}(t, x_2) w_{ik}, \quad 1 \le i < j \le n.$$

If $v^1(t), v^2(t)$ are solutions of (3.13), then

$$\begin{split} w_{ij}(t, x_1, x_2) &= \langle (i, x_1) \land (j, x_2), (v^1 \land v^2)(t) \rangle \\ &= \det \begin{bmatrix} v_i^1(t, x_1) & v_i^2(t, x_1) \\ v_j^1(t, x_2) & v_j^2(t, x_2) \end{bmatrix}, \quad x_1, x_2 \in \Omega, \quad i, j = 1, 2, \cdots, n. \end{split}$$

It can be checked by differentiating the above determinant that $v^1(t) \wedge v^2(t)$ satisfies the second compound differential equation (3.15) and the Dirichlet boundary conditions on v^i imply that

$$w_{ij}(t, x_1, x_2)|_{x_1 \in \partial \Omega} = 0, \quad w_{ij}(t, x_1, x_2)|_{x_2 \in \partial \Omega} = 0.$$

Case 2: A basis representation of the second compound differential equation of (3.13)

Let $\{e^i(x)\}$ be the orthonormal eigenfunctions of the Laplace equations (3.14) with Dirichlet boundary condition (3.10) and $\lambda_i, i = 1, 2, \cdots$ be the corresponding eigenvalues. Then the set of vectors $\{e^i\}$ is an orthonormal basis of X^{α} and $\{e^i \wedge e^j, 1 \leq i < j\}$ is an orthonormal basis of $\bigwedge^2 X^{\alpha}$. Define another linear map $(i, e^j) : v = (v_1, v_2, \cdots, v_n) \mapsto \langle (i, e^j), v \rangle = \int_{\Omega} v^i e^j$. Then

$$v(t,x) = (v_1, v_2, \cdots, v_n) = \left(\sum_j c_{1j}(t)e^j(x), \cdots, \sum_j c_{nj}(t)e^j(x)\right)^T$$

where $c_{ij} := \langle (i, e^j), v \rangle$, $i = 1, 2, \cdots, n, j = 1, 2, \cdots$ satisfies

$$\frac{dc_{ij}}{dt} = [-d_i\lambda_j + a_{ii}(t,j,j)]c_{ij} + \sum_{k \neq j} a_{ii}(t,j,k)c_{ik} + \sum_{l \neq i} \sum_k a_{il}(t,j,k)c_{lk}, \quad (3.16)$$

where $a_{ij}(t,k,l) := \langle e^k, a_{ij}e^l \rangle = \int_{\Omega} a_{ij}(t,x)e^l(x)e^k(x)dx, \ i,j = 1, 2, \cdots, n, k, l = 1, 2, \cdots$. Here $a_{ij}(t,k,l) = a_{ij}(t,l,k)$.

For $w(t) \in \bigwedge^2 X^{\alpha}$, a basis representation of w(t) is

$$\left\langle (i, e^{i_1}) \land (j, e^{i_2}), w(t) \right\rangle := w_{ij}(t, i_1, i_2)$$

and a basis representation of the second compound differential equation of (3.13) in $\bigwedge^2 X^{\alpha}$ is

$$\frac{dw_{ii}}{dt}(t,i_{1},i_{2}) = (-d_{i}\lambda_{i_{1}} - d_{i}\lambda_{i_{2}} + a_{ii}(t,i_{1},i_{1}) + a_{ii}(t,i_{2},i_{2}))w_{ii}(t,i_{1},i_{2}) \\
+ \sum_{k \neq i_{1}} a_{ii}(t,k,i_{1})w_{ii}(t,k,i_{2}) + \sum_{l \neq i} \sum_{k} a_{il}(t,k,i_{1})w_{li}(t,k,i_{2}) \\
+ \sum_{k \neq i_{1}} a_{ii}(t,k,i_{2})w_{ii}(t,i_{1},k) + \sum_{l \neq i} \sum_{k} a_{il}(t,k,i_{2})w_{il}(t,i_{1},k) \\
\frac{dw_{ij}}{dt}(t,i_{1},i_{2}) = (-d_{i}\lambda_{i_{1}} - d_{j}\lambda_{i_{2}} + a_{ii}(t,i_{1},i_{1}) + a_{jj}(t,i_{2},i_{2}))w_{ij}(t,i_{1},i_{2}) \\
+ \sum_{k \neq i} a_{ii}(t,k,i_{1})w_{ij}(t,k,i_{2}) + \sum_{l \neq i} \sum_{k} a_{il}(t,k,i_{1})w_{ll}(t,k,i_{2}) \\
+ \sum_{k \neq i} a_{jj}(t,k,i_{2})w_{ij}(t,i_{1},k) + \sum_{l \neq i} \sum_{k} a_{jl}(t,k,i_{2})w_{ll}(t,i_{1},k).$$
(3.17)

A direct application of Corollary 3.10 implies the following corollary.

Corollary 3.11. If the second compound differential equation (3.15) or (3.17) of the linear variational equation (3.13) is uniformly asymptotically stable in $\bigwedge^2 X^{\alpha}$, then $\phi(t, x)$ is orbitally asymptotically stable with asymptotic phase in X^{α} .

In the following, the two cases when $\Omega = (0, L)$ are discussed in detail.

Scalar Case: n = 1 with periodic boundary condition

Suppose that $u = \phi(t, x)$ is a nonconstant ω -periodic solution of

$$u_t = u_{xx} + f(x, u), \quad 0 < x < L, \quad t > 0,$$

$$u(t, 0) = u(t, L), \quad u_x(t, 0) = u_x(t, L) \quad t \ge 0,$$

(3.18)

where $f \in C^2((0, L) \times \mathbb{R} \to \mathbb{R})$. The linear variational equation of (3.18) at $\phi(t, x)$ is

$$v_t = v_{xx} + a(t, x)v, \quad 0 < x < L, \quad t > 0,$$

$$v(t, 0) = v(t, L), \quad v_x(t, 0) = v_x(t, L) \quad t \ge 0,$$
(3.19)

where

$$a(t,x)=rac{\partial f}{\partial u}(x,\phi(t,x)).$$

If v^1, v^2 are solutions of (3.19), then $w(t) = (v^1 \wedge v^2)(t) \in \bigwedge^2 X$ has a pointwise representation

$$w(t, x_1, x_2) = \left\langle x_1 \wedge x_2, (v^1 \wedge v^2)(t) \right\rangle = \det \left[\begin{array}{cc} v^1(t, x_1) & v^2(t, x_1) \\ v^1(t, x_2) & v^2(t, x_2) \end{array} \right]$$

which satisfies the second compound differential equation of (3.19) defined on $\bigwedge^2 X$

$$w_t = w_{x_1x_1} + w_{x_2x_2} + (a(t, x_1) + a(t, x_2))w.$$
(3.20)

The periodic boundary conditions of v^i implies that

$$w(t,0,x_2) = w(t,L,x_2), \quad w_{x_1}(t,0,x_2) = w_{x_1}(t,L,x_2),$$

$$w(t,x_1,0) = w(t,x_1,L), \quad w_{x_2}(t,x_1,0) = w_{x_2}(t,x_1,L).$$

The eigenvalues and orthonormal eigenfunctions of the Laplace equation

$$\Delta u = -\lambda_i u,
u(0) = u(L), \quad u_x(0) = u_x(L)$$
(3.21)

,

are $\lambda_1 = 0, \lambda_{2n} = \left(\frac{n\pi}{L}\right)^2, \lambda_{2n+1} = \left(\frac{n\pi}{L}\right)^2$, and $e^1(x) = 1, e^{2n}(x) = \sqrt{\frac{2}{L}} \cos\left(\frac{n\pi}{L}\right) x, e^{2n+1}(x) = \sqrt{\frac{2}{L}} \sin\left(\frac{n\pi}{L}\right) x, n = 1, 2, \cdots$, respectively. Consider a Lyapunov function

$$V(t) = \frac{1}{2} \int_{(0,L)^2} w^2 = \frac{1}{2} \int_{(0,L)^2} w^2(t,x_1,x_2) \, dx_1 dx_2.$$

Then

$$\begin{aligned} \frac{d^+V}{dt} &= \int_{(0,L)^2} ww_t \\ &= \int_{(0,L)^2} ww_{x_1x_1} + ww_{x_2x_2} + \int_{(0,L)^2} \left(a(t,x_1) + a(t,x_2)\right) w^2 \\ &= ww_{x_1} |_{x_1=0}^L + ww_{x_2} |_{x_2=0}^L - \int_{(0,L)^2} \left[(w_{x_1})^2 + (w_{x_2})^2 \right] \\ &+ \int_{(0,L)^2} \left(a(t,x_1) + a(t,x_2)\right) w^2 \\ &= -\int_{(0,L)^2} \left[(w_{x_1})^2 + (w_{x_2})^2 \right] + \int_{(0,L)^2} \left(a(t,x_1) + a(t,x_2)\right) w^2 \\ &\leq \int_{(0,L)^2} \left(-\lambda_1 - \lambda_2 \right) w^2 + \int_{(0,L)^2} \left(a(t,x_1) + a(t,x_2)\right) w^2 \\ &= \int_{(0,L)^2} \left(-\left(\frac{\pi}{L}\right)^2 + 2 \max_{0 < x < L} \left\{a(t,x)\right\} \right) w^2. \end{aligned}$$

Corollary 3.12. If

$$\int_0^{\omega} \max_{0 < x < L} \left\{ \frac{\partial f}{\partial u}(x, \phi(t, x)) \right\} dt < -\frac{\pi^2}{2L^2},$$

then $\phi(t,x)$ is orbitally asymptotically stable with asymptotic phase in $L^2(0,L)$.

Suppose that $v(t,x) = \sum_{j} c_{j}(t)e^{j}(x)$, where $c_{j} = \langle e^{j}, v \rangle = \int_{0}^{L} ve^{j}, j = 1, 2, \cdots$. Then

$$\frac{dc_j}{dt} = [-\lambda_j + a(t, j, j)]c_j + \sum_{k \neq j} a_j(t, j, k)c_k,$$
(3.22)

where $a(t,k,l) := \langle e^k, ae^l \rangle = \int_0^L a(t,x)e^l(x)e^k(x)dx$, $k,l = 1,2,\cdots$. If v^1, v^2 are solutions of (3.19), then $w(t) = (v^1 \wedge v^2)(t) \in \bigwedge^2 X$ has a basis

representation

$$w(t, i_1, i_2) = \langle e^{i_1} \wedge e^{i_2}, w(t) \rangle = \det \begin{bmatrix} c_{i_1}^1 & c_{i_1}^2 \\ c_{i_2}^1 & c_{i_2}^2 \end{bmatrix}, \text{ for } 1 \le i_1 < i_2,$$

which satisfies the second compound differential equation of (3.19)

$$\frac{dw(t, i_1, i_2)}{dt} = (-\lambda_{i_1} - \lambda_{i_2} + a(t, i_1, i_1) + a(t, i_2, i_2))w(t, i_1, i_2)
+ \sum_{k \neq i_1, i_2} a(t, i_1, k)w(t, k, i_2) + \sum_{k \neq i_1, i_2} a(t, k, i_2)w(t, i_1, k).$$
(3.23)

Consider a Lyapunov function

$$V(t) = \sum_{1 \le i_1 < i_2} w^2(t, i_1, i_2).$$

Then

$$\frac{d^+V}{dt} \leq \sum_{1 \leq i_1 < i_2} \mu_{i_1 i_2}(t) w^2(t, i_1, i_2)$$

$$\leq \mu(t) V,$$

where

$$\mu_{i_{1}i_{2}}(t) = -\lambda_{i_{1}} - \lambda_{i_{2}} + a(t, i_{1}, i_{1}) + a(t, i_{2}, i_{2}) + \sum_{k \neq i_{1}, i_{2}} (|a(t, i_{1}, k)| + |a(t, k, i_{2})|),$$

$$a(t, x) = \frac{\partial f}{\partial u}(x, \phi(t, x)), \quad a(t, k, l) = \int_{0}^{L} a(t, x)e^{l}(x)e^{k}(x)dx, \quad k, l = 1, 2, \cdots,$$

$$\mu(t) = \sup_{1 \leq i_{1} < i_{2}} \{\mu_{i_{1}i_{2}}(t)\}. \quad (3.24)$$

Corollary 3.13. If

$$\int_0^\omega \mu < 0,$$

where $\mu(\cdot)$ is given by (3.24), then $\phi(t, x)$ is orbitally asymptotically stable with asymptotic phase in $L^2(0, L)$.

Planar Case: n = 2 with Dirichlet boundary condition

Suppose that $u = \phi(t, x) = (\phi_1(t, x), \phi_2(t, x))$ is a nonconstant ω -periodic solution of

$$u_t = d_1 u_{xx} + f(x, u, v), \quad v_t = d_2 v_{xx} + g(x, u, v), \quad 0 < x < L, \quad t > 0,$$

$$u(t, 0) = u(t, L) = 0, \quad v(t, 0) = v(t, L) = 0, \quad t \ge 0,$$

(3.25)

where
$$f, g \in C^2((0, L) \times \mathbb{R}^2 \to \mathbb{R}^2)$$
. Let

$$\begin{bmatrix} a_{11}(t,x) & a_{12}(t,x) \\ a_{21}(t,x) & a_{22}(t,x) \end{bmatrix} := \begin{bmatrix} \frac{\partial f}{\partial u}(x,\phi_1(t,x),\phi_2(t,x)) & \frac{\partial f}{\partial v}(x,\phi_1(t,x),\phi_2(t,x)) \\ \frac{\partial g}{\partial u}(x,\phi_1(t,x),\phi_2(t,x)) & \frac{\partial g}{\partial v}(x,\phi_1(t,x),\phi_2(t,x)) \end{bmatrix}$$

The previous discussion implies that a pointwise representation of the second compound differential equation defined on $\bigwedge^2 X^\alpha$ is

$$\frac{\partial w_{11}}{\partial t} = d_1(\Delta_1 w_{11} + \Delta_2 w_{11}) + (a_{11}(t, x_1) + a_{11}(t, x_2))w_{11} \\
+ a_{12}(t, x_1)w_{21} + a_{12}(t, x_2)w_{12}, \\
\frac{\partial w_{22}}{\partial t} = d_2(\Delta_1 w_{22} + \Delta_2 w_{22}) + (a_{22}(t, x_1) + a_{22}(t, x_2))w_{22} \\
+ a_{21}(t, x_1)w_{12} + a_{21}(t, x_2)w_{21}, \\
\frac{\partial w_{12}}{\partial t} = d_1\Delta_1 w_{12} + d_2\Delta_2 w_{12} + (a_{11}(t, x_1) + a_{22}(t, x_2))w_{12} \\
+ a_{12}(t, x_1)w_{22} + a_{21}(t, x_2)w_{11}.$$
(3.26)

Let

$$\begin{split} V_1(t) &= \frac{1}{2} \int_{(0,L)^2} (w_{11})^2 = \frac{1}{2} \int_{(0,L)^2} (w_{11}(t,x_1,x_2))^2 dx_1 dx_2, \\ V_2(t) &= \frac{1}{2} \int_{(0,L)^2} (w_{22})^2 = \frac{1}{2} \int_{(0,L)^2} (w_{22}(t,x_1,x_2))^2 dx_1 dx_2, \\ V_3(t) &= \frac{1}{2} \int_{(0,L)^2} (w_{12})^2 = \frac{1}{2} \int_{(0,L)^2} (w_{12}(t,x_1,x_2))^2 dx_1 dx_2. \end{split}$$

Then

$$\begin{split} & \frac{1}{2} \frac{d^+ V_1}{dt} \\ &= \int_{(0,L)^2} d_1 (\Delta_1 w_{11} + \Delta_2 w_{11}) w_{11} + \int_{(0,L)^2} (a_{11}(t,x_1) + a_{11}(t,x_2)) (w_{11})^2 \\ &+ \int_{(0,L)^2} \left[a_{12}(t,x_1) w_{21} w_{11} + a_{12}(t,x_2) w_{12} w_{11} \right] \\ &\leq \int_{(0,L)^2} d_1 (-\lambda_1 - \lambda_2 + a_{11}(t,x_1) + a_{11}(t,x_2)) (w_{11})^2 \\ &+ \int_{(0,L)^2} \left[a_{12}(t,x_1) w_{21} w_{11} + a_{12}(t,x_2) w_{12} w_{11} \right]. \end{split}$$

•

It follows from

$$\int_{(0,L)^2} a_{12}(t,x_1)w_{21}(t,x_1,x_2)w_{11}(t,x_1,x_2)dx_1dx_2$$

=
$$\int_{(0,L)^2} a_{12}(t,x_2)w_{21}(t,x_2,x_1)w_{11}(t,x_2,x_1)dx_1dx_2$$

=
$$\int_{(0,L)^2} a_{12}(t,x_2)w_{12}(t,x_1,x_2)w_{11}(t,x_1,x_2)dx_1dx_2$$

that

$$\frac{1}{2} \frac{d^+ V_1}{dt} \leq \int_{(0,L)^2} d_1 \left[-\lambda_1 - \lambda_2 + a_{11}(t,x_1) + a_{11}(t,x_2) \right] (w_{11})^2 \\
+ 2 \int_{(0,L)^2} a_{12}(t,x_2) w_{12} w_{11}.$$

Similarly,

$$\begin{split} \frac{1}{2} \frac{dV_2}{dt} &\leq \int_{(0,L)^2} d_2 \left[-\lambda_1 - \lambda_2 + a_{22}(t,x_1) + a_{22}(t,x_2) \right] (w_{22})^2 \\ &\quad + 2 \int_{(0,L)^2} a_{21}(t,x_1) w_{12} w_{22} \\ \frac{1}{2} \frac{dV_3}{dt} &\leq \int_{(0,L)^2} \left[-\lambda_1 (d_1 + d_2) + a_{11}(t,x_1) + a_{22}(t,x_2) \right] (w_{12})^2 \\ &\quad + \int_{(0,L)^2} \left[a_{12}(t,x_1) w_{22} w_{12} + a_{21}(t,x_2) w_{11} w_{12} \right]. \end{split}$$

Consider a Lyapunov function

$$V(t) = \frac{1}{2!}(V_1 + V_2 + 2V_3).$$

It follows from Young's inequality

$$u(x)v(x) \le \frac{1}{2} \left[\nu(x)u^2(x) + \frac{v^2(x)}{\nu(x)} \right], \quad \text{for every positive function } \nu(x),$$
(3.27)

that for any positive functions $\nu_1(x), \nu_2(x),$

$$\frac{d^{+}V}{dt} \leq \int_{(0,L)^{2}} \left[\mu_{1}(t)(w_{11})^{2} + \mu_{2}(t)(w_{22})^{2} + 2\mu_{3}(t)(w_{12})^{2} \right] \\
\leq 2\mu(t)V,$$
(3.28)
where

$$\mu_{1}(t) = -(\lambda_{1} + \lambda_{2}) d_{1} + \max_{x} \left\{ 2a_{11}(t, x) + \frac{1}{\nu_{1}(x)} |a_{12}(t, x) + a_{21}(t, x)| \right\},$$

$$\mu_{2}(t) = -(\lambda_{1} + \lambda_{2}) d_{2} + \max_{x} \left\{ 2a_{22}(t, x) + \frac{1}{\nu_{2}(x)} |a_{12}(t, x) + a_{21}(t, x)| \right\},$$

$$\mu_{3}(t) = -\lambda_{1}(d_{1} + d_{2}) + \max_{x} \left\{ a_{11}(t, x) + \frac{\nu_{2}(x)}{2} |a_{12}(t, x) + a_{21}(t, x)| \right\}$$

$$+ \max_{x} \left\{ a_{22}(t, x) + \frac{\nu_{1}(x)}{2} |a_{12}(t, x) + a_{21}(t, x)| \right\},$$

$$\mu(t) = \max\{\mu_{1}(t), \mu_{2}(t), \mu_{3}(t)\}$$

$$(3.29)$$

and the maximum is taken over 0 < x < L.

Corollary 3.14. If

$$\int_0^\omega \mu < 0, \tag{3.30}$$

where $\mu(\cdot)$ is defined by (3.29), then $\phi(t, x)$ is orbitally asymptotically stable with asymptotic phase in $L^2((0, L), \mathbb{R}^2)$.

Remark 3.5. Corollary 3.14 is used in Section 3.3 to give a stability condition for a periodic solution of a system arising from an ordinary differential equation.

3.3 Effect of Diffusion

In this section the effect of adding diffusion to the dynamics of an ordinary differential equation in \mathbb{R}^n on the stability of a periodic solution is studied. It has been shown by Henry [42], page 201-202, that for 2-dimensional systems, the orbital stability is preserved under certain circumstances. But he proved that, in general, stability is not preserved for general diffusion terms. Leiva [50] extended Henry's sufficient condition on the preservation of orbital stability to systems in \mathbb{R}^n . Here the necessary and sufficient condition for hyperbolic stability is formulated in terms of stability of an infinite dimensional system of uncoupled ordinary differential equations, which includes both conditions of [42] and [50] with weaker restrictions on the added diffusion.

The following necessary and sufficient condition for a periodic solution of an ordinary differential equation to be hyperbolic stable is from Muldowney [78]. Let $u = \phi(t)$ be a nonconstant ω -periodic solution of

$$\frac{du}{dt} = f(u), \quad f \in C^1(\mathbb{R}^n, \mathbb{R}^n).$$
(3.31)

The linear variational equation at $u = \phi(t)$ is

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(\phi(t))v.$$

Theorem 3.15. The periodic solution $\phi(t)$ of (3.31) is stable hyperbolic if and only if the second compound differential equation

$$\frac{dw}{dt} = \frac{\partial f^{[2]}}{\partial u}(\phi(t))w$$

is uniformly asymptotically stable in $\bigwedge^2 \mathbb{R}^n (\cong \mathbb{R}^{\binom{n}{2}})$.

Corollary 3.16. The periodic solution $\phi(t)$ of (3.31) is stable hyperbolic if, for some Lozinskii measure μ ,

$$\int_0^\omega \mu\left(\frac{\partial f^{[2]}}{\partial u}(\phi(t))\right) < 0.$$

Theorem 3.15 is the special case of Theorem 3.7 when $X = \mathbb{R}^n$ and A = 0. When n = 2, it is the Poncaré stability criterion which asserts that a nonconstant ω -periodic solution $u = \phi(t)$ of

$$rac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^2,$$

is orbitally asymptotically stable with asymptotic phase if

$$\int_0^{\omega} \operatorname{div} \, f(\phi(t)) < 0$$

Suppose that a diffusion term is added to (3.31) as follows. Let $\Omega \subset \mathbb{R}^m$ be a bounded domain with smooth boundary and $D = \text{diag}(d_1, d_2, \dots, d_n), d_i \geq$ 0. Then $u = \phi(t)$ is also a periodic solution of the reaction diffusion equation

$$\begin{aligned} \frac{\partial u}{\partial t} &= D\Delta u + f(u),\\ \frac{\partial u}{\partial n}\Big|_{\partial\Omega} &= 0. \end{aligned} \tag{3.32}$$

Let $X = L^2(\Omega, \mathbb{R}^n)$ and $A : \mathcal{D}(A) \subset X \to X$ be the linear unbounded operator $Au = -\Delta u$, where

$$\mathcal{D}(A) = \left\{ u \in H^2(\Omega, \mathbb{R}^n), \left. \frac{\partial u}{\partial n} \right|_{\partial \Omega} = 0 \right\}.$$

Theorem 1.6.1 in Henry [42], page 39, implies that if $\frac{m}{4} < \alpha \leq 1$, then

$$X^{\alpha} \subset C(\Omega, \mathbb{R}^n), \text{ and } X^{\alpha} \subset L^p(\Omega, \mathbb{R}^n), \ p \geq 2.$$

The initial value problem of (3.32) is well-posed in X^{α} . The linear variational equation of (3.32) at $u = \phi(t)$ is

$$\frac{\partial v}{\partial t} = D\Delta v + \frac{\partial f}{\partial u}(\phi(t))v.$$
(3.33)

The eigenvalues and orthonormal eigenfunctions of the Laplace equation

$$\begin{aligned} \Delta u &= -\lambda_i u, \\ \frac{\partial u}{\partial n} \Big|_{\partial \Omega} &= 0, \end{aligned} \tag{3.34}$$

are $0 = \lambda_0 < \lambda_1 \leq \lambda_2 \leq \cdots$ and $e^0(x), e^1(x), e^2(x), \cdots$, respectively. Let $v(t) \in X^{\alpha}$ be any solution of (3.33). Then $v(t,x) = \sum_{j=0}^{\infty} v_j(t)e^j(x)$ where $v_j(t) := \langle e^j, v(t) \rangle = \int_{\Omega} v(t,x)e^j(x)dx, j = 0, 1, 2, \cdots$ and (3.33) is equivalent to the following infinite dimensional system of linear equations

$$\frac{dv_j}{dt} = -\lambda_j Dv_j + \frac{\partial f}{\partial u}(\phi(t))v_j.$$
(3.35)

While the methods of Section 3.1 may be used directly to analyze the characteristic multipliers of the periodic solution $\phi(t)$ with the help of the second compound differential equation of (3.33), a simpler development presented here is based on the finite dimensional theory of compound matrices by Muldowney [78] and the material in Henry [42], page 201-202. To that end, the above infinite system of uncoupled *n*-dimensional linear differential equations (3.35) is considered.

Lemma 3.17. The characteristic multipliers of the periodic solution $\phi(t)$ of (3.32) are the characteristic multipliers of the infinite dimensional system (3.35).

Proof. Let $v(t) \in X^{\alpha}$ be any solution of (3.33). then $v(t,x) = \sum_{j=0}^{\infty} v_j(t)e^j(x)$, where $v_j(t) = \langle e^j, v(t) \rangle$ satisfies (3.35).

If μ_j is a characteristic multiplier of (3.35), then there exists a nonzero $v_i(t)$ such that

$$v_j(\omega) = \mu_j v_j(0).$$

Since $v_j(t)e^j(x)$ is a solution of (3.33), μ_j is a characteristic multiplier of the periodic solution $\phi(t)$ of (3.32).

If μ is a characteristic multiplier of the periodic solution $\phi(t)$ of (3.32), then there exists a nonzero $v(t, x) = \sum_{j=0}^{\infty} v_j(t)e^j(x)$ such that

$$v(\omega) = \mu v(0),$$

or

$$\sum_{j=0}^{\infty} v_j(\omega) e^j(x) = \mu \sum_{i=0}^{\infty} v_j(0) e^j(x).$$

Thus for all $j = 0, 1, 2, \dots$,

$$v_j(\omega) = \mu v_j(0).$$

Since v(t) is nonzero, at least one of the v_j is not identically zero and thus μ is a characteristic multiplier of (3.35) for some j.

Remark 3.6. The technique developed here can also be applied to other boundary conditions such that p(t) is still a periodic solution for the reaction diffusion equation, for example, periodic boundary condition.

3.3.1 Stability of Periodic Solutions

The following theorem is an analogue for the reaction diffusion equation (3.32) of Poincaré's stability criterion for the ordinary differential equation (3.31).

Theorem 3.18. Suppose that $\phi(t)$ is a periodic solution of the ordinary differential equation (3.31). Then $\phi(t)$ is stable hyperbolic for the diffusive system (3.32) if and only if the second compound differential equation

$$\frac{dw}{dt} = \frac{\partial f}{\partial u}^{[2]}(\phi(t))w, \qquad (3.36)$$

and the linear equations

$$\frac{dv_j}{dt} = -\lambda_j Dv_j + \frac{\partial f}{\partial u}(\phi(t))v_j, \quad j = 1, 2, \cdots,$$
(3.37)

are uniformly asymptotically stable.

Note that the stability of equation (3.36) is a necessary and sufficient condition for the periodic solution $\phi(t)$ of the ordinary differential equation (3.31) to be stable hyperbolic. In addition, Poincaré's stability criterion here also requires the stability of the infinite system of decoupled *n*-dimensional linear equations (3.37). The following corollaries give concrete sufficient conditions on *D* for the preservation of stability in the diffusion system. The concept, "Lozinskiĭ measure", is discussed in Appendix A. Good references on Lozinskiĭ measure are Coppel [19], page 41 and Muldowney [76–78].

Corollary 3.19. Suppose that the periodic solution $\phi(t)$ of the ordinary differential equation (3.31) is stable hyperbolic. If, for some Lozinskii measure μ ,

$$\mu(-D) \le 0 \tag{3.38}$$

and

$$\int_{0}^{\omega} \mu\left(\frac{\partial f}{\partial u}(\phi(t)) - \lambda_{1}D\right) < 0, \qquad (3.39)$$

then $u = \phi(t)$ is stable hyperbolic for the diffusive system (3.32).

Remark 3.7. The Lozinskiĭ measure μ in Corollary 3.19 is assumed to be admissible. The term "admissible" was introduced by Li and Wang [57]. Precisely, a Lozinskiĭ measure μ is said to be admissible if $\mu(-D) \leq 0$ for all diagonal $D = \text{diag}(d_1, d_2, \dots, d_n)$ where $d_i \geq 0, 1 \leq i \leq n$. The Lozinskiĭ measures μ listed in Table A.1 of Appendix A are admissible.

Corollary 3.20. Suppose that the periodic solution $\phi(t)$ of the ordinary differential equation (3.31) is stable hyperbolic. Then there exists K > 0 such that

$$|X(t)X^{-1}(s)| \le K, \quad 0 \le s \le t,$$

where X(t) is a fundamental matrix solution of

$$rac{dv}{dt} = rac{\partial f}{\partial u}(\phi(t))v.$$

If there exists a constant d > 0 such that

$$|D - dI| < \frac{d}{K},\tag{3.40}$$

then $u = \phi(t)$ is stable hyperbolic for the diffusive system (3.32).

Remark 3.8. Neither of Corollary 3.19 and Corollary 3.20 is implied by the other. This is discussed through an example in Section 3.3.3. For the case D = dI, inequality (3.39) is true but inequality (3.40) may not be satisfied.

Remark 3.9. Corollary 3.20 includes as special cases results of Henry [42] and Leiva [50]. When n = 2 and $D = \text{diag}(d_1, d_2)$, Henry [42] proved that if $|d_1 - d_2|$ is small, $u = \phi(t)$ is orbitally asymptotically stable with asymptotic phase for the reaction diffusion system (3.32). Leiva [50] generalized this result for $D = dI + \text{diag}(\varepsilon_1, \varepsilon_2, \cdots, \varepsilon_n)$ with ε_i small enough. The results of Henry and Leiva are strengthened slightly in that a concrete bound is given here on ε_i . A particular example obtaining such a bound is discussed in Section 3.3.3.

Proof of Theorem 3.18:

Proof. Theorem 3.15 implies that $\phi(t)$ is stable hyperbolic for the first equation of (3.37) with λ_0 if and only if the second compound differential equation (3.36) is uniformly asymptotically stable. Theorem 7.2 in Hale [37], page 120, implies that the characteristic multipliers of the infinite dimensional linear equations (3.37) with $\lambda_j = \lambda_1, \lambda_2, \cdots$ have moduli less than 1 if and only if it is uniformly asymptotically stable. By Lemma 3.17, the proof is completed.

Proof of Corollary 3.19:

Proof. Let $A(t) = \frac{\partial f}{\partial u}(\phi(t))$. A solution of (3.37) satisfies

$$|v_j(t)| \le |v_j(s)| \exp\left(\int_s^t \mu(A - \lambda_j D)\right), \quad s \le t, \quad j = 1, 2, \cdots,$$

which implies that

$$|X_j(t)X_j^{-1}(s)| \le \exp\left(\int_s^t \mu(A - \lambda_j D)\right), \quad s \le t, \quad j = 1, 2, \cdots,$$

where $X_j(t)$ is a fundamental matrix solution of (3.37) and $X_j(0) = I$. Thus

$$|X_j(\omega)| \le \exp\left(\int_0^\omega \mu(A - \lambda_j D)\right).$$

If $\mu(-D) \leq 0$ and $\int_0^{\omega} \mu(A - \lambda_1 D) < 0$, then for $j = 2, 3, \cdots$,

$$\int_0^\omega \mu(A - \lambda_j D) = \int_0^\omega \mu(A - \lambda_1 D - (\lambda_j - \lambda_1)D)$$

$$\leq \int_0^\omega (\mu(A - \lambda_1 D) + (\lambda_j - \lambda_1)\mu(-D)) < 0,$$

which implies that all of the characteristic multipliers of (3.37) with $\lambda = \lambda_j, j = 1, 2, \cdots$ have moduli less than 1. Therefore, $u = \phi(t)$ is stable hyperbolic for the diffusive system (3.32).

Proof of Corollary 3.20:

Proof. Since the periodic solution $\phi(t)$ of the ordinary differential equation (3.31) is stable hyperbolic, Theorem 7.2 in Hale [37], page 120, implies that there exists K > 0 such that

$$|X(t)X^{-1}(s)| \le K, \quad s \le t,$$

where X(t) is a fundamental matrix solution of

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(\phi(t))v$$

Let $A(t) = \frac{\partial f}{\partial u}(\phi(t))$. Choose d > 0 and consider the following equation

$$\frac{dy}{dt} = -dIy + A(t)y. \tag{3.41}$$

Then the fundamental matrix solution Y(t) of (3.41) satisfies $Y(t) = X(t) \exp(-dt)$ and thus

$$|Y(t)Y^{-1}(s)| \le K \exp(-d(t-s)), \quad s \le t.$$

Now consider differential equation

$$\frac{dz}{dt} = (-D + A(t))z. \tag{3.42}$$

Then the solution of (3.42),

$$z(t) = Y(t)Y^{-1}(s)z(s) + \int_{s}^{t} Y(t)Y^{-1}(\tau)(dI - D)z(\tau)d\tau, \quad s \le t,$$

satisfies

$$|z(t)| \le K \exp(-d(t-s))|z(s)| + \int_s^t K \exp(-d(t-\tau))|D - dI| \cdot |z(\tau)| d\tau.$$

Gronwall's inequality implies

$$|z(t)| \le K \exp((-d + K|D - dI|)(t - s))|z(s)|, \quad s \le t.$$

Therefore, if

$$|D - dI| < \frac{d}{K},\tag{3.43}$$

then (3.42) is uniformly asymptotically stable. Since $\lambda_j > 0, j = 1, 2, \cdots$ and

$$|\lambda_j D - \lambda_j dI| = \lambda_j |D - dI| < \frac{\lambda_j d}{K},$$

the above argument implies that if $|D - dI| < \frac{d}{K}$, then

$$\frac{dv_j}{dt} = (-\lambda_j D + A(t))v_j$$

is uniformly asymptotically stable. Therefore, from Theorem 3.18, $u = \phi(t)$ is stable hyperbolic for the diffusive system (3.32).

3.3.2 Instability of Periodic Solutions

Since solutions of the ordinary differential equation (3.31) are also solutions of the reaction diffusion equation (3.32), it follows that an orbitally unstable solution of (3.31) is also an orbitally unstable solution of (3.32). In the following, it is assumed that $u = \phi(t)$ is an orbitally stable solution and conditions under which it is an orbitally unstable solution of (3.32) are explored. **Theorem 3.21.** Suppose that for some Lozinskii measure μ ,

$$\mu(-D) < 0.$$

Then the stable manifold of the solution $u = \phi(t)$ of the diffusive system (3.32) has finite codimension.

Proof. Since the Jacobian matrix $A(t) = \frac{\partial f}{\partial u}(\phi(t))$ is a continuous and periodic function of t, there exists a $\beta > 0$ such that

$$\int_{s}^{t} \mu(A) \le \beta(t-s), \quad s \le t.$$
(3.44)

If $\lambda > \frac{\beta}{-\mu(-D)}$, then

$$\int_{s}^{t} \mu(-\lambda D + A) \leq \lambda \mu(-D)(t - s) + \int_{s}^{t} \mu(A) \leq (\lambda \mu(-D) + \beta)(t - s) < 0,$$

which implies that the linear ordinary differential equation

$$rac{dv}{dt} = (-\lambda D + A(t))v$$

is uniformly asymptotically stable. Thus there exists N such that (3.37) with $\lambda = \lambda_j$ is uniformly asymptotically stable for $j \geq N$. Therefore the stable manifold of the solution $u = \phi(t)$ of the diffusive system (3.32) has finite codimension.

Theorem 3.22. Suppose that $u = \phi(t)$ is orbitally asymptotically stable for the ordinary differential equation

$$\frac{du}{dt} = f(u).$$

Let $A(t) = \frac{\partial f}{\partial u}(\phi(t))$. Suppose that there is a principal $m \times m$ submatrix A_{11} of A and a Lozinskii measure μ_1 on \mathbb{R}^m such that

$$\int_{0}^{\omega} -\mu_{1}(-A_{11}) > 0. \tag{3.45}$$

Then, for any integer k > 0, there exist an $\varepsilon > 0$ and a $n \times n$ matrix $D = diag(\varepsilon I_{m \times m}, \frac{1}{\varepsilon} I_{(n-m) \times (n-m)})$ such that $u = \phi(t)$ is orbitally unstable for the diffusive system

$$\frac{\partial u}{\partial t} = D\Delta u + f(u),$$

$$\frac{\partial u}{\partial n}\Big|_{\partial\Omega} = 0,$$
(3.46)

and the dimension of the unstable manifold of $u = \phi(t)$ is at least k.

Condition (3.45) in Theorem 3.22 is satisfied if for some i, $\int_0^{\omega} a_{ii}(t)dt > 0$. Without loss of generality, let i = 1.

Corollary 3.23. If $\int_0^{\omega} a_{11}(t)dt > 0$, then for any integer k > 0, there exist an $\varepsilon > 0$ and a $n \times n$ matrix $D = diag(\varepsilon, \frac{1}{\varepsilon}I_{(n-1)\times(n-1)})$, where $\varepsilon > 0$, such that $u = \phi(t)$ is orbitally unstable for the diffusive system (3.46) and the dimension of the unstable manifold of $u = \phi(t)$ is at least k.

The proof of Theorem 3.22 is based on an argument in Muldowney [77].

Proof. Consider the linear equation

$$\frac{dv_1}{dt} = (A_{11}(t) - \lambda \varepsilon I_{m \times m})v_1 + A_{12}(t)v_2,
\frac{dv_2}{dt} = A_{21}(t)v_1 + \left(A_{22}(t) - \frac{\lambda}{\varepsilon}I_{(n-m)\times(n-m)}\right)v_2,$$
(3.47)

where $\varepsilon > 0$, $\int_0^{\omega} -\mu_1(-A_{11}) > 0$ for some Lozinskiĭ measure μ_1 and $v = (v_1, v_2), v_1 \in \mathbb{R}^m, v_2 \in \mathbb{R}^{n-m}$. Let $|\cdot|_1$ and $|\cdot|_2$ be vector norms in \mathbb{R}^m and \mathbb{R}^{n-m} , μ_1 and μ_2 the corresponding Lozinskiĭ measures, respectively. Let $r_1 = |v_1|_1$ and $r_2 = |v_2|_2$. Then

$$\begin{aligned} \frac{d^+r_1}{dt} &\geq -\mu_1(-A_{11} + \lambda\varepsilon I_{m\times m})r_1 - \alpha_{12}r_2 \geq (\alpha_{11} - \lambda\varepsilon)r_1 - \alpha_{12}r_2, \\ \frac{d^+r_2}{dt} &\leq \alpha_{21}r_1 + \mu_2\left(A_{22} - \frac{\lambda}{\varepsilon}I_{(n-m)\times(n-m)}\right)r_2 \leq \alpha_{21}r_1 + \left(\alpha_{22} - \frac{\lambda}{\varepsilon}\right)r_2, \end{aligned}$$

where

$$\begin{aligned} \alpha_{11}(t) &= -\mu_1(-A_{11}(t)), \qquad \alpha_{22}(t) = -\mu_2(-A_{22}(t)), \\ \alpha_{12}(t) &= |A_{12}(t)|_{21} = \sup_{x \in \mathbb{R}^{n-k}, \ x \neq 0} \frac{|A_{12}(t)x|_1}{|x|_2}, \\ \alpha_{21}(t) &= |A_{21}(t)|_{12} = \sup_{x \in \mathbb{R}^k, \ x \neq 0} \frac{|A_{21}(t)x|_2}{|x|_1}, \end{aligned}$$

and $\frac{d^+}{dt}$ denotes the right-hand derivative. Since $\int_0^{\omega} -\mu_1(-A_{11}) > 0$, we can choose $\varepsilon, \delta > 0$ sufficiently small such that

$$\int_0^{\omega} (\alpha_{11} - \lambda \varepsilon - \delta \alpha_{21}) > 0, \qquad (3.48)$$

and

$$\alpha_{22}(t) - \frac{\lambda}{\varepsilon} + \frac{1}{\delta}\alpha_{12}(t) \le \alpha_{11}(t) - \lambda\varepsilon - \delta\alpha_{21}(t), \quad \text{for all } t \ge 0.$$
(3.49)

Let $r = r_1 - \delta r_2$. Then

$$\frac{d^{+}r}{dt} \geq (\alpha_{11} - \lambda\varepsilon)r_{1} - \alpha_{12}r_{2} - \delta\alpha_{21}r_{1} - \delta\left(\alpha_{22} - \frac{\lambda}{\varepsilon}\right)r_{2} \\
= (\alpha_{11} - \lambda\varepsilon - \delta\alpha_{21})r_{1} + \left(\frac{\lambda}{\varepsilon} - \alpha_{22} - \frac{1}{\delta}\alpha_{12}\right)\delta r_{2} \\
\geq (\alpha_{11} - \lambda\varepsilon - \delta\alpha_{21})r,$$

which implies

$$r(t) \ge r(s) \exp\left(\int_{s}^{t} (\alpha_{11} - \lambda \varepsilon - \delta \alpha_{21})\right), \text{ for } t \ge s$$

Choose $v_1(0)$ and $v_2(0)$ such that r(0) > 0. Then

$$r(n\omega) \ge r(0) \exp\left(n \int_0^\omega (\alpha_{11} - \lambda \varepsilon - \delta \alpha_{21})\right) \longrightarrow \infty, \text{ as } n \to \infty,$$

and thus

$$v_1(n\omega) \longrightarrow \infty$$
, as $n \to \infty$.

Therefore (3.47) is unstable and at least one of the characteristic multipliers of (3.47) has modulus greater than 1. Let $\lambda_1, \lambda_2, \dots, \lambda_k$ be the first k nonzero eigenvalues of the Laplace equation

$$\Delta u = -\lambda_i u,$$

 $\left. \frac{\partial u}{\partial n} \right|_{\partial \Omega} = 0.$

Then there exists an $\varepsilon > 0$ such that the inequalities (3.48) and (3.49) hold for all $\lambda_1, \lambda_2, \dots, \lambda_k$. The above argument for λ implies that for each $j = 1, 2, \dots, k$, the linear equation (3.47) has at least one of the characteristic multipliers with modulus greater than 1. Hence, Lemma 3.5 and Lemma 3.17 imply that $u = \phi(t)$ is orbitally unstable for the diffusive system and the dimension of the unstable manifold of $u = \phi(t)$ is at least k.

3.3.3 Planar Reaction Diffusion Systems

The following results include Henry [42], page 201-202, as the special case n = 2 of Theorem 3.18 and Corollary 3.23.

Suppose that the planar system

$$\frac{du}{dt} = f(u, v),$$

$$\frac{dv}{dt} = g(u, v),$$
(3.50)

has a nonconstant periodic solution (p(t), q(t)) with period ω , where $f, g \in C^1(\mathbb{R}^2 \to \mathbb{R})$. In this case the second compound differential equation of the linear variational equation at (p(t), q(t)) is the Liouville equation

$$\begin{aligned} \frac{dw}{dt} &= \left(\frac{\partial f}{\partial u}(p,q) + \frac{\partial f}{\partial v}(p,q)\right)w \\ &= (f_u(p,q) + g_v(p,q))w. \end{aligned}$$

Theorem 3.15 implies that (p(t), q(t)) is stable hyperbolic for the ordinary differential equation (3.50) if and only if

$$\int_{0}^{\omega} (f_{u}(p,q) + g_{v}(p,q)) < 0.$$
(3.51)

Solution (p(t), q(t)) is also a periodic solution of the diffusive system

$$u_t = d_1 u_{xx} + f(u, v), v_t = d_2 v_{xx} + g(u, v), \quad \text{in } (0, 2\pi) \times (0, \infty), \quad d_1, d_2 \ge 0,$$
(3.52)

with Neumann boundary conditions

$$u_x(t,0) = u_x(t,2\pi) = 0,$$

$$v_x(t,0) = v_x(t,2\pi) = 0, \quad \text{for } t \ge 0,$$
(3.53)

or periodic boundary conditions

$$u(t,0) = u(t,2\pi), \quad u_x(t,0) = u_x(t,2\pi)$$

$$v(t,0) = v(t,2\pi), \quad v_x(t,0) = v_x(t,2\pi), \quad \text{for } t \ge 0.$$
(3.54)

Let

$$M_{j}(t) = \begin{bmatrix} f_{u}(p,q) - \lambda_{j}d_{1} & f_{v}(p,q) \\ g_{u}(p,q) & g_{u}(p,q) - \lambda_{j}d_{2} \end{bmatrix},$$
 (3.55)

where $\lambda_0 = 0, \lambda_j = (j/2)^2$ for Neumann boundary conditions and $\lambda_0 = 0, \lambda_{2j-1} = \lambda_{2j} = j^2$ for periodic boundary conditions, $j = 1, 2, \cdots$. The characteristic multipliers of (p(t), q(t)) for the reaction diffusion system (3.52) are the characteristic multipliers of

$$\begin{bmatrix} \frac{du}{dt} \\ \frac{dv}{dt} \end{bmatrix} = M_j(t) \begin{bmatrix} u \\ v \end{bmatrix}, \quad j = 0, 1, 2, \cdots.$$
(3.56)

Let $d_1, d_2 \ge 0$ and $d_1 + d_2 \ne 0$. Corollary 3.24, 3.25 and 3.26 are given in Henry [42]. Corollary 3.27 and 3.28 are new and follow from Corollary 3.23.

Corollary 3.24. Suppose $\int_0^{\omega} (f_u(p,q) + g_v(p,q)) > 0$. Then (p(t),q(t)) is orbitally unstable for the diffusive system (3.52).

Corollary 3.25. Suppose $\int_0^{\omega} (f_u(p,q) + g_v(p,q)) < 0$. If $\int_0^{\omega} g_v(p,q) > 0$, then (p(t), q(t)) is orbitally unstable for the diffusive system (3.52) with $d_1 = \varepsilon^{-1}, d_2 = \varepsilon$ where ε is sufficiently small.

Corollary 3.26. Suppose $\int_0^{\omega} (f_u(p,q) + g_v(p,q)) < 0$. If $|d_1 - d_2|$ is small, then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system (3.52).

Corollary 3.27. Suppose $\int_{0}^{\omega} (f_u(p,q) + g_v(p,q)) < 0$. If

$$\int_{0}^{\omega} \max\left\{ f_{u}(p,q) - \lambda_{1}d_{1} + |f_{v}(p,q)|, g_{v}(p,q) - \lambda_{1}d_{2} + |g_{u}(p,q)| \right\} < 0,$$
(3.57)

or

$$\int_{0}^{\omega} \max\left\{f_{u}(p,q) - \lambda_{1}d_{1} + |g_{u}(p,q)|, g_{u}(p,q) - \lambda_{1}d_{2} + |f_{v}(p,q)|\right\} < 0,$$
(3.58)

or

$$\int_0^\omega \lambda\left(\frac{M_1+M_1^*}{2}\right) < 0, \tag{3.59}$$

where $\lambda\left(\frac{M_1+M_1^*}{2}\right)$ denotes the largest eigenvalue of $\frac{M_1+M_1^*}{2}$, then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system (3.52).

The orbital asymptotic stability of (p(t), q(t)) can also be discussed by using the result of Corollary 3.14 in Section 3.2. Let

$$\mu_{1}(t) = -\lambda_{1}d_{1} + \max_{x} \left\{ 2f_{u}(p,q) + \frac{1}{\nu_{1}(x)} |f_{v}(p,q) + g_{u}(p,q)| \right\},$$

$$\mu_{2}(t) = -\lambda_{1}d_{2} + \max_{x} \left\{ 2g_{v}(p,q) + \frac{1}{\nu_{2}(x)} |f_{v}(p,q) + g_{u}(p,q)| \right\},$$

$$\mu_{3}(t) = \max_{x} \left\{ f_{u}(p,q) + \frac{\nu_{2}(x)}{2} |f_{v}(p,q) + g_{u}(p,q)| \right\}$$

$$+ \max_{x} \left\{ g_{v}(p,q) + \frac{\nu_{1}(x)}{2} |f_{v}(p,q) + g_{u}(p,q)| \right\},$$

$$\mu(t) = \sup\{\mu_{1}(t), \mu_{2}(t), \mu_{3}(t)\},$$

$$(3.60)$$

where $\nu_1(x), \nu_2(x)$ are positive functions and the maximum is taken over $0 < x < 2\pi$.

Corollary 3.28. Suppose that

$$\int_0^\omega \mu < 0,$$

where $\mu(\cdot)$ is defined by (3.60). Then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system (3.52).

The next example is an application of Corollary 3.27.

Example 3.1. ([18, 35]) Consider the following predator-prey system with logistic growth for prey u in the absence of predation and Holling type II functional response for predator v:

$$\frac{du}{dt} = ru\left(1 - \frac{u}{K}\right) - \frac{Mu}{N+u}v := f(u,v),$$

$$\frac{dv}{dt} = \left(\frac{Mu}{N+u} - \beta\right)v := g(u,v),$$
(3.61)

where $M, N, K, r, \beta > 0$. Let $M > \beta$ and $\eta = \frac{N\beta}{M-\beta}$. This system has been studied in Cheng [18]. If $\frac{K-N}{2} < \eta < K$, then the equilibrium (u^*, v^*) is globally asymptotically stable where $u^* = \eta, v^* = \frac{r}{M}(N + u^*)\left(1 - \frac{u^*}{K}\right)$. If $0 < \eta < \frac{K-N}{2}$, then the equilibrium (u^*, v^*) is unstable and a unique periodic solution exists. Let (p(t), q(t)) denote the unique periodic solution. Then

$$\gamma = \{ (p(t), q(t)), \ 0 \le t \le \omega \} \subseteq \{ (u, v), \ 0 < u < K, \ 0 < v \}.$$

The second compound differential equation of the linear variational equation at (p(t), q(t)) is the Liouville equation

$$\frac{dw}{dt} = \left(r\left(1 - \frac{2p}{K}\right) - \frac{MNq}{(N+p)^2} + \frac{Mp}{N+p} - \beta\right)w.$$

It has been shown in Cheng [18] that

$$\int_0^\omega \left(\frac{Mp}{(N+p)^2} - \beta\right) = 0, \qquad (3.62)$$

and

$$\int_0^\omega \left(r \left(1 - \frac{2p}{K} \right) - \frac{MNq}{(N+p)^2} \right) < 0.$$
(3.63)

Thus if $0 < \eta < \frac{K-N}{2}$, then Poincaré's stability criterion implies that the periodic solution (p(t), q(t)) of (3.61) is orbitally asymptotically stable with asymptotic phase. Let

$$M_{1}(t) = \begin{bmatrix} f_{u}(p,q) - \lambda_{1}d_{1} & f_{v}(p,q) \\ g_{u}(p,q) & g_{v}(p,q) - \lambda_{1}d_{2} \end{bmatrix}$$
$$= \begin{bmatrix} r\left(1 - \frac{2p}{K}\right) - \frac{MNq}{(N+p)^{2}} - \lambda_{1}d_{1} & -\frac{Mp}{N+p} \\ \frac{MNq}{(N+p)^{2}} & \frac{Mp}{N+p} - \beta - \lambda_{1}d_{2} \end{bmatrix}$$

Then $\frac{2MK}{N} - \beta > 0$ since $0 < \eta < \frac{K-N}{2}$. If

$$\lambda_1 d_1 \ge r, \ \lambda_1 d_2 > \frac{2MK}{N} - \beta,$$

then

$$\begin{split} \max\left\{r\left(1-\frac{2p}{K}\right) - \frac{MNq}{(N+p)^2} - \lambda_1 d_1 + \frac{MNq}{(N+p)^2}, \frac{Mp}{N+p} - \beta - \lambda_1 d_2 + \frac{Mp}{N+p}\right\} \\ &\leq \max\left\{r\left(1-\frac{2p}{K}\right) - \lambda_1 d_1, \frac{2MK}{N} - \beta - \lambda_1 d_2\right\} \\ &< 0. \end{split}$$

Corollary 3.27 implies the following result.

Corollary 3.29. Suppose that $0 < \frac{N\beta}{M-\beta} < \frac{K-N}{2}$. If $\lambda_1 d_1 \ge r$, $\lambda_1 d_2 > \frac{2MK}{N} - \beta$, then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system

$$u_t = d_1 u_{xx} + ru\left(1 - \frac{u}{K}\right) - \frac{Mu}{N+u}v,$$

$$v_t = d_2 v_{xx} + \left(\frac{Mu}{N+u} - \beta\right)v, \quad in \ (0, 2\pi) \times (0, \infty),$$
(3.64)

with Neumann boundary conditions (3.53) where $\lambda_1 = \frac{1}{4}$ or periodic boundary conditions (3.54) where $\lambda_1 = 1$.

The following example will show that neither of Corollary 3.26 and Corollary 3.27 is implied by the other. This example will also show how to find a concrete bound on $|d_1 - d_2|$ in Corollary 3.26.

Example 3.2. Consider a planar system

$$\frac{du}{dt} = \beta u + (1 - u^2 - \alpha v^2)u := f(u, v),
\frac{dv}{dt} = -\beta v + (1 - u^2 - \alpha v^2)v := g(u, v),$$
(3.65)

where $1 \leq \alpha < 2$. Rewrite (3.65) in polar coordinators as

$$\frac{dr}{dt} = r(1 - r^2 \cos^2(\theta) - \alpha r^2 \sin^2(\theta)),$$

$$\frac{d\theta}{dt} = -\beta.$$
(3.66)

Then $\frac{dr}{dt} > 0$ if $0 < r^2 \cos^2(\theta) + \alpha r^2 \sin^2(\theta) < 1$ and $\frac{dr}{dt} < 0$ if $r^2 \cos^2(\theta) + \alpha r^2 > 1$. 1. Thus, every solution of (3.65) except (0,0) ultimately enters and remains in the annular region

$$G = \left\{ (u, v) : \frac{1}{\alpha} \le u^2 + v^2 \le 1 \right\}.$$

The Poincaré-Bendixson theorem implies that there exists a periodic solution (p(t), q(t)) in G. System (3.66) shows that the period of the periodic solution (p(t), q(t)) is $\frac{2\pi}{\beta}$. The linear variational equation of (3.65) at any solution (p(t), q(t)) is

$$\frac{du}{dt} = (1 - 3p^2 - \alpha q^2)u + (\beta - 2\alpha pq)v,
\frac{dv}{dt} = -(\beta + 2pq)u + (1 - p^2 - 3\alpha q^2)v.$$
(3.67)

Since

$$1 - 3p^2 - \alpha q^2 + 1 - p^2 - 3\alpha q^2 = 2 - 4(p^2 + \alpha q^2) \le 2 - \frac{4}{\alpha} < 0, \qquad (3.68)$$

Poincaré's stability criterion implies that (p(t), q(t)) is asymptotically orbitally stable with asymptotic phase in G. A result in [60] implies that the periodic solution of (3.65) is unique. Let

$$M_{1}(t) = \begin{bmatrix} f_{u}(p,q) - \lambda_{1}d_{1} & f_{v}(p,q) \\ g_{u}(p,q) & g_{v}(p,q) - \lambda_{1}d_{2} \end{bmatrix}$$
$$= \begin{bmatrix} 1 - 3p^{2} - \alpha q^{2} - \lambda_{1}d_{1} & \beta - 2\alpha pq \\ -(\beta + 2pq) & 1 - p^{2} - 3\alpha q^{2} - d_{2} \end{bmatrix}.$$

If

$$\lambda_1 d_1 > 1 + |\beta|, \ \lambda_1 d_2 > 2\alpha + |\beta|,$$

then it follows from $\frac{1}{\alpha} \leq p^2 + q^2 \leq 1$ that

$$\begin{split} &\max\{1 - 3p^2 - \alpha q^2 - \lambda_1 d_1 + |\beta + 2pq|, 1 - p^2 - 3\alpha q^2 - \lambda_1 d_2 + |\beta - 2\alpha pq|\} \\ < &\max\{1 + |\beta| - \lambda_1 d_1, |\beta| + 2\alpha - \lambda_1 d_2\} \end{split}$$

< 0.

Corollary 3.27 implies the following result.

Corollary 3.30. If $d_1 > 1 + |\beta|$ and $d_2 > 2\alpha + |\beta|$, then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system

$$u_t = d_1 u_{xx} + \beta u + (1 - u^2 - \alpha v^2) u,$$

$$v_t = d_2 v_{xx} - \beta v + (1 - u^2 - \alpha v^2) v, \qquad \text{in } (0, 2\pi) \times (0, \infty),$$
(3.69)

with Neumann boundary conditions (3.53) where $\lambda_1 = \frac{1}{4}$ or periodic boundary conditions (3.54) where $\lambda_1 = 1$.

From the argument developed in the proof of Proposition 4.15 in Section 4.2.1, if $1 \le \alpha < \frac{4}{3}$, then the solution of (3.67) satisfies

$$\begin{split} \sqrt{y^2(t) + z^2(t)} &\leq K\sqrt{y^2(s) + z^2(s)}, \quad 0 \leq s \leq t, \\ \text{where } K &= 4e^{\frac{2\pi}{\beta} + 2T_2} \left(\frac{2\sqrt{\alpha(\beta^2+1)}}{\beta} + 1\right) \text{ and } T_2 = \frac{\frac{4\pi}{\beta} + 4\ln(2)}{\frac{4}{\alpha} - 3}. \text{ Let } d = \frac{d_1 + d_2}{2} \text{ and} \\ \varepsilon &= \frac{d_1 - d_2}{2}. \text{ Then} \\ \begin{pmatrix} d_1 & 0\\ 0 & d_2 \end{pmatrix} = \begin{pmatrix} d + \varepsilon & 0\\ 0 & d - \varepsilon \end{pmatrix}. \\ \text{Corollary 3 20 implies the following corollary.} \end{split}$$

Corollary 3.20 implies the following corollary.

Corollary 3.31. Suppose that $d_1, d_2 \ge 0, d_1 + d_2 \ne 0$ and $1 \le \alpha < \frac{4}{3}$. Let $K = 4e^{\frac{2\pi}{\beta} + 2T_2} \left(\frac{2\sqrt{\alpha(\beta^2+1)}}{\beta} + 1\right)$ and $T_2 = \frac{\frac{4\pi}{\beta} + 4\ln(2)}{\frac{4}{\alpha} - 3}$. If $|d_1 - d_2| \le \frac{d_1 + d_2}{K}$, then (p(t), q(t)) is orbitally asymptotically stable with asymptotic phase for the diffusive system (3.69) with Neumann boundary conditions (3.53) or periodic boundary conditions (3.54).

3.4 Scalar Reaction Diffusion Equations

Another type of reaction diffusion equation for which the existence of periodic solutions can be established is given in the form

$$u_t = u_{xx} + f(x, u, u_x), \quad x \in S^1 = \mathbb{R}/2\pi\mathbb{Z}, \ f \in C^2.$$
 (3.70)

Fiedler and Mallet-Paret [30] proved that the classical theorem of Poincaré and Bendixson holds for the scalar reaction diffusion equation (3.70): the ω -limit set of any bounded solution satisfies exactly one of the following alternatives:

(i) it consists of precisely one periodic solution, or

(ii) for every orbit in the ω -limit set, its ω -limit set and α -limit set are contained in the set of steady state solutions.

When $f = f(u, u_x)$ is analytic, the Poincaré-Bendixson theorem was proved by Angenent and Fiedler [4]. It was shown in [4] that any ω limit set contains either a rotating wave or a steady state. All periodic solutions of (3.70) in this case are rotating waves, that is solutions of the form u = u(x - ct), which are always unstable. Independently, Massatt [68] proved that either the ω -limit set is a single rotating wave, or a set of equilibria which differ only by shifting x. The same result was also obtained by Matano [71], who further shows that the ω -limit set is a single equilibrium if $f = f(u, u_x)$ is even in the second argument.

The scalar reaction diffusion equation (3.70) with Dirichlet boundary conditions or Neumann boundary conditions is gradient-like with respect to a continuous Lyapunov functional F of the form

$$F(u)=\int_0^{2\pi}g(x,u,u_x)dx.$$

In particular, $\omega(u_0)$ consists entirely of equilibria if the orbit of $u(t, u_0)$ is bounded. In the general case of a scalar semilinear parabolic equation on the line segment (0, 1), Zelenyak [114] and, independently, Matano [70, 71] prove this result, where Zelenyak uses a special Lyapunov function to obtain this result and Matano uses a nontrivial application of the maximum principle. This result is also proved by Fiedler and Mallet-Paret [30], who show that the scalar reaction diffusion equation (3.70) with Dirichlet boundary conditions does not have periodic solutions (see page 339), and thus $\omega(u_0)$ consists entirely of equilibria. Hale and Raugel [39] consider an extension of a result of Hale and Massatt [38] on convergence of orbits to a single equilibrium point for gradient-like systems using the theory of integral manifolds in dynamical systems.

One of the tools in the analysis of ω -limit sets and the maximal compact attractor is the zero number, see Nickel [84], Matano [70], Henry [43], Brunovský and Fiedler [10]. For any continuous $u: S^1 \to \mathbb{R}$, the zero number z(u) is the number of sign changes of u, not counting multiplicity, that is z(u)is the maximal integer $n \leq \infty$ such that there exist $0 \leq x_{n+1} = x_0 < x_1 < \cdots < x_n < 2\pi$ with

$$u(x_i) \cdot u(x_{i+1}) < 0, \quad 0 \le i \le n.$$

Set z(0) := 0. The crucial property of z(u) is the following. For any solution $u(t, u_0)$ of

$$u_t = u_{xx} + f(x, u, u_x), \quad x \in S^1,$$

with f sufficiently regular and f(x, 0, 0) = 0, the function $t \mapsto z(u(t, \cdot))$ is nonincreasing in t.

Now suppose that $u = \phi(t, x)$ is a nonconstant ω -periodic solution of

$$u_t = u_{xx} + f(x, u, u_x), \quad 0 < x < 2\pi, \quad t > 0, u(t, 0) = u(t, 2\pi), u_x(t, 0) = u_x(t, 2\pi), \quad t \ge 0,$$
(3.71)

where $f \in C^2((0, 2\pi) \times \mathbb{R} \times \mathbb{R} \to \mathbb{R})$. The initial value problem associated to (3.71) generates a local semiflow on the Sobolev space X which contains all functions in $H^s(0, 2\pi)$ satisfying the periodic boundary conditions. Choose $s > \frac{5}{2}$ so that X embeds into $C^2(0, 2\pi)$. For $u_0 \in H^s(0, 2\pi)$, u(t, x) =

 $u(t, u_0) \in C^2(0, 2\pi)$ is a solution which exists on [0, T) for some T > 0. The linear variational equation of (3.71) at $\phi(t, x)$ is

$$v_t = v_{xx} + a(t, x)v + b(t, x)v_x, \quad 0 < x < 2\pi, \quad t > 0,$$

$$v(t, 0) = v(t, 2\pi), v_x(t, 0) = v_x(t, 2\pi), \quad t \ge 0,$$
(3.72)

where

$$a(t,x) := \frac{\partial f}{\partial u}(x,\phi,\phi_x) = f_u(x,\phi,\phi_x), \quad b(t,x) := \frac{\partial f}{\partial u_x}(x,\phi,\phi_x) = f_{u_x}(x,\phi,\phi_x)$$

If v_1, v_2, \dots, v_k are solutions of (3.72), then $w(t) = (v_1 \wedge v_2 \wedge \dots \wedge v_k)(t) \in \bigwedge^k X$ has pointwise representation

$$w(t, x_1, x_2, \cdots, x_4) = (v_1 \wedge v_2 \wedge \cdots \wedge v_k)(t, x_1, x_2) = \det(v_i(t, x_j)),$$

which satisfies the kth compound differential equation of (3.72) defined on $\bigwedge^k X$

$$w_t = \sum_{j=1}^k w_{x_j x_j} + \sum_{j=1}^k a(t, x_j) w_{x_j} + \sum_{j=1}^k b(t, x_j) w.$$
(3.73)

The periodic boundary conditions of v^i implies that

$$w(t, x_1, \cdots, x_{j-1}, 0, x_{j+1}, \cdots, x_k) = w(t, x_1, \cdots, x_{j-1}, 2\pi, x_{j+1}, \cdots, x_k)$$
$$w_{x_j}(t, x_1, \cdots, x_{j-1}, 0, x_{j+1}, \cdots, x_k) = w_{x_j}(t, x_1, \cdots, x_{j-1}, 2\pi, x_{j+1}, \cdots, x_k)$$
$$j = 1, 2, \cdots, k.$$

The eigenvalues and orthonormal eigenfunctions of the Laplace equation

$$\Delta u = -\lambda_i u,$$

$$u(0) = u(2\pi), u_x(0) = u_x(2\pi)$$
(3.74)

are $\lambda_0 = 0, \lambda_{2n-1} = n^2, \lambda_{2n} = n^2, n = 1, 2, \cdots$, and $e^0(x) = 1, e^{2n-1}(x) = \frac{1}{\sqrt{\pi}} \cos nx, e^{2n}(x) = \frac{1}{\sqrt{\pi}} \sin nx, \cdots$ respectively.

A particular case of Theorem 3.8 is the following theorem.

Theorem 3.32. Suppose that $f \in C^2((0, 2\pi) \times \mathbb{R} \times \mathbb{R} \to \mathbb{R})$ and that $\phi(t, x)$ is an ω -periodic solution of (3.71). If the kth compound equation (3.73) is uniformly asymptotically stable in $\bigwedge^k X$, then $\phi(t, x)$ has a stable manifold with codimension at most k - 2.

Consider a Lyapunov function

$$V(t) = \frac{1}{2} \int_{(0,2\pi)^k} w^2 = \frac{1}{2} \int_{(0,2\pi)^k} w^2(t, x_1, x_2, \cdots, x_k) dx_1 dx_2 \cdots dx_k.$$

It follows from Young's inequality that

$$\begin{aligned} \frac{d^+V}{dt} &= \int_{(0,2\pi)^k} ww_t \\ &= \int_{(0,2\pi)^k} \left(\sum_{j=1}^k ww_{x_jx_j} + \sum_{j=1}^k a(t,x_j)ww_{x_j} + \sum_{j=1}^k b(t,x_j)w^2 \right) \\ &\leq -\sum_{j=1}^k \int_{(0,2\pi)^k} (w_{x_j})^2 + \sum_{j=1}^k \int_{(0,2\pi)^k} \frac{1}{2} |a(t,x_j)| \left[\nu_j(x_j)w^2 + \frac{(w_{x_j})^2}{\nu_j(x_j)} \right] \\ &+ \int_{(0,2\pi)^k} \sum_{j=1}^k b(t,x_j)w^2 \\ &= \sum_{j=1}^k \int_{(0,2\pi)^k} \left(\frac{|a(t,x_j)|}{2\nu_j(x_j)} - 1 \right) (w_{x_j})^2 \\ &+ \int_{(0,2\pi)^k} \sum_{j=1}^k \left(b(t,x_j) + \frac{|a(t,x_j)|\nu_j(x_j)}{2} \right) w^2, \end{aligned}$$

for any positive functions $\nu_1(x), \nu_2(x)$. If $\max_x |a(t,x)| > 0, 0 < x < 2\pi$, then a choice

$$\nu_j(x) = \max_{0 < x < 2\pi} |a(t, x)| := \nu(t), \quad j = 1, 2, \cdots, k,$$

implies that

$$\frac{d^+V}{dt} \le \mu(t)V,$$

where

$$\mu(t) = -\sum_{j=1}^{k-1} \lambda_j + 2k \max_x \left\{ b(t,x) + \frac{|a(t,x)|\nu(t)|}{2} \right\}.$$

If $\max_{x} |a(t,x)| = 0$, then $a(t,x) \equiv 0$ for all $0 < x < 2\pi$ and

$$\mu(t) = -\sum_{j=1}^{k-1} \lambda_j + k \max_x b(t,x).$$

Theorem 3.32 implies the following corollaries.

Corollary 3.33. Suppose that $f \in C^2((0, 2\pi) \times \mathbb{R} \times \mathbb{R} \to \mathbb{R})$ and $\phi(t, x)$ is an ω -periodic solution of (3.71). If $\max_x |a(t, x)| > 0$, $0 < x < 2\pi$ and

$$\int_0^\omega \max_x \left\{ b(t,x) + \frac{|a(t,x)|\nu(t)}{2} \right\} dt < \frac{\omega}{2k} \sum_{j=1}^{k-1} \lambda_j;$$

or if $a(t, x) \equiv 0$ for all $0 < x < 2\pi$ and

$$\int_0^\omega \max_x b(t,x) < \frac{\omega}{k} \sum_{j=1}^{k-1} \lambda_j$$

where $\lambda_{2j-1} = j^2$, $\lambda_{2j} = j^2$, $j = 1, 2, \cdots$ and $\nu(t) = \max_x |a(t, x)|$, then $\phi(t, x)$ has a stable manifold with codimension at most k - 2.

Example 3.3. Consider the van der Pol equation

$$\frac{d^2u}{dt^2} + \varepsilon (u^2 - 1)\frac{du}{dt} + u = 0.$$
(3.75)

Equation (3.75) has a unique stable limit cycle $\phi(t)$ with period ω for all values of $\varepsilon \geq 0$. The formula for ω as a function of ε can be found in [15, 25, 90, 106, 107]. Moreover, $\omega \geq 2\pi$. If c > 0, then $\phi(x + ct)$ is a $\frac{\omega}{c}$ -periodic solution of

$$u_t = u_{xx} + cu_x + \varepsilon (u^2 - 1)u_x + u, \quad 0 < x < \omega, \quad t > 0, u(t, 0) = u(t, \omega), \quad u_x(t, 0) = u_x(t, \omega), \quad t \ge 0.$$
(3.76)

The initial value problem associated to (3.76) generates a local semiflow on the Sobolev space $X = H^s_{per}(0,\omega)$ which contains all functions in $H^s(0,2\pi)$ satisfying the periodic boundary conditions. Choose $s > \frac{5}{2}$ so that X embeds into $C^2(0,\omega)$. Now for $u_0 \in H^s_{per}(0,\omega)$, $u(t,x) \in C^2(0,\omega)$ is a classical solution which exists on [0,T) for some T > 0. Since

$$\begin{split} \int_{0}^{\omega} u_{t} dx &= \int_{0}^{\omega} \left[u_{xx} + c u_{x} + \varepsilon (u^{2} - 1) u_{x} + u \right] dx \\ &= u_{x} |_{0}^{\omega} + c u |_{0}^{\omega} + u^{3} |_{0}^{\omega} - u |_{0}^{\omega} + \int_{0}^{\omega} u dx \\ &= \int_{0}^{\omega} u dx, \end{split}$$

 $G \; = \; \left\{ u \in H^s_{per}(0,\omega) : \int_0^\omega u(x) dx = 0 \right\} \text{ is an invariant set with respect to}$ (3.76). In this invariant set G,

$$\int_0^\omega u u_{xx} dx \le -\frac{2\pi}{\omega} \int_0^\omega u^2 dx,$$

and thus,

$$\frac{1}{2}\frac{d}{dt}\int_{0}^{\omega}u^{2}dx = \int_{0}^{\omega}\left[uu_{xx} + cu_{x}u + \varepsilon(u^{2} - 1)u_{x}u + u^{2}\right]dx$$
$$= \int_{0}^{\omega}\left[uu_{xx} + u^{2}\right]dx$$
$$\leq \left[-\left(\frac{2\pi}{\omega}\right)^{2} + 1\right]\int_{0}^{\omega}u^{2}dx,$$

which implies that the L^2 norm of u exists for all $t \ge 0$. For any T > 0,

$$\int_{0}^{\omega} u^{2}(t,x)dx \le \exp\left[\left(-4\pi^{2}/\omega^{2}+1\right)t\right] \int_{0}^{\omega} u^{2}(0,x)dx, \qquad (3.77)$$

which implies that $\int_0^{\omega} u^2 dx$ is uniformly bounded for all $0 \le t \le T$. For each $k \ge 2$, multiply $u^{2^{k-1}}$ on both sides of (3.76) and integrate it to obtain

$$\int_0^\omega u^{2^k - 1} u_t dx = \int_0^\omega \left[u^{2^k - 1} u_{xx} + c u_x u^{2^k - 1} + \varepsilon (u^2 - 1) u_x u^{2^k - 1} + u^{2^k} \right] dx,$$

that is,

$$\frac{1}{2^k}\frac{d}{dt}\int_0^\omega u^{2^k}dx = -\frac{2^k-1}{2^{2k-2}}\int_0^\omega |\nabla u^{2^{k-1}}|^2dx + \int_0^\omega u^{2^k}dx,$$

where $\nabla u = u_x$ or equivalently,

$$\frac{1}{2}\frac{d}{dt}\int_0^\omega (u^{2^{k-1}})^2 dx = -\frac{2^k-1}{2^{k-1}}\int_0^\omega |\nabla u^{2^{k-1}}|^2 dx + 2^{k-1}\int_0^\omega (u^{2^{k-1}})^2 dx.$$

Let

$$u^* = |u|^{2^{k-1}}, \quad \alpha_k = \frac{2^k - 1}{2^{k-1}}, \quad \sigma_k = 2^{k-1}.$$

Then

$$\frac{1}{2}\frac{d}{dt}\int_0^\omega (u^*)^2 dx = -\alpha_k \int_0^\omega |\nabla u^*|^2 dx + \sigma_k \int_0^\omega (u^*)^2 dx.$$
(3.78)

$$\|\xi\|_{L^2} \le \sqrt{C} \|\xi\|_{W^{1,2}}^{1/2} \|\xi\|_{L^1}^{1/2},$$

to obtain

$$\|\xi\|_{L^{2}}^{2} \leq C\varepsilon^{-1} \|\xi\|_{L^{1}}^{2} + C\varepsilon \|\xi\|_{W^{1,2}}^{2} = C\varepsilon^{-1} \|\xi\|_{L^{1}}^{2} + C\varepsilon \|\nabla\xi\|_{L^{2}}^{2} + C\varepsilon \|\xi\|_{L^{2}}^{2}.$$
(3.79)

Substitute u^* in the place of ξ in (3.79) and ε_k in the place of ε to get

$$\int_0^\omega (u^*)^2 dx \le C\varepsilon_k^{-1} \left(\int_0^\omega u^* dx\right)^2 + C\varepsilon_k \int_0^\omega |\nabla u^*|^2 dx + C\varepsilon_k \int_0^\omega (u^*)^2 dx$$

and thus

$$\left[(C\varepsilon_k)^{-1} - 1 \right] \int_0^\omega (u^*)^2 dx - \varepsilon_k^{-2} \left(\int_0^\omega u^* dx \right)^2 \le \int_0^\omega |\nabla u^*|^2 dx.$$
(3.80)

Choose $\varepsilon_k > 0$ small such that $-\frac{\alpha_k}{C} + (\alpha_k + \sigma_k)\varepsilon_k + (\varepsilon_k)^2 < 0$. It follows from (3.78) and (3.80) that

$$\frac{1}{2} \frac{d}{dt} \int_{0}^{\omega} (u^{*})^{2} dx$$

$$\leq -\alpha_{k} \left[(C\varepsilon_{k})^{-1} - 1 \right] \int_{0}^{\omega} (u^{*})^{2} dx + \alpha_{k} \varepsilon_{k}^{-2} \left(\int_{0}^{\omega} u^{*} dx \right)^{2} + \sigma_{k} \int_{0}^{\omega} (u^{*})^{2} dx$$

$$= \left[\sigma_{k} - \alpha_{k} \left(C\varepsilon_{k} \right)^{-1} + \alpha_{k} \right] \int_{0}^{\omega} (u^{*})^{2} dx + \alpha_{k} \varepsilon_{k}^{-2} \left(\int_{0}^{\omega} u^{*} dx \right)^{2}$$

$$\leq -\varepsilon_{k} \int_{0}^{\omega} (u^{*})^{2} dx + \alpha_{k} \varepsilon_{k}^{-2} \left(\int_{0}^{\omega} u^{*} dx \right)^{2}.$$
(3.81)

If $\int_0^{\omega} u^* dx$ is uniformly bounded for all $0 \le t \le T$, then

$$\int_0^\omega (u^*)^2 dx \le \max\left\{\frac{\alpha_k}{(\varepsilon_k)^3} \left[\sup_t \int_0^\omega u^* dx\right]^2, \int_0^\omega (u^*(0,x))^2 dx\right\}.$$

In particular, if k = 2, then $u^* = u^2$, $\sigma_2 = 2$ and $\alpha_2 = \frac{3}{2}$. Choose ε_2 such that $(2 + \varepsilon_2)C\varepsilon_2 \leq \frac{3}{2}$ and thus

$$\int_0^\omega u^4 dx \le \max\left\{\frac{3}{2(\varepsilon_2)^3} \left[\sup_t \int_0^\omega u^2 dx\right]^2, \int_0^\omega u^4(0,x) dx\right\}.$$

Therefore, $\sup_{t} \int_{0}^{\omega} u^{4} dx$ is bounded since $\sup_{t} \int_{0}^{\omega} u^{2} dx$ is bounded from (3.77). Induction will imply that for $k \geq 1$, $\sup_{t} \int_{0}^{\omega} u^{2^{k}} dx$ is bounded. By taking the $\frac{1}{2^{k}}$ power of both sides and passing to the limit, the L^{∞} estimate is obtained. Therefore, for any T > 0, there exists a M, which depends on $||u(0, x)||_{L^{\infty}}$, such that

$$\|u(t)\|_{\infty} \le M, \quad 0 \le t \le T.$$

Since

$$\begin{aligned} \frac{d}{dt} \int_0^{\omega} u_x^2 dx &= -\int_0^{\omega} \left[u_{xx}^2 + c u_x u_{xx} + \varepsilon (u^2 - 1) u_x u_{xx} + u u_{xx} \right] dx \\ &= -\int_0^{\omega} [u_{xx}^2 dx + \varepsilon u^2 u_x u_{xx} dx - u_x^2] dx, \\ &\leq -\int_0^{\omega} u_{xx}^2 dx + M^2 \varepsilon \int_0^{\omega} |u_x u_{xx}| dx + \int_0^{\omega} u_x^2 dx, \\ &\leq -\int_0^{\omega} u_{xx}^2 dx + \frac{M^4 \varepsilon^2}{2} \int_0^{\omega} u_x^2 dx + \frac{1}{2} \int_0^{\omega} u_{xx}^2 dx + \int_0^{\omega} u_x^2 dx, \\ &\leq -\frac{1}{2} \int_0^{\omega} u_{xx}^2 dx + \left(1 + \frac{M^4 \varepsilon^2}{2}\right) \int_0^{\omega} u_x^2 dx, \end{aligned}$$

or

$$\int_0^\omega u_x^2(t,x)dx \le \exp\left[\left(1 + M^4\varepsilon^2/2\right)t\right]\int_0^\omega u_x^2(0,x)dx$$

the L^2 norm of u_x exists for all $0 \le t \le T$. Therefore u(t) exists in $H_{per}^1(0,\omega)$ for all $t \ge 0$.

The linear variational equation of (3.75) at $\phi(x+ct)$ is

$$v_{t} = v_{xx} + (\varepsilon(\phi^{2} - 1) + c) v_{x} + (1 + 2\varepsilon\phi\phi_{x})v, \quad 0 < x < \omega, \quad t > 0,$$

$$v(t, 0) = v(t, \omega), \quad v_{x}(t, 0) = v_{x}(t, \omega), \quad t \ge 0,$$

$$v(0, x) = v_{0}(x), \quad 0 \le x \le \omega.$$

(3.82)

It can be shown that u(t) exists in $H^1_{per}(0,\omega)$ for all $t \ge 0$ since $\phi(x+ct) \in C^2(0,\omega)$. The pointwise representation of the kth compound equation of (3.82) defined on $\bigwedge^k G$ is

$$w_t = \sum_{j=1}^k w_{x_j x_j} + \sum_{j=1}^k a(t, x_j) w_{x_j} + \sum_{j=1}^k b(t, x_j) w, \qquad (3.83)$$

where

$$a(t, x) = \varepsilon(\phi^2(x + ct) - 1) + c,$$

$$b(t, x) = 1 + 2\varepsilon\phi(x + ct)\phi_x(x + ct)$$

Consider a Lyapunov function

$$V(t) = \frac{1}{2} \int_{(0,2\pi)^k} w^2 = \frac{1}{2} \int_{(0,2\pi)^k} w^2(t, x_1, x_2, \cdots, x_k) dx_1 dx_2 \cdots dx_n.$$

Then

$$\begin{aligned} \frac{d^{+}V}{dt} &= \int_{(0,2\pi)^{k}} ww_{t} \\ &= \int_{(0,2\pi)^{k}} \left(w \sum_{j=1}^{k} w_{x_{j}x_{j}} + \sum_{j=1}^{k} a(t,x_{j})ww_{x_{j}} + \sum_{j=1}^{k} b(t,x_{j})w^{2} \right) \\ &= \int_{(0,2\pi)^{k}} \left(w \sum_{j=1}^{k} w_{x_{j}x_{j}} + \sum_{j=1}^{k} \left(\varepsilon(\phi^{2}(x_{j}+ct)-1)+c\right)ww_{x_{j}} \right) \\ &+ \int_{(0,2\pi)^{k}} \left(kw^{2} + 2\varepsilon \sum_{j=1}^{k} \phi(x_{j}+ct)\phi_{x}(x_{j}+ct)w^{2} \right) \\ &= \int_{(0,2\pi)^{k}} \left(w \sum_{j=1}^{k} w_{x_{j}x_{j}} + kw^{2} \right) + \varepsilon \int_{\Omega} \sum_{j=1}^{k} \phi(x_{j}+ct)\phi_{x}(x_{j}+ct)w^{2} \\ &\leq 2 \left(\sum_{j=1}^{k} \lambda_{j} + k + k\varepsilon \|\phi(x+ct)\phi_{x}(x+ct)\|_{\infty} \right) \cdot 2V \\ &= 2 \left(\sum_{j=1}^{k} \lambda_{j} + k + k\varepsilon \|\phi\phi_{x}\|_{\infty} \right) V, \end{aligned}$$

where

$$\lambda_{2j-1} = \lambda_{2j} = -j^2 \left(\frac{2\pi}{\omega}\right)^2, \quad j = 1, 2, \cdots.$$

Thus, for any k characteristic multipliers $\mu_1, \mu_2, \cdots, \mu_k$,

$$|\mu_1\mu_2\cdots\mu_k| \leq \exp\left(\sum_{j=1}^k \lambda_j + k + k\varepsilon \|\phi\phi_x\|_{\infty}\right).$$

Since 1 is a characteristic multiplier of the periodic solution $\phi(x + ct)$, say $\mu_1 = 1$, the other k - 1 characteristic multipliers

$$|\mu_2 \mu_2 \cdots \mu_k| \le \exp\left(\sum_{j=1}^k \lambda_j + k + k\varepsilon \|\phi \phi_x\|_{\infty}\right).$$
(3.84)

Case 1: $\varepsilon = 0.2$

An approximation of the periodic solution of the ordinary differential equation (3.75) is shown in Figure 3.1, which implies $2\|\phi\phi_x\|_{\infty} \leq (2.2)^2$. The Urabe formula of the period implies that $\omega \approx 6.3$. Then



Figure 3.1: $\varepsilon = 0.2$. The black trajectory is $u^2 + (u')^2 = 2.2^2$.

Theorem 3.32 implies the following corollary.

Corollary 3.34. If $\varepsilon = 0.2$, then the periodic solution $\phi(x+ct)$ of the reaction diffusion equation (3.75) has a stable manifold with codimension at most 1 in G.

Case 2: $\varepsilon = 1$

An approximation of the periodic solution of the ordinary differential equation (3.75) is shown in Figure 3.2, which implies $\|\phi\|_{\infty} \leq A \approx 2.009$, $2\|\phi\phi_x\|_{\infty} \leq (2.9)^2$. The Urabe formula of the period implies that $\omega \approx 6.687$. Then

$$\sum_{j=1}^{9} \lambda_j + 9 + 9\varepsilon \|\phi\phi_x\|_{\infty} \le -85 \left(\frac{2\pi}{\omega}\right)^2 + 9 + 9 \cdot 1 \cdot 2.009 \cdot 2.9 \approx -13.609.$$

Theorem 3.32 implies the following corollary.



Figure 3.2: $\varepsilon = 1$. The black trajectory is $u^2 + (u')^2 = 2.9^2$.

Corollary 3.35. If $\varepsilon = 1$, then the periodic solution $\phi(x+ct)$ of the reaction diffusion equation (3.75) has a stable manifold with codimension at most 7 in G.

Case 3: $\varepsilon = 10$

An approximation of the periodic solution of the ordinary differential equation (3.75) is shown in Figure 3.3, which implies $\|\phi\|_{\infty} \leq A \approx 2.0145$, $\|\phi\phi_x\|_{\infty} \leq 16$. The Urabe formula of the period implies that $\omega \approx 19.1550$. Then

$$\sum_{j=1}^{1} 28\lambda_j + 128 + 128\varepsilon \|\phi\phi_x\|_{\infty} \le -178880 \left(\frac{2\pi}{\omega}\right)^2 + 9 + 9 \cdot 1 \cdot 2.009 \cdot 2.9 \approx -488.181.$$



Figure 3.3: $\varepsilon = 10$. The black trajectory is $u^2 + (u')^2 = 14.5^2$.

Theorem 3.32 implies the following corollary.

Corollary 3.36. If $\varepsilon = 10$, then the periodic solution $\phi(x+ct)$ of the reaction diffusion equation (3.75) has a stable manifold with at most codimension 126 in G.

For the case $c = \varepsilon > 0$, it can be shown that $u = \phi(x + \varepsilon t)$ is the only nonconstant periodic solution of the reaction diffusion equation (3.76). The full discussion is as follows. If $c = \varepsilon > 0$, then the reaction diffusion equation (3.76) is

$$u_{t} = u_{xx} + \varepsilon u^{2}u_{x} + u, \quad 0 < x < \omega, \quad t > 0,$$

$$u(t,0) = u(t,\omega), \quad u_{x}(t,0) = u_{x}(t,\omega), \quad t \ge 0,$$

$$u(0,x) = u_{0}(x), \quad 0 \le x \le \omega.$$

(3.85)

The steady state solutions of (3.85) satisfies

$$0 = u_{xx} + \varepsilon u^2 u_x + u, \quad 0 < x < \omega, \quad t > 0, u(0) = u(\omega), \quad u_x(0) = u_x(\omega).$$
(3.86)

Let $v = u_x$. Then (3.86) is equivalent to

$$\begin{split} & u_x = v := f(u, v), \\ & v_x = -\varepsilon u^2 v + u := g(u, v), \quad 0 < x < \omega, \quad t > 0, \\ & u(0) = u(\omega), \quad v(0) = v(\omega). \end{split}$$

Since

$$\operatorname{div}(f,g) = f_u(u,v) + g_v(u,v) = -\varepsilon u^2,$$

Bendixson's Condition implies that the only solution of (3.86) is u = 0. The linear variational equation of (3.86) at u = 0 is

$$egin{aligned} v_t &= v_{xx} + v, \quad 0 < x < \omega, \ v(0) &= v(\omega), \quad v_x(0) = v_x(\omega), \end{aligned}$$

whose eigenvalue problem

$$\begin{split} \lambda \phi &= \phi'' + \phi, \quad 0 < x < \omega, \\ \phi(0) &= \phi(\omega), \quad \phi(0) = \phi(\omega), \end{split}$$

has solutions

$$\begin{split} \lambda_0 &= 1, & \phi_0(x) = 1, \\ \lambda_{2j-1} &= -j^2 + 1, & \phi_{2j-1}(x) = \cos\left(\frac{2\pi j}{\omega}x\right), \\ \lambda_{2j-1} &= -j^2 + 1, & \phi_{2j}(x) = \sin\left(\frac{2\pi j}{\omega}x\right), \end{split}$$

where $j = 1, 2, 3, \cdots$.

Corollary 3.37. The solution u = 0 of the reaction diffusion equation (3.85) has a 2-dimensional center manifold, 1-dimensional unstable manifold and a stable manifold with codimension 3.

The result of Angenent and Fiedler [4] implies that the periodic solutions of (3.85) are in form of $u = u(x - c_1 t)$. Let $z = x - c_1 t$. Then

$$-c_1 \frac{du}{dz} = \frac{d^2 u}{dz^2} + \varepsilon u^2 \frac{du}{dz} + u,$$

$$u(0) = u(\omega), \quad u_x(0) = u_x(\omega).$$

or

$$\frac{d^2u}{dz^2} + (\varepsilon u^2 + c_1)\frac{du}{dz} + u = 0,$$

$$u(0) = u(\omega), \quad u_x(0) = u_x(\omega),$$
(3.87)

which is equivalent to

$$\begin{aligned} \frac{du}{dz} &= v := f^1(u, v), \\ \frac{dv}{dz} &= -(\varepsilon u^2 + c_1)v - u := f^2(u, v) \\ u(0) &= u(\omega), \quad v(0) = v(\omega). \end{aligned}$$

Since

$$\operatorname{div}(f,g) = f_u^1(u,v) + f_v^2(u,v) = -(\varepsilon u^2 + c_1),$$

Bendixson's condition implies that the only solution of (3.87) is u = 0 if $c_1 \ge 0$. Thus the periodic solutions of (3.85) are in form of $u = u(x + c_1 t)$ where $c_1 > 0$. Let $p = \sqrt{\frac{c_1}{\varepsilon}} u$. Then

$$\begin{split} &\sqrt{\frac{c_1}{\varepsilon}}\frac{d^2p}{dz^2} + \left(\varepsilon\frac{c_1}{\varepsilon}p^2 - c\right)\sqrt{\frac{c_1}{\varepsilon}}\frac{dp}{dz} + \sqrt{\frac{c_1}{\varepsilon}}p = 0,\\ &p(0) = p(\omega), \ \ p_x(0) = p_x(\omega), \end{split}$$

 \mathbf{or}

$$\frac{d^2 p}{dz^2} + c_1 \left(p^2 - 1 \right) \frac{dp}{dz} + p = 0,$$

$$p(0) = p(\omega), \quad p_x(0) = p_x(\omega).$$
(3.88)

The result of van de Pol equation implies that equation

$$\frac{d^2p}{dz^2} + c_1 \left(p^2 - 1\right) \frac{dp}{dz} + p = 0$$

has an ω -periodic solution if and only if $c_1 = \varepsilon$. Therefore, the only periodic solution of the reaction diffusion equation (3.85) is $\phi(x + \varepsilon t)$ where ϕ is a solution of the van der Pol equation

$$\frac{d^2p}{dz^2} + \varepsilon \left(p^2 - 1\right) \frac{dp}{dz} + p = 0.$$

The result in Angenent and Fiedler [4] implies that the ω -limit set of any bounded solution of the reaction diffusion equation (3.85) is either u = 0 or $u = \phi(x + \varepsilon t)$ whose stability is given by Corollary 3.34- Corollary 3.37.

Example 3.4. Consider a scalar reaction diffusion equation

$$\begin{cases} u_t = u_{xx} + u + \varepsilon u(1 - u^2 - u_x^2) + cu_x, & 0 < x < 2\pi, \quad t > 0, \\ u(t,0) = u(t,2\pi), & u_x(t,0) = u_x(t,2\pi), \end{cases}$$
(3.89)

where $\varepsilon, c > 0$. Similar arguments used in Chapter 2 show that the initial value problem associated to (3.89) generates a global semiflow on the Sobolev space $X = H_{per}^1(0, 2\pi) = \{\phi \in L^2(0, 2\pi) : \phi' \in L^2(0, 2\pi), \phi(0) = \phi(2\pi), \phi'(0) = \phi'(2\pi)\}$. Since $u = \cos(x)$ is a solution of

$$\left\{ \begin{array}{ll} u_t = u_{xx} + u + \varepsilon u(1 - u^2 - u_x^2), \quad 0 < x < 2\pi, \quad t > 0, \\ u(t,0) = u(t,2\pi), \quad u_x(t,0) = u_x(t,2\pi), \end{array} \right.$$

equation (3.89) has a $\frac{2\pi}{c}$ -periodic solution $u = \cos(x+ct)$. In fact, the circle of equilibria $\cos(x+a), a \in \mathbb{R}$ of (3.4) becomes the periodic orbit $u = \cos(x+ct)$ of (3.89).

The linear variational equation of (3.89) at $u = \cos(x + ct)$ is

$$v_t = v_{xx} + (c + \varepsilon \sin(2x + 2ct))v_x + (1 - \varepsilon - \varepsilon \cos(2x + 2ct))v,$$

$$v(t, 0) = v(t, 2\pi), \quad v_x(t, 0) = v_x(t, 2\pi),$$
(3.90)

which has solutions

$$e_1 = \sin(x + ct)$$
 and $e_2 = e^{-2\epsilon t} \cos(x + ct)$,

with the characteristic multipliers

$$\mu_1 = 1$$
 and $\mu_2 = e^{-2\varepsilon \cdot \frac{2\pi}{c}} < 1$.

The linear variational equation of (3.4) at $u = \cos(x)$ is

$$\begin{aligned} v_t &= v_{xx} + \varepsilon \sin(2x)v_x + (1 - \varepsilon - \varepsilon \cos(2x))v, \ 0 < x < 2\pi, \ t > 0, \\ v(t,0) &= v(t,2\pi), \ v_x(t,0) = v_x(t,2\pi), \ t \ge 0, \end{aligned}$$

whose eigenvalue problem is given by

$$\lambda \phi = \phi'' + \varepsilon \sin(2x)\phi' + (1 - \varepsilon - \varepsilon \cos(2x))\phi,$$

$$\phi(0) = \phi(2\pi), \quad \phi'(0) = \phi'(2\pi).$$
(3.91)

If (3.91) has an eigenvalue λ with the corresponding eigenfunction $\phi(x)$, then (3.90) has a characteristic multiplier $e^{\lambda \cdot \frac{2\pi}{c}}$ with a corresponding nonzero solution $e^{\lambda t}\phi(x+ct)$. In fact, substitute $v = e^{\lambda t}\phi(x+ct)$ into (3.90) to get

$$\begin{aligned} \lambda e^{\lambda t} \phi(x+ct) + e^{\lambda t} \phi_x(x+ct)c \\ &= e^{\lambda t} \phi_{xx}(x+ct) + (c+\varepsilon \sin(2x+2ct))e^{\lambda t} \phi_x(x+ct) \\ &+ (1-\varepsilon - \varepsilon \cos(2x+2ct))e^{\lambda t} \phi(x+ct), \end{aligned}$$

or

$$\lambda\phi(x+ct) = \phi_{xx}(x+ct) + \varepsilon \sin(2x+2ct)\phi_x(x+ct) + (1-\varepsilon - \varepsilon \cos(2x+2ct))\phi(x+ct),$$

which always holds from (3.91). The stability of the periodic solution $u = \cos(x + ct)$ of (3.89) can be discussed by studying the stability of the steady state solution $u = \cos(x)$ of (3.4), which has been studied in Example 2.1 of Section 2.2. In particular, Corollary 2.13 and Corollary 2.14 imply the following results on the estimate of the stability of $u = \cos(x + ct)$.

Corollary 3.38. If $0 < \varepsilon < 2(\sqrt{202} - 14)$, then $u = \cos(x + ct)$ has a 1-dimensional center manifold, a 1-dimensional unstable manifold and a stable manifold with codimension 2.

Corollary 3.39. If $2(\sqrt{202} - 14) \le \varepsilon < 2(13\sqrt{2} - 18)$, then $u = \cos(x + ct)$ has a stable manifold with codimension at most 3.

Chapter 4 Convergence Theorems

In this chapter the implications of various types of stability of a solution of a differential equation for the structure of its omega limit set are considered.

A fundamental result of this type is the Poincaré-Bendixson theorem for a planar autonomous system of ordinary differential equations. The ω -limit set of a bounded solution of such a system is a periodic orbit if it does not contain an equilibrium. Massera's theorem [69] infers the existence of a periodic solution to a nonautonomous time-periodic scalar differential equation from the existence of a bounded solution. Other expositions and extensions of the Poincaré-Bendixson theorem were obtained by Hirsch [45], R. A. Smith [98, 99, 101], Mallet-Paret and H. L. Smith [66]. An application of the Poincaré-Bendixson theorem is that it can be used to deduce the existence of a periodic solution in the case when no equilibrium occurs in the ω -limit set of a bounded solution.

In dynamics, it is natural to enquire how much of the nontransient behaviour can be detected from an analysis of an individual orbit and its relationship with its neighbours. Sell [96] proved for a general semiflow on a metric space that a Lagrange stable orbit has a phase asymptotically stable periodic orbit as its omega limit set if the orbit itself is phase asymptotically stable. A Lagrange stable orbit means that the closure of the orbit is compact. Good expositions of Sell's results may be found in Cronin [22] Chapter 6 and Saperstone [93] Chapter III. Yoshizawa [112] also discussed these results and extended the application to functional differential equations. Pliss [89] established a closely related result for autonomous differential equations in \mathbb{R}^n where the stability requirements are somewhat different from those of Sell: Lyapunov stability is not required but a certain uniformity is imposed on the manner in which the orbit attracts its neighbours. Li and Muldowney [54] and Muldowney [79] extended these results in the context of dynamical systems on a general metric space with greatly simplified proofs. A good summary of these results may also be found in Muldowney [80].

Another result of Muldowney in [80] is to deduce the existence of an equilibrium, even when the algebraic equations yielding the equilibria can not be solved explicitly. For example, Muldowney proved that, for an autonomous ordinary differential equation

$$\frac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^n,$$

the ω -limit set of a bounded solution u(t) is a stable hyperbolic equilibrium if and only if the linear variational equation at u(t),

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(u(t))v,$$

is uniformly asymptotically stable. In particular, if the bounded solution $u(t) = u^*$ is an equilibrium, it reduces to the familiar observation that u^* is stable hyperbolic if and only if the linear variational equation at u^* is uniformly asymptotically stable. Li and Muldowney [55] and Muldowney [79] also proved that, if the ω -limit set of a bounded solution u(t) contains no equilibrium, it is a stable hyperbolic periodic orbit if and only if the second compound differential equation

$$\frac{dw}{dt} = \frac{\partial f}{\partial u}^{[2]}(u(t))w$$

is uniformly asymptotically stable. This is in fact the necessary and sufficient condition cited in Theorem 3.15, Section 3.3.3 for stable hyperbolicity when u(t) is itself a periodic solution.

An infinite dimensional version of the Poincaré-Bendixson theorem for a scalar reaction diffusion equation was studied by Henry [43], Brunovský and Fiedler [10, 11], Massatt [68], Matano [71], Angenent and Fielder [4], Fiedler and Mallet-Paret [30]. It was proved by Fiedler and Mallet-Paret that, for a general scalar reaction diffusion equation

$$u_t = u_{xx} + f(x, u, u_x), \quad x \in S^1 = \mathbb{R}/2\pi\mathbb{Z}, \ f \in C^2,$$

the ω -limit set of any bounded solution satisfies exactly one of the following alternatives:

(i) it consists in precisely one periodic solution, or

(ii) for every orbit in the ω -limit set, its ω -limit set and α -limit set are contained in the set of steady state solutions.

In this chapter, an extension is given of the results in [55, 79, 80] to an infinite dimensional differential equation

$$\frac{du}{dt} + Au = f(u), \quad u(\cdot) \in X, \tag{4.1}$$

where A is a sectorial operator in a Banach space X. The characterization of the stability of steady state solutions in terms of stability of linearizations and orbital stability of periodic orbits in terms of stability of second compound differential equations, discussed in Chapter 2 and Chapter 3 respectively, are special cases of this chapter.

4.1 Existence of Stable Steady State Solutions

The following notation, definitions, and lemmas are taken from Henry [42], page 53-54, page 91-92 and page 98-104 and Smoller [103], page 114-122.

Let A be a sectorial operator in a Banach space X and f be continuously differentiable from X^{α} into X where $0 \leq \alpha < 1$. Let $\mathcal{D}(A)$ denote the domain of A. Consider an initial value problem

$$\frac{du}{dt} + Au = f(u), \ u \in X,$$

$$u(0) = u_0.$$
(4.2)

In the following, let $u(t) = u(t, u_0)$ denote the solution of (4.2). A solution is a steady state solution or (an equilibrium) if $u = u^* \in \mathcal{D}(A)$ and $Au^* = f(u^*)$.

Definition 4.1. A solution $u(t, u_0)$ of (4.2) on $[0, \infty)$ is said to be *stable* if, for each $\varepsilon > 0$, there exists a $\delta > 0$ such that any solution $u(t, u_1)$ with $||u_1 - u_0||_{\alpha} < \delta$ exists on $[0, \infty)$ and satisfies $||u(t, u_1) - u(t, u_0)||_{\alpha} < \varepsilon$ for all $t \ge 0$. A solution $u(t, u_0)$ of (4.2) on $[0, \infty)$ is said to be uniformly stable if there exists a $\delta > 0$ such that for each $\varepsilon > 0$, any solution $u(t, u_1)$ with $||u_1 - u_0||_{\alpha} < \delta$ exists and satisfies $||u(t, u_1) - u(t, u_0)||_{\alpha} < \varepsilon$ for all $t \ge 0$. A solution $u(t, u_0)$ of (4.2) on $[0, \infty)$ is said to be uniformly asymptotically stable if it is uniformly stable and there exists a $\delta_0 > 0$ and for each $\varepsilon > 0$, there is a corresponding $T = T(\varepsilon)$ such that if $||u_1 - u_0||_{\alpha} \le \delta_0$, then $||u(t, u_1) - u(t, u_0)||_{\alpha} < \varepsilon$ for all $t \ge T$.

The linear variational equation of (4.2) at the steady state solution $u(t) = u^*$ is

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u^*)v. \tag{4.3}$$

Lemma 4.1. Let f be continuously differentiable from X^{α} into X in a neighbourhood of a steady state solution u^* of (4.2). If the spectrum of $A - \frac{\partial f}{\partial u}(u^*)$ lies in $\{\operatorname{Re} \lambda > \beta\}$ for some $\beta > 0$, then the steady state solution u^* of (4.2) is uniformly asymptotically stable in X^{α} . More precisely, there exist $\rho > 0, M \ge 1$ such that if $||u_0 - u^*||_{\alpha} \le \frac{\rho}{2M}$, then there is a unique solution of

$$\frac{du}{dt} + Au = f(u), \ u \in X,$$
$$u(0) = u_0,$$

satisfying for $0 \leq t < \infty$,

$$||u(t, u_0) - u^*||_{\alpha} \le 2Me^{-\beta t} ||u_0 - u^*||_{\alpha}.$$

Lemma 4.2. Let f be continuously differentiable from X^{α} into X in a neighbourhood of a steady state solution u^* of (4.2). Let $L = A - \frac{\partial f}{\partial u}(u^*)$. If $\sigma(L) \cap \{ \text{Re } \lambda < 0 \}$ is a nonempty spectral set, then the equilibrium u^* of (4.2) is unstable. Specifically, there exist $\varepsilon_0 > 0$ and $\{u_n, n \ge 1\}$ with $||u_n - u^*||_{\alpha} \to 0$ as $n \to \infty$, but for all n

$$\sup_{t\geq 0} \|u(t,u_n) - u^*\|_{\alpha} \geq \varepsilon_0 > 0.$$

Here the supremum is taken over the maximal interval of existence of $u(t, u_n)$.

Definition 4.2. For any $u_0 \in X^{\alpha}$, the positive orbit of $u(t, u_0)$ is $C_+(u_0) = \{u(t, u_0) : t \ge 0\}$ and the ω -limit set of u_0 is

$$\omega(u_0) = \{ u \in X^{\alpha} : \text{there exist } t_n \to \infty \text{ such that } u(t_n, u_0) \to u \}.$$
Remark 4.1. If $u_1 \in C_+(u_0)$, then $\omega(u_1) = \omega(u_0)$. Also

$$\omega(u_0) = \bigcap_{t \ge 0} \operatorname{cl} \, C_+(u(t, u_0)),$$

where cl denotes the topological closure.

Lemma 4.3. If $C_+(u_0)$ lies in a compact set in X^{α} , then $\omega(u_0)$ is nonempty, closed, connected, invariant and $dist(u(t, u_0), \omega(u_0)) \to 0$ as $t \to \infty$.

The following Lemma 4.4, Lemma 4.5 and Lemma 4.6 are taken from Smoller [103], page 114-122.

Lemma 4.4. Let $u(t, u_0)$ be a solution of (4.2) which exists on [0, T). Then there exists a neighborhood N of u_0 such that if $u_1 \in N$, there is a solution $u(t, u_1)$ of (4.2) which exists on [0, T). Moreover, there is a constant c > 0such that for all such $u_1 \in N$,

$$\|u(t, u_1) - u(t, u_0)\|_{\alpha} \le c \|u_1 - u_0\|_{\alpha}.$$
(4.4)

Let $u(t, u_0)$ be a solution of (4.2) which exists on [0, T). Let N be as in Lemma 4.4. Define the solution operator $(t, u_0) \mapsto S(t)u_0$ on the set $[0, T) \times X^{\alpha}$ such that the solution $u(t, u_0) = S(t)u_0$ satisfies $u(0, u_0) = u_0$.

Lemma 4.5. Let $u(t, u_0)$ be a solution of (4.2) and N be as in Lemma 4.4. If $u_1 \in N$, then $S(t), 0 \leq t \leq T$, is (Fréchet) differentiable, and if $u_0 \in N$, then $v(t, v_0) = dS(t)_{u_0}v_0$ solves the linear variational equation of (4.2) at $u(t, u_0)$,

$$\frac{dv}{dt} = Av + \frac{\partial f}{\partial u}(u(t, u_0))v,$$
$$v(0) = v_0.$$

Lemma 4.6. S(t) is continuous (Fréchet) differentiable on N.

For an autonomous finite dimensional ordinary differential equation, a necessary and sufficient condition for the ω -limit set to be a stable hyperbolic

equilibrium was proved by Muldowney [80]. Consider an autonomous differential equation

$$\frac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^n, \tag{4.5}$$

where $f \in C^1(\mathbb{R}^n, \mathbb{R}^n)$. This defines a semiflow $(t, u_0) \mapsto \phi(t, u_0)$ on the set $\mathbb{R} \times \mathbb{R}^n$ such that the solution $u(t, u_0) = \phi(t, u_0)$ satisfies $u(0, u_0) = u_0$. The linear variational equation of (4.5) at a solution $u(t, u_0)$ is

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(u(t, u_0))v.$$
(4.6)

Because of linearity, every solution has the same stability properties as the zero solution. Therefore, it is permissible to say that a linear system is stable, uniformly stable, and so on.

Proposition 4.7. The equation (4.6) is uniformly asymptotically stable if and only if there exist constants $K, \sigma > 0$ such that if $u \in C_+(u_0)$,

$$\left\|\frac{\partial}{\partial u}\phi(t,u)\right\| \le Ke^{-\sigma t}, \quad for \ t \ge 0.$$

Definition 4.3. An equilibrium u^* , $f(u^*) = 0$, is stable hyperbolic if $\operatorname{Re} \lambda < 0$ for every eigenvalue of $\frac{\partial f}{\partial u}(u^*)$. This is equivalent to the uniform asymptotic stability of (4.6) with $u(t, u_0) = u^*$.

Theorem 4.8. Suppose that $u(t, u_0)$ is a bounded solution of (4.5). Then $\lim_{t \to +\infty} u(t, u_0) = u^*$, where u^* is a stable hyperbolic equilibrium if and only if the linear variational equation (4.6) of (4.5) with respect to $u(t, u_0)$ is uniformly asymptotically stable.

The generalizations of Definition 4.3 and Theorem 4.8 for a differential equation (4.2) in a Banach space X are as follows.

Definition 4.4. A steady state solution u^* is said to be stable hyperbolic if the spectrum of $A - \frac{\partial f}{\partial u}(u^*)$ lies in $\{Re\lambda > \beta\}$ for some $\beta > 0$.

Remark 4.2. Definition 4.4 is equivalent to Definition 11.19 in Smoller [103], page 120. Theorem 11.20 in Smoller [103], page 120, shows that the hyperbolic stability of u^* is equivalent to the uniform asymptotic stability of (4.3).

Definition 4.5. The equation

$$\frac{dv}{dt} = Av + \frac{\partial f}{\partial u}(u(t, u_0))v,$$

$$v(0) = v_0$$
(4.7)

is said to be uniformly asymptotically stable if there exist constants $K, \sigma > 0$ such that $||dS(t)_u||_{\alpha} \leq Ke^{-\sigma t}$ for $t \geq 0$ if $u \in C_+(u_0)$.

Theorem 4.9. Suppose that $u(t, u_0)$ is a solution of (4.2) and that $C_+(u_0) = \{u(t, u_0), t \geq 0\}$ is relatively compact in X^{α} with $0 < \alpha < 1$. Then $\lim_{t \to +\infty} u(t, u_0) = u^*$, where u^* is a stable hyperbolic steady state solution if and only if the linear variational equation (4.3) of (4.2) with respect to $u(t, u_0)$ is uniformly asymptotically stable.

Theorem 4.9 will be proved in Section 4.1.1.

Remark 4.3. For a steady state solution $u = u^*$, its omega limit set is itself and thus Theorem 2.8 in Section 2.2 is a special case of Theorem 4.9.

Remark 4.4. An example of a diffusive epidemiology disease model will be discussed in Chapter 6 as an application of Theorem 4.9.

4.1.1 **Proof of Theorem 4.9**

To prove Theorem 4.9, a particular upper bound of the estimate in the inequality in Lemma 7.1.1 in Henry [42] is given in Lemma 4.10.

Lemma 4.10. Suppose that $a, b \ge 0, \beta > 0$ and u(t) is nonnegative and locally integrable on $0 \le t < \infty$ with

$$u(t) \le a + b \int_0^t (t-s)^{\beta-1} u(s) ds.$$

Then

$$u(t) \le \left(\left[\frac{1}{\beta} \right] + 1 \right) a e^{2\rho t}, \tag{4.8}$$

where $\rho = (b\Gamma(\beta))^{\frac{1}{\beta}}$ and $[\cdot]$ is the greatest integer function.

Proof. Lemma 7.1.1 in Henry [42] shows that

$$u(t) \le a \sum_{n=0}^{\infty} \frac{(b\Gamma(\beta)t^{\beta})^n}{\Gamma(n\beta+1)}.$$
(4.9)

An alternative proof of the above inequality can be obtained by first showing that $a \sum_{n=0}^{\infty} \frac{(b\Gamma(\beta)t^{\beta})^n}{\Gamma(n\beta+1)}$ is a solution of $v(t) = a + b \int_0^t (t-s)^{\beta-1} v(s) ds$, and then showing that $u(t) \leq v(t)$ for all $0 \leq t < \infty$. Let $\rho = (b\Gamma(\beta))^{\frac{1}{\beta}}$. Notice that

$$([\beta n])! \leq \Gamma(\beta n), \quad t^{\beta n} \leq \begin{cases} t^{[\beta n]}, & 0 \leq t < 1, \\ t^{[\beta n]+1}, & t \geq 1. \end{cases}$$

If $\rho t \geq 1$, then

$$a\sum_{n=0}^{\infty}\frac{(\rho t)^{\beta n}}{\Gamma(\beta n)} \le a\sum_{n=0}^{\infty}\frac{(\rho t)^{[\beta n]+1}}{([\beta n])!} \le \left(\left[\frac{1}{\beta}\right]+1\right)a\rho t\sum_{k=0}^{\infty}\frac{(\rho t)^{k}}{k!} \le \left(\left[\frac{1}{\beta}\right]+1\right)a\rho te^{\rho t},$$

and thus,

$$u(t) \le \left(\left[\frac{1}{\beta}\right] + 1\right) a e^{2\rho t}.$$

If $0 \leq \rho t < 1$, then

$$a\sum_{n=0}^{\infty} \frac{(\rho t)^{\beta n}}{\Gamma(\beta n)} \le a\sum_{n=0}^{\infty} \frac{(\rho t)^{[\beta n]}}{([\beta n])!} \le \left(\left[\frac{1}{\beta}\right] + 1\right) a\sum_{k=0}^{\infty} \frac{(\rho t)^k}{k!} \le \left(\left[\frac{1}{\beta}\right] + 1\right) ae^{\rho t},$$

and thus,

$$u(t) \le \left(\left[\frac{1}{\beta}\right] + 1\right) a e^{2\rho t}.$$

Therefore for $0 \leq t < \infty$,

$$u(t) \le \left(\left[\frac{1}{\beta}\right] + 1\right) a e^{2\rho t}.$$

The proof of Theorem 4.9 adapts the proof of Theorem 11.12 in Smoller [103], page 121-122 and Theorem 5.1 in Muldowney [80]. **Proof of Theorem 4.9:**

Proof. Suppose that $\lim_{t \to +\infty} u(t, u_0) = u^*$ and u^* is a stable hyperbolic steady state solution. Then the spectrum of $L = A - \frac{\partial f}{\partial u}(u^*)$ lies in $\{\operatorname{Re} \lambda > \beta\}$ for some $\beta > 0$. If $0 < \beta < \beta_1 < \operatorname{Re} \sigma(L)$, there exists $M_1 \ge 1$ such that for $t > 0, v \in X^{\alpha}$,

$$\|e^{-Lt}v\|_{\alpha} \le M_1 e^{-\beta_1 t} \|v\|_{\alpha},$$

$$\|e^{-Lt}v\|_{\alpha} \le M_1 t^{-\alpha} e^{-\beta_1 t} \|v\|.$$
 (4.10)

In fact, $dS(t)_{u^*} = e^{-Lt}$. Since f is continuous differentiable from X^{α} in to X, choose T > 0 and an open set $N = \{u \in X^{\alpha} : ||u - u^*||_{\alpha} < r\}$ with the property that if $u_1, u_2 \in N$ and $u(t, u_1), u(t, u_2)$ are solutions of (4.2), then there exists $M_2 \geq 1$ such that for $0 \leq s \leq T$,

$$\left\| f(u(s,u_1)) - f(u(s,u_2)) - \frac{\partial f}{\partial u}(u^*)u(s,u_1) + \frac{\partial f}{\partial u}(u^*)u(s,u_2) \right\|$$

$$\leq M_2 \|u(s,u_1) - u(s,u_2)\|_{\alpha}.$$

Rewrite (4.2) as

$$\frac{du}{dt} + \left(A - \frac{\partial f}{\partial u}(u^*)\right)u = f(u) - \frac{\partial f}{\partial u}(u^*)u.$$

Then any solution $u(t, u_0)$ of (4.2) satisfies

$$u(t, u_0) = e^{-Lt}u_0 + \int_0^t e^{-L(t-s)} \left(f(u(s, u_0)) - \frac{\partial f}{\partial u}(u^*)u(s, u_0) \right) ds.$$

Thus,

$$\begin{split} \|u(t,u_{1}) - u(t,u_{2})\|_{\alpha} \\ &\leq \|e^{-Lt}(u_{1} - u_{2})\|_{\alpha} \\ &+ \int_{0}^{t} \left\|e^{-L(t-s)} \left(f(u(s,u_{1})) - f(u(s,u_{2})) - \frac{\partial f}{\partial u}(u^{*})u(s,u_{1}) + \frac{\partial f}{\partial u}(u^{*})u(s,u_{2})\right)\right\|_{\alpha} ds \\ &\leq M_{1}e^{-\beta_{1}t}\|u_{1} - u_{2}\|_{\alpha} \\ &+ \int_{0}^{t} M_{1}(t-s)^{-\alpha}e^{-\beta_{1}(t-s)} \cdot \\ &\cdot \left\|f(u(s,u_{1})) - f(u(s,u_{2}) - \frac{\partial f}{\partial u}(u^{*})u(s,u_{1}) + \frac{\partial f}{\partial u}(u^{*})u(s,u_{2})\right\| ds \\ &\leq M_{1}e^{-\beta_{1}t}\|u_{1} - u_{2}\|_{\alpha} + \int_{0}^{t} M_{1}(t-s)^{-\alpha}e^{-\beta_{1}(t-s)}M_{2}\|u(s,u_{1}) - u(s,u_{2})\|_{\alpha}, \end{split}$$

Lemma 4.10 implies that there exist constants $r, M, \beta_2 > 0$ such that if $||u_i - u^*||_{\alpha} < r, i = 1, 2$, then

$$\|u(t, u_1) - u(t, u_2)\|_{\alpha} \le M e^{-\beta_2 t} \|u_1 - u_2\|_{\alpha}, \quad \text{for } 0 \le t \le T.$$
(4.11)

If $||u_1 - u^*||_{\alpha} < \frac{r}{2}$ and $||u_2 - u_1||_{\alpha} < \frac{r}{2}$, then $||u_2 - u^*||_{\alpha} < r$. Choose $0 < \gamma < \beta_2(<\beta_1)$, so that $Me^{-\beta_2 T} < e^{-\gamma T}$. For any t > 0, there exists a

nonnegative integer n_1 such that $n_1T \leq t < (n_1+1)T$. Let $t_1 = t - n_1T$. Then $0 \leq t_1 < T$. Notice that

$$\begin{aligned} \|u(nT, u_1) - u^*\|_{\alpha} &\leq Me^{-\beta_2 T} \|u((n-1)T, u_1) - u^*\|_{\alpha} \\ &\leq \|u((n-1)T, u_1 - u^*\|_{\alpha} \\ &\leq \cdots \leq \|u_1 - u^*\|_{\alpha} < \frac{r}{2}, \\ \|u(nT, u_2) - u^*\|_{\alpha} &\leq Me^{-\beta_2 T} \|u((n-1)T, u_2) - u^*\|_{\alpha} \\ &\leq \|u((n-1)T, u_2) - u^*\|_{\alpha} \\ &\leq \cdots \leq \|u_2 - u^*\|_{\alpha} < r. \end{aligned}$$

Thus from (4.11),

$$\begin{aligned} \|u(t, u_1) - u(t, u_2)\|_{\alpha} \\ &= \|u(t_1 + n_1 T, u_1) - u(t_1 + n_1 T_1, u_2)\|_{\alpha} \\ &\leq M e^{-\beta_2 t_1} \|u(n_1 T, u_1) - u(n_1 T, u_2)\|_{\alpha} \\ &\leq M e^{-\beta_2 t_1} M^{n_1} e^{-n_1 \beta_2 T} \|u_1 - u_2\|_{\alpha} \\ &\leq M e^{-\gamma (t - n_1 T)} e^{-n_1 \gamma T} \|u_1 - u_2\|_{\alpha} = M e^{-\gamma t} \|u_1 - u_2\|_{\alpha}. \end{aligned}$$

Hence if $||u_1 - u^*||_{\alpha} < \frac{r}{2}$ and $||u_2 - u_1||_{\alpha} < \frac{r}{2}$, then

$$\|u(t, u_1) - u(t, u_2)\|_{\alpha} \le M e^{-\gamma t} \|u_1 - u_2\|_{\alpha}, \quad \text{for } t \ge 0.$$
(4.12)

This implies that there exists $r_1 > 0$ such that if $u_1 \in X^{\alpha}$, $||u_1 - u^*||_{\alpha} < r_1$, then $v(t, v_0) = dS(t)_{u_1}v_0$ solves the linear variational equation of (4.2) at $u(t, u_1)$,

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u(t, u_1))v,$$

$$v(0) = v_0,$$

where $v_0 \in X^{\alpha}$ and

$$\|dS(t)_{u_1}\|_{\alpha} \le K_1 e^{-\gamma_1 t}, \quad \text{for } t \ge 0,$$
(4.13)

for some constants K_1 and $\gamma_1 > 0$. Notice that if $\lim_{t \to +\infty} u(t, u_0) = u^*$, then there exists a constant $T_1 > 0$ such that for $t \ge T_1$, $||u(t, u_0) - u^*||_{\alpha} < r_1$. Inequality (4.13) implies that if $u \in \{u(t, u_0) : t \ge T_1\}$,

$$||dS(t)_u||_{\alpha} \le K_1 e^{-\gamma_1 t}, \quad \text{for } t \ge 0.$$

Since the orbit $C_+(u_0)$ is relatively compact, Lemma 4.4 and Lemma 4.6 imply that there exist constants $K_2, \gamma_3 > 0$ such that if $u \in \{u(t, u_0) : 0 \le t \le T_1\}$,

$$||dS(t)_u||_{\alpha} \le K_2 e^{-\gamma_2 t}, \text{ for } t \in [0, T_1].$$

Let $K = \max\{K_1, K_2\}$ and $\sigma = \max\{\gamma_1, \gamma_2\}$. Then if $u \in C_+(u_0) = \{u(t, u_0) : t \ge 0\}$,

$$||dS(t)_u||_{\alpha} \le Ke^{-\sigma t}, \text{ for } t \ge 0.$$

Therefore, the linear variational equation (4.7) of (4.2) at $u(t, u_0)$ is uniformly asymptotically stable.

Conversely, suppose that the linear variational equation (4.7) of (4.2) at $u(t, u_0)$ is uniformly asymptotically stable: there exist constants $K, \sigma > 0$ such that if $u \in C_+(u_0) = \{u(t, u_0) : t \ge 0\}$,

$$\|dS(t)_u\|_{\alpha} \le Ke^{-\sigma t}, \quad \text{for } t \ge 0.$$
(4.14)

If $u(t, u_0)$ is a solution of (4.2), then

$$\frac{d^2u}{dt^2} = \frac{d}{dt}(-Au + f(u)) = -A\frac{du}{dt} + \frac{\partial f}{\partial u}\frac{du}{dt}.$$

Thus $\frac{du}{dt}(t, u_0)$ is a solution of the linear variational equation (4.7) and $\frac{du}{dt}(t, u_0) = dS(t)_{u_0} \frac{du}{dt}(0, u_0)$. Inequality (4.14) implies that $\lim_{t \to +\infty} \frac{du}{dt}(t, u_0) = 0$ and thus

$$\lim_{t \to +\infty} -Au(t, u_0) + f(u(t, u_0)) = 0.$$
(4.15)

Since $C_+(u_0)$ is relatively compact in X^{α} , $\omega(u_0)$ is nonempty and connected. If $u^* \in \omega(u_0)$, then there exists a sequence $t_n \to \infty$ as $n \to \infty$ such that $\lim_{n \to +\infty} u(t_n, u_0) = u^*$. From (4.15),

$$\lim_{n \to +\infty} -Au(t_n, u_0) + f(u(t_n, u_0)) = 0.$$

Since f is continuous and A is closed on X^{α} when $0 < \alpha < 1$,

$$\lim_{n \to +\infty} -Au(t_n, u_0) + f(u(t_n, u_0)) = -Au^* + f(u^*).$$

Thus u^* is a steady state solution of (4.2). From Lemma 4.4 and Lemma 4.6, inequality (4.14) is satisfied with $u = u^*$ which means that u^* is a stable hyperbolic steady state solution. Hence there exists an open neighbourhood

N of u^* such that any solution $u(t, u_1)$ of (4.2) with initial condition $u_1 \in N$ stays in N when t is large enough. Consequently, u^* is the unique steady state solution in N since the omega limit set $\omega(u_0)$ is connected.

4.2 Existence of Stable Periodic Solutions

In this section, conditions for an omega limit set to be a stable hyperbolic periodic solution will be studied. A result of Li and Muldowney [55] for n-dimensional ordinary differential equation is the following. Consider an autonomous differential equation

$$\frac{du}{dt} = f(u), \quad u(\cdot) \in \mathbb{R}^n, \tag{4.16}$$

where $f \in C^1(\mathbb{R}^n, \mathbb{R}^n)$. This defines a semiflow $(t, u_0) \mapsto \phi(t, u_0)$ on the set $\mathbb{R} \times \mathbb{R}^n$ such that the solution $u(t, u_0) = \phi(t, u_0)$ satisfies $u(0, u_0) = u_0$.

The linear variational equation of (4.16) at a solution $u(t, u_0)$ is

$$\frac{dv}{dt} = \frac{\partial f}{\partial u}(u(t, u_0))v. \tag{4.17}$$

The second compound differential equation of (4.17) is

$$\frac{dw}{dt} = \frac{\partial f^{[2]}}{\partial u}(u(t, u_0))w.$$
(4.18)

Theorem 4.11. Suppose that $u(t, u_0)$ is a bounded solution of (4.16) whose omega limit set $\omega(u_0)$ contains no equilibrium. Then $\omega(u_0)$ is a stable hyperbolic periodic orbit if and only if the second compound differential equation (4.18) of (4.17) is uniformly asymptotically stable.

A generalization of the sufficiency statement in Theorem 4.11 for a differential equation in a Banach space X is Theorem 4.12.

Theorem 4.12. Let A be a sectorial operator in a Banach space X and f be continuously differentiable from X^{α} into X and $0 \leq \alpha < 1$. Suppose that $u(t, u_0)$ is a solution of

$$\frac{du}{dt} + Au = f(u),

u(0) = u_0.$$
(4.19)

such that

(i) $C_+(u_0) = \{u(t, u_0) : t \ge 0\}$ is relatively compact in X^{α} ; (ii) $v^1(t) = \frac{du}{dt}(t, u_0)$ is uniformly bounded in X^{α} with respect to all $t \ge 0$;. (iii) the omega limit set $\omega(u_0)$ of $u(t, u_0)$ contains no steady state solution; (iv) the second compound differential equation

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f}{\partial u}^{[2]}(u(t, u_0))w, \qquad (4.20)$$

is uniformly asymptotically stable in $\bigwedge^2 X^{\alpha}$. Then $\omega(u_0)$ is a stable hyperbolic periodic solution in X^{α} .

The proof of Theorem 4.12 will be discussed in Section 4.2.1. The key consequence of the asymptotic stability assumption (iv) is that $v^1(t)$, which is tangent to the periodic orbit and a solution of the linear variational equation

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u(t, u_0))v \tag{4.21}$$

satisfies

$$\|(v^{1} \wedge v)(t)\| \le K\|(v^{1} \wedge v)(s)\|e^{-\gamma(t-s)}, \quad 0 \le s \le t,$$
(4.22)

for some constants $K, \gamma > 0$. The exponential decay of $v^1 \wedge v$ is deduced from the uniform asymptotic stability of the second compound differential equation (4.20). This means that, infinitesimally, the area of a parallelogram with one side tangent to the periodic orbit decays exponentially in the dynamics.

The remainder of this section will be devoted to discussion of an application of Theorem 4.12. Consider a reaction diffusion system

$$\frac{\partial u}{\partial t} = D\Delta u + f(x, u), \quad x \in \Omega,$$
(4.23)

with Dirichlet boundary condition

$$u|_{\partial\Omega} = 0, \tag{4.24}$$

or Neumann boundary condition

$$\left. \frac{\partial u}{\partial n} \right|_{\partial \Omega} = 0,$$
 (4.25)

or a more general boundary condition of the form

$$Bu|_{\partial\Omega} = 0, \tag{4.26}$$

where $\Omega \subset \mathbb{R}^m$ is a bounded domain with smooth boundary, $f : \Omega \times \mathbb{R}^n \to \mathbb{R}^n$ is locally Lipschitz continuous in u, uniformly in x and $D = \text{diag}(d_1, \dots, d_n)$, $d_i \geq 0$. Let $X = L^2(\Omega, \mathbb{R}^n)$ and $A : \mathcal{D}(A) \subset X \to X$ be the linear unbounded operator $Au = -\Delta u$, where

$$\mathcal{D}(A) = \left\{ u \in H^2(\Omega, \mathbb{R}^n), \ Bu|_{\partial\Omega} = 0 \right\}.$$

The initial value problem of (4.23) is well-posed in X^{α} . Two representations of the second compound differential equation of (4.23) are (3.15) and (3.17) in Section 3.2 where the coefficient $a_{jk}(t,x) = \frac{\partial f_j}{\partial u_k}(x,u(t,x))$. Theorem 4.12 gives a sufficient condition for the ω -limit set of a bounded solution $u(t, u_0)$ to be a periodic orbit in X^{α} .

Example 4.1. Consider a special case of (4.23) with n = 2 and $\Omega = (0, L)$:

$$u_t = d_1 u_{xx} + f(x, u, v), \quad v_t = d_2 v_{xx} + g(x, u, v), \qquad 0 < x < L, \quad t > 0,$$

$$u(t, 0) = u(t, L) = 0, \quad v(t, 0) = v(t, L) = 0, \quad t \ge 0,$$

(4.27)

where $f, g \in C^2((0,L) \times \mathbb{R}^2 \to \mathbb{R}^2)$. Suppose that (u(t,x), v(t,x))is a bounded solution of (4.27) in $X = H_0^1(0,L) \times H_0^1(0,L)$. Then $\left(\frac{\partial u}{\partial t}(t,x), \frac{\partial v}{\partial t}(t,x)\right) \in X$ (see Henry [42], Theorem 3.5.2, page 71). A pointwise representation of the second compound differential equation of the linear variation equation at any solution (u(t,x), v(t,x)) is

$$\frac{\partial w_{11}}{\partial t} = d_1(\Delta_1 w_{11} + \Delta_2 w_{11}) + (a_{11}(t, x_1) + a_{11}(t, x_2))w_{11}
+ a_{12}(t, x_1)w_{21} + a_{12}(t, x_2)w_{12},
\frac{\partial w_{22}}{\partial t} = d_2(\Delta_1 w_{22} + \Delta_2 w_{22}) + (a_{22}(t, x_1) + a_{22}(t, x_2))w_{22}
+ a_{21}(t, x_1)w_{12} + a_{21}(t, x_2)w_{21},
\frac{\partial w_{12}}{\partial t} = d_1\Delta_1 w_{12} + d_2\Delta_2 w_{12} + (a_{11}(t, x_1) + a_{22}(t, x_2))w_{12}
+ a_{12}(t, x_1)w_{22} + a_{21}(t, x_2)w_{11},$$
(4.28)

where

$$\begin{bmatrix} a_{11}(t,x) & a_{12}(t,x) \\ a_{21}(t,x) & a_{22}(t,x) \end{bmatrix} := \begin{bmatrix} \frac{\partial f}{\partial u}(x,u(t,x),v(t,x)) & \frac{\partial f}{\partial v}(x,u(t,x),v(t,x)) \\ \frac{\partial g}{\partial u}(x,u(t,x),v(t,x)) & \frac{\partial g}{\partial v}(x,u(t,x),v(t,x)) \end{bmatrix}.$$

Let $\lambda_1 \leq \lambda_2 \leq \cdots$ denote the eigenvalues of the Laplace equation

$$\Delta u = -\lambda_i u, \qquad \Delta v = -\lambda_i v$$

$$u(t,0) = u(t,L) = 0, \quad v(t,0) = v(t,L) = 0.$$

Consider a Lyapunov function

$$V(t) = \frac{1}{2!} \int_{(0,L)^2} \left[(w_{11})^2 + (w_{22})^2 + 2(w_{12})^2 \right].$$

Then the derivative of V calculated for (3.28) of Section 3.2 implies that for any positive functions $\nu_1(x), \nu_2(x)$,

$$\frac{d^{+}V}{dt} \leq \int_{(0,L)^{2}} \left[\mu_{1}(t)(w_{11})^{2} + \mu_{2}(t)(w_{22})^{2} + 2\mu_{3}(t)(w_{12})^{2} \right] \\
\leq 2\mu(t)V,$$
(4.29)

where

$$\begin{split} \mu_1(t) &= -(\lambda_1 + \lambda_2) \, d_1 + \max_x \left\{ 2a_{11}(t, x) + \frac{1}{\nu_1(x)} \left| a_{12}(t, x) + a_{21}(t, x) \right| \right\}, \\ \mu_2(t) &= -(\lambda_1 + \lambda_2) \, d_2 + \max_x \left\{ 2a_{22}(t, x) + \frac{1}{\nu_2(x)} \left| a_{12}(t, x) + a_{21}(t, x) \right| \right\}, \\ \mu_3(t) &= -\lambda_1(d_1 + d_2) + \max_x \left\{ a_{11}(t, x) + \frac{\nu_2(x)}{2} \left| a_{12}(t, x) + a_{21}(t, x) \right| \right\}, \\ &+ \max_x \left\{ a_{22}(t, x) + \frac{\nu_1(x)}{2} \left| a_{12}(t, x) + a_{21}(t, x) \right| \right\}, \\ \mu(t) &= \max\{\mu_1(t), \mu_2(t), \mu_3(t)\}, \end{split}$$

and the maximum is taken over 0 < x < L.

Since $H^1(0, L) \subset L^2(0, L)$ is compact, Theorem 4.12 implies the following corollary.

Corollary 4.13. Suppose that n = 2 and $u(t, u_0)$ is a bounded solution of (4.23) in X and that there exist constants a, b > 0 such that

$$\int_{s}^{t} \mu < -a(t-s) + b, \quad 0 \le s \le t < \infty,$$
(4.31)

where $\mu(\cdot)$ is defined by (4.30). If the omega limit set $\omega(u_0)$ of $u(t, u_0)$ contains no steady state solution, then it is a stable hyperbolic periodic solution in $L^2(0, L)$.

(4.30)

Remark 4.5. The above procedure also applies to other boundary conditions where λ_i are the eigenvalues of the Laplace equation with correspond boundary conditions.

Remark 4.6. To apply Theorem 4.12, it is not necessary to find an explicit bounded solution $u(t, u_0)$. For example, if S is an invariant bounded set and no steady state solutions exist in S, then there exists a stable hyperbolic periodic solution in S as long as inequality (4.31) is true for an upper bound of $\mu(t)$ in S.

Fiedler and Mallet-Paret [30] proved the classical Poincaré-Bendixson theorem for a scalar reaction diffusion equation

$$u_t = u_{xx} + f(x, u, u_x), \quad 0 < x < 2\pi, \quad t > 0, u(t, 0) = u(t, 2\pi), \quad u_x(t, 0) = u_x(t, 2\pi), \quad t \ge 0,$$
(4.32)

where $f \in C^2((0, 2\pi) \times \mathbb{R} \times \mathbb{R} \to \mathbb{R})$. However, Theorem 4.12 establishes both the existence and stability of periodic solutions. Let $X = \{u \in$ $H^2(0, 2\pi) : u(0) = u(2\pi), u_x(0) = u_x(2\pi)\}$. The initial value problem associated to (4.32) generates a local semiflow on X. The linear variational equation of (4.32) at a solution u = u(t, x) is

$$v_t = v_{xx} + a(t, x)v_x + b(t, x)v_x, \quad 0 < x < 2\pi, \quad t > 0, v(t, 0) = v(t, 2\pi), \quad v_x(t, 0) = v_x(t, 2\pi), \quad t \ge 0,$$
(4.33)

where

$$a(t,x) = \frac{\partial f}{\partial u_x}(x,u,u_x) = f_{u_x}(x,u,u_x), \quad b(t,x) = \frac{\partial f}{\partial u}(x,u,u_x) = f_u(x,u,u_x).$$

A similar argument to that in Corollary 3.33 of Section 3.4 implies Corollary 4.14.

Corollary 4.14. Suppose that u(t, x) is a bounded solution in X and that there exist a, b > 0 such that $\max_{x} |a(t, x)| > 0$, $0 \le x \le 2\pi$ and

$$\int_{s}^{t} \max_{0 \le x \le 2\pi} \left\{ 4b(t,x) + 2|a(t,x)|\nu(t) \right\} dt < (1-a)(t-s) + b;$$

r) = 0 for $0 \le x \le 2\pi$ and

or $a(t, x) \equiv 0$ for $0 \leq x \leq 2\pi$ and

$$2\int_s^t \max_x b(t,x) < (1-a)(t-s) + b,$$

where $\nu(t) = \max_{x} |a(t,x)|$. If the omega limit set $\omega(u_0)$ of u(t,x) contains no steady state solution, then it is a stable hyperbolic periodic solution in $L^2(0,L)$.

4.2.1 Proof of Theorem 4.12:

The proof of Theorem 4.12 is adapted from the argument used in the proof of Theorem 4.1 in Li and Muldowney [55]. To prove Theorem 4.12, the following Proposition 4.15 is established to show that, under the assumptions of the theorem, the solution space of the linear variational equation (4.21) is the direct sum of two subspaces X_1 and X_2 ; the 1-dimensional subspace X_1 is strongly stable and spanned by $v^1(t) = \frac{du}{dt}(t, u_0)$ and the codimension 1 subspace X_2 is uniformly asymptotically stable.

Let X be a normed space and $t \in [0, \infty)$ and X^* be its (continuous) dual space. Let V be a linear space of maps $t \mapsto v(t)$ from $[0, \infty)$ to X. If $x \in X$ and $\psi \in X^*$, define

$$\psi(x) = \langle \psi, x \rangle \,.$$

For $x^1, x^2 \in X$, define a norm of $x^1 \wedge x^2$ as follows (see Section 1.1):

$$\|x^{1} \wedge x^{2}\|_{\bigwedge^{2} X} = \sup_{\psi_{i}} \left| \det \left[\begin{array}{cc} \langle \psi_{1}, x^{1} \rangle & \langle \psi_{1}, x^{2} \rangle \\ \langle \psi_{2}, x^{1} \rangle & \langle \psi_{2}, x^{2} \rangle \end{array} \right] \right|.$$
(4.34)

Here the supreme is taken over $\psi_i \in X'$, $\|\psi_i\|_{X'} \leq 1$, i = 1, 2. For simplicity, the symbol $\|\cdot\|$ will be used instead of $\|\cdot\|_{\bigwedge^k X}$ except when the relationship with the norm in X is to be emphasized.

Proposition 4.15. Suppose that assumptions (i), (ii), (iii) are satisfied:

(i) there exists a constant L > 1 and a nonzero $v^1 \in V$ such that

$$||v^{1}(t)|| \leq L||v^{1}(s)||, \quad 0 \leq s, t;$$

(ii) there exist constants $K, \gamma > 0$ such that for every $v \in V$,

$$||(v^1 \wedge v)(t)|| \le K ||(v^1 \wedge v)(s)||e^{-\gamma(t-s)}, \quad 0 \le s \le t;$$

(iii) there exist constants H > 0 and $\beta \ge 0$ such that for every $v \in V$,

$$||v(t)|| \le H ||v(s)|| e^{\beta(t-s)}, \quad 0 \le s \le t.$$

Then there exists a constant C > 0 such that

$$V = V_1 \oplus V_2$$

where

$$V_1 = span\{v^1\}, \quad V_2 = \{y \in V : \|v(t)\| \le C \|y(s)\| e^{-\gamma(t-s)}, \ 0 \le s \le t\}$$

Remark 4.7. When $X = \mathbb{R}^n$, Proposition 4.15 is proved by Li and Muldowney [55].

Proposition 4.15 will be proved by the following sequence of lemmas.

Lemma 4.16. For $x^1, x^2 \in X$,

$$\eta \|x^1\| \|x^2\| \le \|x^1 \wedge x^2\| \le 2\|x^1\| \|x^2\|,$$

where

$$\eta = \min_{-1 \le \nu \le 1} \left\{ \left\| \frac{x^1}{\|x^1\|} - \nu \frac{x^2}{\|x^2\|} \right\| \right\}.$$

Proof. For $x^i \in X$ and $\psi_i \in X'$, $\|\psi_i\| \le 1$, i = 1, 2,

$$\begin{aligned} \left| \det \begin{bmatrix} \langle \psi_1, x^1 \rangle & \langle \psi_1, x^2 \rangle \\ \langle \psi_2, x^1 \rangle & \langle \psi_2, x^2 \rangle \end{bmatrix} \right| \\ &\leq |\langle \psi_1, x^1 \rangle \cdot \langle \psi_2, x^2 \rangle| + |\langle \psi_1, x^2 \rangle \cdot \langle \psi_2, x^1 \rangle| \\ &\leq 2 \|x^1\| \|x^2\|. \end{aligned}$$

Thus

$$\|x^1 \wedge x^2\| \le 2\|x^1\| \|x^2\|$$

Let $(x^1)^* \in X'$ such that $||x|| = \langle x^*, x \rangle$, $||x^*||_{X'} = 1$. Then

$$\begin{split} \|x^{1} \wedge x^{2}\| &\geq \sup_{\psi} | \ \|x^{1}\| \cdot \langle \psi, x^{2} \rangle - \langle (x^{1})^{*}, x^{2} \rangle \cdot \langle \psi, x^{1} \rangle | \\ &\geq \|x^{1}\| \|x^{2}\| \sup_{\psi} \left| \frac{\langle \psi, x^{2} \rangle}{\|x^{2}\|} - \frac{\langle \psi, x^{1} \rangle}{\|x^{1}\|} \cdot \frac{\langle (x^{1})^{*}, x^{2} \rangle}{\|x^{2}\|} \right| \\ &= \|x^{1}\| \|x^{2}\| \sup_{\psi} \left| \left\langle \psi, \frac{x^{2}}{\|x^{2}\|} - \nu \frac{x^{1}}{\|x^{1}\|} \right\rangle \right| \\ &= \|x^{1}\| \|x^{2}\| \left\| \frac{x^{2}}{\|x^{2}\|} - \nu \frac{x^{1}}{\|x^{1}\|} \right\|, \end{split}$$

where
$$\nu = \frac{\langle (x^1)^*, x^2 \rangle}{\|x^2\|}$$
 and $-1 \le \nu \le 1$. Therefore,
 $\|x^1 \wedge x^2\| \ge \eta \|x^1\| \|x^2\|$,

where

$$\eta = \min_{-1 \le \nu \le 1} \left\{ \left\| \frac{x^1}{\|x^1\|} - \nu \frac{x^2}{\|x^2\|} \right\| \right\}.$$

Remark 4.8. The number η in Lemma 4.16 is a measure of the angle between x^1 and x^2 , see Figure 4.1. When X is a Hilbert space with inner product $\langle \cdot, \cdot \rangle$,

$$\|x^1 \wedge x^2\|^2 = \det \left| \begin{array}{c} \langle x^1, x^1 \rangle & \langle x^1, x^2 \rangle \\ \langle x^2, x^1 \rangle & \langle x^2, x^2 \rangle \end{array} \right| = \sin^2 \theta \|x^1\|^2 \|x^2\|^2,$$

where θ is the angle of the vector x^1, x^2 and

$$\cos heta=rac{\langle x^1,x^2
angle}{\|x^1\|\|x^2\|}.$$

Thus

$$\|x^1 \wedge x^2\|^2 = \eta^2 \|x^1\|^2 \|x^2\|^2,$$

where $\eta = \left\| \frac{x^1}{\|x^1\|} - \nu \frac{x^2}{\|x^2\|} \right\|$ with $\nu = \frac{\langle x^1, x^2 \rangle}{\|x^1\|}$.



Figure 4.1: A measure of the angle between x^1 and x^2

Lemma 4.17 is from Daleckiĭ and Kreĭn [23], page 156.

Lemma 4.17. Suppose that the normed space X decomposes into a direct sum $X = X_1 \oplus X_2$ of closed subspaces and $P_1, P_2 = I - P_1$ are the corresponding projections. Then

$$\inf_{x_i \in X_i} \left\{ \left\| \frac{x^1}{\|x^1\|} - \frac{x^2}{\|x^2\|} \right\| \right\} \le \frac{2}{\|P_i\|}, \quad i = 1, 2.$$

Proof. Suppose that $x^i \in X_i$, i = 1, 2. Let $x = x^1 + x^2$ where $x^i = P_i x$, i = 1, 2. Then

$$\begin{aligned} \left\| \frac{x^{1}}{\|x^{1}\|} - \frac{x^{2}}{\|x^{2}\|} \right\| &= \left\| \frac{P_{1}x}{\|P_{1}x\|} - \frac{P_{2}x}{\|P_{2}x\|} \right\| \\ &= \frac{1}{\|P_{1}x\|} \left\| P_{1}x - \frac{\|P_{1}x\|}{\|P_{2}x\|} P_{2}x \right\| \\ &= \frac{1}{\|P_{1}x\|} \left\| x - \frac{\|P_{1}x\| - \|P_{2}x\|}{\|P_{2}x\|} P_{2}x \right\| \\ &\leq \frac{1}{\|P_{1}x\|} \left(\|x\| + \frac{\|P_{1}x + P_{2}x\|}{\|P_{2}x\|} \|P_{2}x\| \right) \\ &= \frac{2\|x\|}{\|P_{1}x\|}, \end{aligned}$$

and thus

$$\inf_{x_i \in X_i} \left\{ \left\| \frac{x^1}{\|x^1\|} - \frac{x^2}{\|x^2\|} \right\| \right\} \le 2 \inf_{x \in X} \frac{\|x\|}{\|P_1x\|} = \frac{2}{\|P_1\|}$$

$$\inf_{x_i \in X_i} \left\{ \left\| \frac{x^1}{\|x^1\|} - \frac{x^2}{\|x^2\|} \right\| \right\} \le \frac{2}{\|P_2\|}.$$

Similarly,

Lemma 4.18. Suppose that for every $v \in V$, the following two conditions are satisfied:

- (i) $\limsup_{t\to\infty} \|v(t)\| < \infty;$
- (ii) $\liminf_{t\to\infty} \|v(t)\| = 0 \text{ implies } \lim_{t\to\infty} \|v(t)\| = 0.$

Then

$$codim\left\{v \in V : \lim_{t \to \infty} \|v(t)\| = 0\right\} < 2$$

if and only if, for all $v^1, v^2 \in V$,

$$\lim_{t \to \infty} \|(v^1 \wedge v^2)(t)\| = 0.$$

The proof of Proposition 4.15 is adapted from Li and Muldowney [55].

Proof of Proposition 4.15:

Proof. The proof of Proposition 4.15 will be divided into two parts according to the value of β in assumption (*iii*).

Case 1: $\beta = 0$ in assumption (*iii*)

Let $V_0 = \{v \in V : \lim_{t \to \infty} ||v(t)|| = 0\}$. For any $v \in V_0$ and $t \ge 0$, there exists a *T* such that $||v(t+T)|| \le \frac{1}{2L} ||v(t)||$. It follows from assumption (*i*) that for any $-1 \le \nu \le 1$,

$$\begin{split} \frac{1}{2L} &= \frac{1}{L} - \frac{1}{2L} \le \left\| \frac{v^1(t+T)}{\|v^1(t)\|} - \nu \frac{v(t+T)}{\|v(t)\|} \right\| \le H \left\| \frac{v^1(t)}{\|v^1(t)\|} - \nu \frac{v(t)}{\|v(t)\|} \right\|,\\ \text{or} & \\ \frac{1}{2HL} \le \left\| \frac{v^1(t)}{\|v^1(t)\|} - \nu \frac{v(t)}{\|v(t)\|} \right\|. \end{split}$$

Thus, Lemma 4.16 implies that

$$\begin{aligned} \frac{1}{2HL} \|v^{1}(t)\| \|v(t)\| &\leq \|(v^{1} \wedge v)(t)\| \leq K \|(v^{1} \wedge v)(s)\| e^{-\gamma(t-s)} \\ &\leq 2K \|v^{1}(s)\| \|v(s)\| e^{-\gamma(t-s)}. \end{aligned}$$

It follows from assumption (i) again that

$$\frac{1}{2HL} \|v(t)\| \le 2KL \|v(s)\| e^{-\gamma(t-s)},$$

or

$$||v(t)|| \le 4HKL^2 ||v(s)||e^{-\gamma(t-s)}.$$

Therefore,

$$V_0 = \{ v \in V : \|v(t)\| \le 4HKL^2 \|v(s)\| e^{-\gamma(t-s)}, \text{ for all } 0 \le s \le t \}.$$

Now under assumptions (*ii*) and (*iii*), Lemma 4.18 implies that for every $v \in V$, there exists a nontrivial $v_0 = c_1v^1 + c_2v$ such that

$$\lim_{t \to \infty} \|v_0(t)\| = 0.$$

From assumption (i), $c_2 \neq 0$ and thus $v = -\frac{c_1}{c_2}v^1 + \frac{1}{c_2}v_0$. Therefore $V = \operatorname{span}\{v^1\} \bigoplus V_0$.

Case 2: $\beta > 0$ in assumption (*iii*)

In the following, it will be shown that assumptions (i)-(iii) imply that there exists a constant H' > 0 such that

$$||v(t)|| \le H' ||v(s)||, \ 0 \le s \le t,$$

and thus Proposition 4.15 follows from Case 1.

For each $t \ge 0$, let $(v^1)^*(t) \in X'$ be such that $||v^1(t)|| = \langle (v^1)^*(t), v^1(t) \rangle$ and $||(v^1)^*(t)||_{X'} = 1$. It follows from assumption (i) that $||v^1(t)|| \ne 0$. For each $v \in V$, let

$$\mathcal{V} = \operatorname{span}\{v^1, v\}.$$

Given $0 < \sigma < \tau$, define

$$v^{2}(t) = -\frac{\langle (v^{1})^{*}(\tau), v(\tau) \rangle}{\|v^{1}(\tau)\|} v^{1}(t) + v(t).$$
(4.35)

Then $v_2 \in \mathcal{V}$ and $\langle (v^1)^*(\tau), v^2(\tau) \rangle = 0$. For each $t \ge 0$, let $(v^2)^*(t) \in X'$ be such that $||v^2(t)|| = \langle (v^2)^*(t), v^2(t) \rangle$ and $||(v^2)^*(t)||_{X'} = 1$. Then

$$\begin{aligned} \|v^{1}(\tau)\|\|v^{2}(\tau)\| &= \left| \det \begin{bmatrix} \langle (v^{1})^{*}, v^{1} \rangle & \langle (v^{1})^{*}, v^{2} \rangle \\ \langle (v^{2})^{*}, v^{1} \rangle & \langle (v^{2})^{*}, v^{2} \rangle \end{bmatrix} (\tau) \\ &\leq \|(v^{1} \wedge v^{2})(\tau)\| \\ &\leq K\|(v^{1} \wedge v^{2})(\sigma)\|e^{-\gamma(\tau-\sigma)} \\ &\leq 2K\|v^{1}(\sigma)\|\|v^{2}(\sigma)\|e^{-\gamma(\tau-\sigma)}. \end{aligned}$$

Thus assumption (i) implies that

$$\begin{aligned} \|v^{2}(\tau)\| &\leq 2K \frac{\|v^{1}(\sigma)\|}{\|v^{1}(\tau)\|} \|v^{2}(\sigma)\|e^{-\gamma(\tau-\sigma)} \\ &\leq 2KL \|v^{2}(\sigma)\|e^{-\gamma(\tau-\sigma)}. \end{aligned}$$

Therefore, every $v \in V$ can be written as

$$v(t) = \frac{\langle (v^1)^*(\tau), v(\tau) \rangle}{\|v^1(\tau)\|} v^1(t) + v^2(t), \qquad (4.36)$$

where $v^2 \in \mathcal{V}$ and $||v^2(\tau)|| \leq 2KL ||v^2(\sigma)||e^{-\gamma(\tau-\sigma)}$. Choose T > 0 such that

$$\delta := 2KLe^{-\gamma T} < \frac{1}{L} < 1. \tag{4.37}$$

Let $s_0 \ge 0$ and define $s_k = s_0 + kT$, $k = 0, 1, 2, \cdots$. Then every $v \in V$ can be written as

$$v(t) = c_k v^1(t) + v^{2,k}(t), \qquad k = 0, 1, 2, \cdots,$$
 (4.38)

where $v^{2,k} \in \mathcal{V}$ and

$$c_{k} = \frac{\langle (v^{1})^{*}(s_{k+1}), v(s_{k+1}) \rangle}{\|v^{1}(s_{k+1})\|}, \quad \langle (v^{1})^{*}(s_{k+1}), v^{2,k}(s_{k+1}) \rangle = 0,$$
$$\|v^{2,k}(s_{k+1})\| \le \delta \|v^{2,k}(s_{k})\|.$$

Let

$$\mathcal{V}_1 = \operatorname{span}\{v^1\}, \quad \mathcal{V}_{2,k} = \{v \in \mathcal{V} : \|v(s_{k+1})\| \le \delta \|v(s_k)\|\}.$$

Then (4.38) implies that on the interval $[s_k, s_{k+1}]$, $\mathcal{V} = \mathcal{V}_1 \oplus \mathcal{V}_{2,k}$ and $\mathcal{V}_1, \mathcal{V}_{2,k}$ are closed subspaces of \mathcal{V} . Notice that if $v^2 \in \mathcal{V}_{2,k}$, then

$$He^{\beta T} \left\| \frac{v^1(s_k)}{\|v^1(s_k)\|} - \frac{v^2(s_k)}{\|v^2(s_k)\|} \right\| \ge \left\| \frac{v^1(s_{k+1})}{\|v^1(s_k)\|} - \frac{v^2(s_{k+1})}{\|v^2(s_k)\|} \right|$$
$$\ge \frac{1}{L} - \delta > 0.$$

Thus, on the interval $[s_k, s_{k+1}]$, Lemma 4.17 implies that the projections $P_i(\cdot), i = 1, 2$ onto \mathcal{V}_1 and $\mathcal{V}_{2,k}$ respectively satisfy

$$P_i(s_k) \le \mu H e^{\beta T} := \rho,$$

where $\mu = \frac{2L}{1-L\delta}$. For every $v \in V$, it follows from (4.38) that

$$v(s_k) = c_k v^1(s_k) + v^{2,k}(s_k) = c_{k-1} v^1(s_k) + v^{2,k-1}(s_k).$$

Then

$$v^{2,k}(s_k) = P_2(s_k)v^{2,k-1}(s_k),$$

$$c_kv^1(s_k) = c_{k-1}v^1(s_k) + P_1(s_k)v^{2,k-1}(s_k).$$

Thus, for i = 1, 2,

$$\begin{aligned} \|P_{i}(s_{k})v^{2,k-1}(s_{k})\| &\leq \rho \|v^{2,k-1}(s_{k})\| \\ &\leq \rho \delta \|v^{2,k-1}(s_{k-1})\| := R \|v^{2,k-1}(s_{k-1})\| \\ &\leq \cdots \leq R^{k} \|v^{2,0}(s_{0})\| \\ &\leq \rho R^{k} \|v(s_{0})\|, \end{aligned}$$

since $||v^{2,0}(s_0)|| = ||P_2(s_0)v(s_0)|| \le \rho ||v(s_0)||$, where

$$R = \rho \delta = 2KL\mu H e^{(\beta - \gamma)T}.$$
(4.39)

Hence

$$\begin{aligned} |c_k| \|v^1(s_k)\| &\leq |c_{k-1}| \|v^1(s_k)\| + \|P_1(s_k)v^{2,k-1}(s_k)\| \\ &\leq |c_{k-1}| \|v^1(s_k)\| + \rho R^k \|v(s_0)\|, \end{aligned}$$

or

$$\begin{aligned} |c_{k}| &\leq |c_{k-1}| + \rho \|v(s_{0})\| \frac{R^{k}}{\|v^{1}(s_{k})\|}, \\ &\leq \cdots \leq |c_{0}| + \rho \|v(s_{0})\| \sum_{i=1}^{k} \frac{R^{i}}{\|v^{1}(s_{i})\|}, \\ &\leq \rho \frac{\|v(s_{0})\|}{\|v^{1}(s_{0})\|} + \rho \|v(s_{0})\| \sum_{i=1}^{k} \frac{R^{i}}{\|v^{1}(s_{i})\|}, \\ &\leq \rho \|v(s_{0})\| \sum_{i=0}^{k} \frac{R^{i}}{m}. \end{aligned}$$

$$(4.40)$$

since $|c_0| = \frac{\|P_1(s_0)v(s_0)\|}{\|v^1(s_0)\|} \le \rho \frac{\|v(s_0)\|}{\|v^1(s_0)\|}$, where $m := \frac{\|v^1(s_0)\|}{L} \le \|v^1(t)\|$ for all $t \ge s_0$.

Case 2.1: $\beta < \gamma$

It can be assumed that R < 1 in addition to (4.37) by choosing T sufficiently large if necessary. Now inequality (4.40) implies

$$|c_k| \le \frac{\rho}{m(1-R)} ||v(s_0)||,$$

and thus

$$\begin{aligned} \|v(s_k)\| &\leq \|c_k\| \|v^1(s_k)\| + \|v^{2,k}(s_k)\| \\ &\leq \frac{\rho}{m(1-R)} \|y(s_0)\| \|v^1(s_k)\| + \rho R^k \|v(s_0)\| \\ &\leq \rho \left(\frac{M}{m(1-R)} + 1\right) \|v(s_0)\|, \end{aligned}$$

where $M := L \|v^1(s_0\|) \ge \|v^1(t)\|$ for all $t \ge s_0$. It follows from assumption (*iii*) that for every $v \in V$,

$$||v(t)|| \le He^{\beta T} ||v(s_k)||$$
 for $s_k \le t < s_{k+1}$,

and thus

$$||v(t)|| \le H' ||v(s_0)||, \quad 0 \le s_0 \le t,$$

where $H' = He^{\beta T} \rho \left(\frac{M}{m(1-R)} + 1 \right)$ which is independent of s_0 . Case 2.2: $\beta \geq \gamma$

The case $\beta = \gamma$ can be included in the case $\beta > \gamma$ by replacing γ by a slightly smaller constant in assumption (*ii*) or β by a slightly larger constant in assumption (*iii*). Thus, without loss of generality, only the case $\beta > \gamma$ will be discussed.

Step 1: Inequality (4.40) implies

$$|c_k| \le \rho \frac{R^k - 1}{m(R-1)} ||v(s_0)||,$$

and thus

$$\begin{aligned} \|v(s_k)\| &\leq \|c_k\| \|v^1(s_k)\| + \|v^{2,k}(s_k)\| \\ &\leq \rho \frac{R^k - 1}{m(R-1)} \|v(s_0)\| \|v^1(s_k)\| + \rho R^k \|v(s_0)\| \\ &\leq \rho \left(\frac{M}{m(R-1)} + 1\right) R^k \|v(s_0)\|. \end{aligned}$$

Let $0 < \gamma_1 < \gamma$. Choose $T > \frac{1}{\gamma_1} \ln(4KL^2H + 2KL^2)$ so that

 $2KL\mu He^{-\gamma_1 T} < 1.$

Notice that if $\beta_1 = \beta + \gamma_1 - \gamma < \beta$, then

$$R = 2KL\mu H e^{-\gamma_1 T} e^{\beta_1 T} \le e^{\beta_1 T},$$

and thus

$$\|v(s_k)\| \leq \rho\left(\frac{M}{m(R-1)} + 1\right) e^{k\beta_1 T} \|v(s_0)\|$$

$$\leq \rho\left(\frac{M}{m(R-1)} + 1\right) e^{\beta_1(s_k - s_0)} \|v(s_0)\|.$$

Hence, for $s_k \leq t < s_{k+1}$,

$$||v(t)|| \le He^{\beta T} ||v(s_k)|| \le H_1 ||v(s_0)||e^{\beta_1(t-s_0)}, \quad 0 \le s_0 \le t,$$

where $H_1 = He^{\beta T} \rho \left(\frac{M}{m(R-1)} + 1 \right)$ which is independent of s_0 . Step 2: Repeat Step 1 until $\beta_n < \gamma$ for some n and

$$||v(t)|| \le H_n ||v(s_0)|| e^{\beta_n (t-s_0)}, \quad 0 \le s_0 \le t,$$

where H_n is independent of s_0 . In fact, this can be done by choosing T sufficiently large in each step. Since $\beta_n < \gamma$, **Case 2.1** implies that there exists a constant H' > 0 such that

$$||v(t)|| \le H' ||v(s)||, \ 0 \le s \le t.$$

Therefore, Proposition 4.15 is proved.

Proof of Theorem 4.12:

Proof. Since $u(t, u_0)$ is bounded and $\frac{\partial f}{\partial u}(u(t, u_0)) \in \mathscr{L}(X^{\alpha}, X)$, there exists a constant C_1 such that $\|\frac{\partial f}{\partial u}(u(t, u_0))\|_{\mathscr{L}(X^{\alpha}, X)} \leq C_1$. For a sectorial operator A, there exists a constant M_1 such that for $0 \leq \beta_1 \leq \alpha < 1$,

$$||e^{-A(t-s)}y(s)||_{\alpha} \le M_1(t-s)^{\beta_1-\alpha}||y(s)||_{\beta_1}.$$

The solution of the linear variational equation of (4.19) at $u(t, u_0)$,

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u(t, u_0))v,$$

$$v(0) = v_0,$$
(4.41)

satisfies

$$\begin{aligned} \|v(t,v_0)\|_{\alpha} &\leq \|e^{-A(t-t_0)}v(0)\|_{\alpha} + \int_0^t \left\|e^{-A(t-s)}\frac{df}{du}(u(s,u_0))v(s,v_0)\right\|_{\alpha} ds \\ &\leq M_1 \|v(0)\|_{\alpha} + \int_0^t \left\|e^{-A(t-s)}\frac{\partial f}{\partial u}(u(s,u_0))v(s,v_0)\right\|_{\alpha} ds \\ &\leq M_1 \|v(0)\|_{\alpha} + M_1 \int_0^t (t-s)^{-\alpha} \left\|\frac{\partial f}{\partial u}(u(s,u_0))v(s,v_0)\right\| ds \\ &\leq M_1 \|v(0)\|_{\alpha} + M_1 C_1 \int_0^t (t-s)^{-\alpha} \|v(s,v_0)\|_{\alpha} ds. \end{aligned}$$

Thus, Lemma 4.10 implies that there exist constants $M, \beta > 0$ such that

$$\|v(t, v_0)\|_{\alpha} \le M e^{\beta t} \|v_0\|_{\alpha}.$$
(4.42)

Since $v^1(t) = \frac{du}{dt}(u(t, u_0)) \in X^{\alpha}$ (see Henry [42], Theorem 3.5.2, page 71) is uniformly bounded, there exists a constant a > 0 such that

$$\|v^1(t)\|_{\alpha} < a, \qquad t \ge 0.$$
 (4.43)

Moreover, since its omega limit set $\omega(u_0)$ contains no steady state solution, there exists a constant b > 0 such that

$$\|v^{1}(s)\|_{\alpha} \ge b > 0, \qquad s \ge 0.$$
(4.44)

If not, then there exists a sequence $\{s_n\}, s_n \to \infty$ as $n \to \infty$, such that

$$\lim_{n \to \infty} -Au(s_n, u_0) + f(u(s_n, u_0)) = \lim_{n \to \infty} v^1(s_n) = 0.$$

Since f is continuous and A is closed on X^{α} when $0 < \alpha < 1$, this implies that $\omega(u_0)$ contains a steady state solution contradicting the assumption that $\omega(u_0)$ contains no such solution. Thus, (4.43) and (4.44) imply

$$\|v^{1}(t)\|_{\alpha} \le L \|v^{1}(s)\|_{\alpha}, \quad \text{for } t, s \ge 0,$$
(4.45)

where $L = \frac{a}{b} > 1$. Let $\{T(t,s), 0 \le s \le t\}$ be the evolution operator generated by the solution of (4.41). The space X decomposes into a direct sum $X = X_1(t) \oplus X_2(t)$ where $X_1(t) = \operatorname{span}\{\frac{du}{dt}(t, u_0)\} \subset X^{\alpha}$ is the tangent vector space at time t of the bounded solution $u(t, u_0)$. If $v(t) = v(t, v_0)$ is a



Figure 4.2: Evolution of the oriented infinitesimal parallelogram $v^1 \wedge v$

solution of the linear variational equation (4.41), then $(v^1 \wedge v)(t)$ is a solution of the second compound differential equation (4.20); see Figure 4.2. Thus the uniform asymptotic stability of (4.20) implies that there exist constant $K, \gamma > 0$ such that

$$\|(v^{1} \wedge v)(t)\| \le K\|(v^{1} \wedge v)(s)\|e^{-\gamma(t-s)}, \quad 0 \le s \le t.$$
(4.46)

From (4.42), (4.45) and (4.46), Proposition 4.15 implies that there exists a constant C > 0 such that

$$||T(t,s)x_1||_{\alpha} \le C||x_1||_{\alpha} \quad \text{for } x_1 \in X_1(s), \quad 0 \le s, t, ||T(t,s)x_2||_{\alpha} \le Ce^{-\gamma(t-s)}||x_2||_{\alpha} \quad \text{for } x_2 \in X_2^{\alpha}(s), \quad 0 \le s \le t.$$
(4.47)

Since the stability of the second compound differential equation (4.20) is sufficiently robust, the above argument can be applied to any solution in the omega limit set $\omega(u_0)$.

In the following, consider $u = u(t, u_1)$ where $u_1 \in \omega(u_0)$ and so $u(t, u_1) \in \omega(u_0)$. Let $X = X_1(t) \oplus X_2(t)$ where $X_1(t) = \operatorname{span}\{\frac{du}{dt}(t, u_1)\} \subset X^{\alpha}$ and $P_i(s)$ be the projections onto $X_i(s), i = 1, 2$. Next apply an argument used in Coppel [19], page 82-85, Henry [42], page 251-253 and Li and Muldowney [55]. For any solution u(t) of (4.19), let $z = u - u(t, u_1)$. Then (4.19) is equivalent to

$$\frac{dz}{dt} + \left(A - \frac{\partial f}{\partial u}(u(t, u_1))\right)z = g(t, z), \qquad (4.48)$$

where

$$g(t,z) = f(u(t,u_1) + z) - f(u(t,u_1)) - \frac{\partial f}{\partial u}(u(t,u_1))z$$

satisfies g(t, 0) = 0 and

$$||g(t, z_1) - g(t, z_2)|| \le k(\delta) ||z_1 - z_2||_{\alpha}, \text{ if } ||z_1||_{\alpha}, ||z_2||_{\alpha} \le \delta,$$

with $k(\delta) \to 0$ as $\delta \to 0^+$. Given $\varepsilon > 0$, there exists $\delta > 0$ such that $||z_1||_{\alpha}, ||z_2||_{\alpha} \leq \delta$ implies

$$\|g(t, z_1) - g(t, z_2)\| \le \varepsilon \|z_1 - z_2\|_{\alpha}.$$
(4.49)

If $0 < \beta < \gamma$, consider a Banach space

$$\mathcal{B}_{\alpha,\beta} = \{ z \in C([0,\infty), X^{\alpha}) : \|z\|_{\alpha,\beta} < \infty \},\$$

where $||z||_{\alpha,\beta} = \sup_{t\geq 0} ||z(t)||_{\alpha} e^{\beta t}$. If $z \in \mathcal{B}_{\alpha,\beta}$, $||z||_{\alpha,\beta} \leq \delta$ and $a \in X_2^{\alpha}(0)$, define $F_a(z)$ as

$$F_{a}(z)(t) = T(t,0)a + \int_{0}^{t} T(t,s)P_{2}(s)g(s,z(s))ds - \int_{t}^{\infty} T(t,s)P_{1}(s)g(s,z(s))ds, \quad t \ge 0.$$
(4.50)

From (4.47), (4.49) and (4.50),

$$\begin{aligned} \|F_{a}(z)(t)\|_{\alpha} &\leq C\left[e^{-\gamma t}\|a\|_{\alpha} + \varepsilon \int_{0}^{t} e^{-\gamma(t-s)}\|z(s)\|_{\alpha}ds + \varepsilon \int_{t}^{\infty} \|z(s)\|_{\alpha}ds\right] \\ &\leq C\left[e^{-\gamma t}\|a\|_{\alpha} + \varepsilon \|z\|_{\alpha,\beta} \left(\int_{0}^{t} e^{-\gamma(t-s)}e^{-\beta s}ds + \int_{t}^{\infty} e^{-\beta s}ds\right)\right] \\ &\leq C\left[e^{-\gamma t}\|a\|_{\alpha} + \varepsilon \|z\|_{\alpha,\beta} \left(\frac{e^{-\beta t}}{\gamma - \beta} + \frac{e^{-\beta t}}{\beta}\right)\right] \\ &= Ce^{-\gamma t}\|a\|_{\alpha} + \theta \|z\|_{\alpha,\beta}e^{-\beta t}, \qquad \theta = C\varepsilon \frac{\gamma}{\beta(\gamma - \beta)}. \end{aligned}$$

$$(4.51)$$

Choose ε and δ so that $0 < \theta < 1$ and $C ||a||_{\alpha} \le (1 - \theta)\delta$ and thus

$$\begin{aligned} \|F_a(z)(t)\|_{\alpha,\beta} &\leq C \|a\|_{\alpha} + \theta \|z\|_{\alpha,\beta} \\ &\leq (1-\theta)\delta + \theta\delta = \delta. \end{aligned}$$

$$(4.52)$$

Let $z, z_i \in \mathcal{B}_{\alpha,\beta}, \|z\|_{\alpha,\beta} \leq \delta, \|z_i\|_{\alpha,\beta} \leq \delta, i = 1, 2$ and $a \in X_2^{\alpha}(0), \|a\|_{\alpha} \leq \frac{(1-\theta)}{C}\delta$. Then $F_a(z) \in \mathcal{B}_{\alpha,\beta}$ and $\|F_a(z)\|_{\alpha,\beta} \leq \delta$. A similar estimate shows that

$$||F_a(z_1) - F_a(z_2)||_{\alpha,\beta} \le \theta ||z_1 - z_2||_{\alpha,\beta}.$$

Hence the equation $z = F_{\xi}(z)$ has a unique solution $z^* = z^*(t, a)$. Notice that

$$(t,a) \mapsto u(t,\xi) := u(t,u_1) + z^*(t,a)$$

is a solution of (4.19) with initial condition

$$\xi = u(0, u_1) + z^*(0, a) = u_1 + a - \int_0^\infty T(0, s) P_1(s) g(s, z(s)) ds.$$
(4.53)

From the first inequality of (4.52),

$$\|z^*(t,a)\|_{\alpha,\beta} \le \frac{C}{1-\theta} \|a\|_{\alpha}.$$

Since $||g(t,z)||_{\alpha} = o(||z||_{\alpha})$ uniformly in t for $||z||_{\alpha} \to 0$, it follows that

$$z^*(0,a) = a + o(||a||_{\alpha})\dot{u}(0,u_1).$$
(4.54)

If ρ is sufficiently small, the set of all ξ satisfying (4.53) with $\|\xi\|_{\alpha} < \rho$ and $P_1(0)\xi = 0$ is a manifold S_{ρ} of codimension 1 and if $\xi \in S_{\rho}$,

$$\lim_{t \to \infty} \|u(t,\xi) - u(t,u_1)\|_{\alpha} = 0.$$
(4.55)

Let $u(t,\xi)$ be a solution of (4.19) with $u(0,\xi) = \xi$. For $\xi = u_1$, equation

$$u(t,\xi) - u_1 - z^*(0,a) = 0 \tag{4.56}$$

has a solution when t = 0, a = 0. Observe that the linear map $(t, a) \mapsto t \frac{du}{dt}(0, u_1) - a$ is invertible. From (4.54), the implicit function theorem implies that equation (4.56) is solvable when $\|\xi - u_1\|_{\alpha} < \sigma$ for some constant $\sigma > 0$, that is, there exist $t' = t'(\xi)$ and $a' \in X_2^{\alpha}(0)$ such that

$$u(t',\xi) = u_1 + z^*(0,a'),$$

where

$$\|a'\|_{\alpha} \leq \frac{1-\theta}{C}\delta.$$

Since $u_1 \in \omega(u_0)$, there exists $\xi_1, \xi_2 \in S_\rho$ such that $u(t_1, \xi_1) = \xi_2, t_1 > 0$ and a sequence $t_k \to \infty, u(t_k, \xi_1) \to u_1, k \to \infty$. Now (4.55) implies

$$\lim_{t \to \infty} \|u(t, \xi_1) - u(t, \xi_2)\|_{\alpha} = 0$$

so that

$$\lim_{t \to \infty} \|u(t,\xi_1) - u(t_1, u(t,\xi_1))\|_{\alpha} = 0$$
(4.57)

since $u(t,\xi_2) = u(t,u(t_1,\xi_1)) = u(t+t_1,\xi_1) = u(t_1,u(t,\xi_1))$. Let $t = t_k$ in (4.57). Then

$$\lim_{k \to \infty} \|u(t_k, \xi_1) - u(t_1, u(t_k, \xi_1))\|_{\alpha} = 0$$

and thus $u_1 = u(t_1, u_1)$, which shows that $u(t, u_1)$ is periodic with period t_1 . From Theorem 3.7 in Section 3.1, this orbit is stable hyperbolic, and thus attracts all nearby orbits. Therefore this orbit is the whole set $\omega(u_0)$.

Remark 4.9. If the bounded solution $u = u(t, u_0)$ is a periodic solution, then its omega limit set is itself. Thus Theorem 3.7 in Section 3.1 is a special case of Theorem 4.12.

Chapter 5 Bendixson Criterion

In this chapter a generalization of the Bendixson criterion for the nonexistence of periodic orbits to differential equations in Banach spaces is established.

In the early 1900s, I. Bendixson and H. Dulac gave conditions for a 2dimensional autonomous system of ordinary differential equations to rule out the existence of nontrivial periodic solutions, which are called the Bendixson and Dulac criteria. In particular, Bendixson [6] showed that

$$\frac{du}{dt} = f(u), \ u \in \mathbb{R}^n \tag{5.1}$$

has no nonconstant periodic solution if n = 2 and

div
$$f \neq 0$$
 on \mathbb{R}^2 .

Dulac [26] generalized this to the statement that if div $(\alpha f) \neq 0$ on a simply connected open subset D of \mathbb{R}^2 , where α is a real-valued function on D, then there is no closed path of (5.1) which lies entirely in D.

Higher dimensional Bendixson criteria have been developed by Busenberg and van den Driessche [12] and by R. A. Smith [100, 102]. Busenberg and van den Driessche obtained conditions which preclude the possibility of certain oriented loops occurring in the dynamics of (5.1) and are not confined to finite dimensional spaces. In particular, an extension is obtained for functional differential equations. R. A. Smith [102] shows that, if the system (5.1) is dissipative and $\lambda_1(x) + \lambda_2(x) < 0$ or $\lambda_{n-1}(x) + \lambda_n(x) > 0$, then each bounded solution converges to an equilibrium. Here $\lambda_1(x) \geq \lambda_2(x) \geq \cdots \geq \lambda_n(x)$ are the eigenvalues of $\frac{1}{2} \left(\left(\frac{\partial f}{\partial x} \right)^* + \frac{\partial f}{\partial x} \right)$, $\frac{\partial f}{\partial x}$ is the Jacobian matrix of f and the asterisk denotes transposition. In particular, there are no simple closed curves that are invariant with respect to (5.1). Smith's proof contains an error; see Li and Muldowney [53], page 465. However the result is correct.

Based on compound differential equations, Muldowney [78] observed that Smith's conditions imply that, in the dynamics of (5.1) the usual Euclidean measure of 2-dimensional surface area decreases with increasing or decreasing time respectively. Also it was noted that, for any 2-dimensional surface which has a given simple closed curve as its boundary, there is a positive lower bound on the area of the surface that depends only on the boundary curve. These observations were used in [78] to develop two new approaches to higher dimensional Bendixson conditions based on various measures of surface area. Both approaches give the Bendixson-Dulac results when n = 2.

The first approach in [78] uses the fact that, if some measure of 2dimensional surface area decreases in the dynamics of (5.1), then no simply connected open region $D \subset \mathbb{R}^n$ can contain a periodic orbit or any invariant simple closed curve if such a curve is the boundary of a 2-dimensional surface of minimum area. Since the boundary is invariant and the surface area decreases strictly, the minimality of the area is contradicted.

The preceding approach has the advantage that it requires only local existence of solutions. However, it places a restriction on the region D to which it is applied: it must have a shape that permits the existence of a minimal surface for any simple closed curve in D. As explained in [78], the "minimal surface" can exist in a fairly abstract sense, such as the existence of a minimizing sequence of surfaces.

The second approach requires that solutions originating in D exist globally, but replaces the requirement that 2-dimensional surface area decreases with a condition that the area of any surface originating in D tends to zero as ttends to infinity under (5.1). If the boundary is an invariant simple closed curve in D, this would contradict the existence of a positive lower bound for the surface area.

Both of these approaches are facilitated through consideration of the

second compound differential equation

$$\frac{dw}{dt} = \frac{\partial f^{[2]}}{\partial u} (u(t, u_0))w$$
(5.2)

with respect to a solution $u(t, u_0), u_0 \in D$, which governs the evolution of 2-dimensional surface areas near $u(t, u_0)$. In particular, the second approach can be implemented in terms of Bendixson conditions related to stability requirement on (5.2); see Li and Muldowney [58], Muldowney [78].

This chapter begins with the establishment of a positive lower bound for a 2-dimensional surface area whose boundary is a given simple closed curve in a Banach space X; the bound depends only on the curve. Bendixson conditions for differential equations in Banach spaces are developed in terms of stability of associated compound linear differential equations.

5.1 Surfaces and Boundaries in Banach Spaces

In this section, a measure of 2-dimensional surface is introduced and the existence of a positive lower bound for 2-dimensional surface area whose boundary is a given simple closed curve in a Banach space X is obtained.

Throughout this section, the real Banach space X is assumed to satisfy the *Radon-Nikodym property* (see [5, 27, 88]) and its norm is *Gâteaux* differentiable (see [29, 46, 72]).

Definition 5.1. Suppose that G is a nonempty open subset of a real Banach space X and that $f: G \to Y$ is a map from G into a real Banach space Y. The map f is *Gâteaux differentiable* at $x \in G$ if for each $h \in X$, the limit

$$\lim_{t \to 0} \frac{f(x+th) - f(x)}{t}$$
(5.3)

exists in the norm topology of Y and defines a linear (in h) map which is continuous from X to Y. The map f is Frechét differentiable at $x \in G$ if the limit (5.3) is uniform for $h \in S_X = \{x \in X : ||x|| = 1\}$. The Gâteaux derivative of f at x is denoted by df(x).

Definition 5.2. A real Banach space X is said to satisfy the *Radon-Nikodym* property if every function of bounded variation from [0, 1] into X is Gâteaux differentiable almost everywhere.

Earlier results of Dunford and Pettis (see [27]) showed that such spaces include separable dual spaces and reflexive spaces. The Aronszajn Theorem (see [88]) shows that every Lipschitz continuous map $f: G \to Y$ is Gâteaux differentiable almost everywhere provided that X is separable and Y satisfies the Radon-Nikodym property.

Definition 5.3. A norm $\|\cdot\|$ on a Banach space X is *Gâteaux differentiable* if it is Gâteaux differentiable at every point of $X \setminus \{0\}$.

Lemma 5.1. Let X be a Banach space whose norm is Gâteaux differentiable at the point $x \neq 0$ and X^* be its (continuous) dual space. Then $\lim_{t\to 0} \frac{\|x + ty\| - \|x\|}{t} = G_x(y) \text{ defines an element } G_x \in X^* \text{ such that } \|G_x\| \leq 1.$ Furthermore, if $x \in X$ and $|x| \leq 1$, then $\|G_x\| = 1$.

Let $B = \{x \in \mathbb{R}^2 : |x| < 1\}$ be the Euclidean unit ball in \mathbb{R}^2 centered at the origin, whose closure and boundary are denoted by \overline{B} and ∂B . The boundary ∂B can also be associated with $S^1 = \mathbb{R}/2\pi\mathbb{Z}$. The following definitions are similar to those in Li [51] and Li and Muldowney [53, 58].

Definition 5.4.

(i) A map $\psi \in \text{Lip}(S^1 \to X)$ is a simple rectifiable closed curve if it is one to one.

(ii) A map $h \in \operatorname{Lip}(\overline{B} \to X)$ is a rectifiable 2-dimensional surface in X, and the restriction $h|_{\partial B}$ is the boundary of h. The map h is a simple rectifiable 2dimensional surface if $h|_{\partial B} : \partial B \to X$ is one to one. Henceforth, for brevity, the term "rectifiable" will be omitted.

Define

$$\sum(\psi, X) := \left\{ h \in \operatorname{Lip}(\overline{B} \to X), \ h|_{\partial B} \text{ is one to one and } h(\partial B) = \psi(S^1) \right\},$$
(5.4)

where $\psi \in \operatorname{Lip}(S^1 \to X)$ is a simple closed curve. If D is a simply connected open subset of X, then

$$\sum(\psi, D) := \left\{ h \in \operatorname{Lip}(\overline{B} \to D), \ h|_{\partial B} \text{ is one to one and } h(\partial B) = \psi(S^1) \right\},$$
(5.5)

is nonempty (see Li and Muldowney [58]).

Definition 5.5. If X satisfies the Radon-Nikodym property and its norm is Gâteaux differentiable, define a surface area measure σ_2 on $\operatorname{Lip}(\overline{B} \to X)$ by

$$\sigma_2(h) := \int_B \left\| d^{(2)}h \right\| = \int_B \left\| \frac{\partial h}{\partial r_1}(r) \wedge \frac{\partial h}{\partial r_2}(r) \right\| dr, \qquad h \in \operatorname{Lip}(\overline{B} \to X),$$
(5.6)

where d h denotes the Gâteaux derivative of h. Here $\|\cdot\|$ is any norm on $\bigwedge^2 X$ and the associated operator norm on $\mathscr{L}(\bigwedge^2 \mathbb{R}^2, \bigwedge^2 X)$.

The integral in (5.6) exists and is finite since h is Gâteaux differentiable almost everywhere in B and the integrand is bounded.

Example 5.1. Suppose that $X = \{(x_1, x_2, \dots), x_i \in \mathbb{R}, i = 1, 2, \dots\}$ is a sequence space. Then $h(r) = (h_1(r), h_1(r), \dots)$ and for each $h \in \operatorname{Lip}(\overline{B} \to X)$,

$$\int\limits_{B} \left(\sum_{1 \le i < j}^{\infty} \left| \frac{\partial(h_i, h_j)}{\partial(r_1, r_2)} \right|^p \right)^{\frac{1}{p}} dr_1 dr_2 \quad \text{and} \quad \int\limits_{B} \sup_{1 \le i < j} \left| \frac{\partial(h_i, h_j)}{\partial(r_1, r_2)} \right| dr_1 dr_2$$

are examples of measures $\sigma_2(h)$ of 2-dimensional surface area if the integrals exist. When $X = \mathbb{R}^n$ and p = 2, this is the usual Euclidean measure of the surface area.

Example 5.2. Let X be a space of functions $f : s \mapsto f_s$, where $s \in \Omega$, a measurable subset in \mathbb{R}^n , and $f_s \in \mathbb{R}$. A surface h in X is a Lipschitz function $r \in \overline{B} \mapsto h(r) \in X$ and may be expressed in the form $(r, s) \mapsto h_s(r), r \in \overline{B}, s \in \Omega$ and $h_s(r) \in \mathbb{R}$. Then

$$\int_{B} \left(\int_{\Omega^2} \left| \frac{\partial(h_{s_1}, h_{s_2})}{\partial(r_1, r_2)} \right|^p ds_1 ds_2 \right)^{\frac{1}{p}} dr_1 dr_2 \quad \text{and} \quad \int_{B} \sup_{s_1, s_2 \in \Omega} \left| \frac{\partial(h_{s_1}, h_{s_2})}{\partial(r_1, r_2)} \right| dr_1 dr_2$$

are measures $\sigma_2(h)$ of 2-dimensional surface area if the integrals exist.

While Example 5.1 is a special case of Example 5.2, it is useful to state the examples separately. For instance, the first example might be used to define surface area for a function space in terms of the Fourier coefficients. The second example uses the pointwise representation of the functions. When $X = \mathbb{R}^n$ and ψ is a simple closed curve in \mathbb{R}^n , the existence of a positive lower bound for $\sigma_2(h)$ where $h \in \sum(\psi, \mathbb{R}^n)$ has been established by Li and Muldowney [51, 53, 78]. In the following, the existence of a positive lower bound for $\sigma_2(h)$ for general 2-dimensional surfaces in X with a given boundary will be proved.



Figure 5.1: The boundary of the 2-dimensional surface h in X

In general outline, the procedure is as follows. At any point on the curve where there is a tangent, a whole segment of the curve lies inside a cone whose axis is the tangent line. Because ψ is one to one, there is a ball centered on the point of tangency which contains no points of the curve other than those inside the cone. A rotation about the tangent is a map from X to \mathbb{R}^2 which does not increase the area. The region inside the ball and outside the cone "traps" a section of the surface which is mapped onto a sector of a disc in the upper half-plane of \mathbb{R}^2 ; see Figure 5.1. The area of the sector is a lower bound for the area of any surface in X that has this boundary curve.

Suppose that $\psi \in \operatorname{Lip}(S^1 \to X)$ is a simple closed curve. Then ψ is Gâteaux differentiable almost everywhere since X satisfies the Radon-Nikodym property. Moreover, ψ is Frechét differentiable almost everywhere since Frechét differentiability is equivalent to Gâteaux differentiability in \mathbb{R} . Without loss of generality, assume that ψ is Frechét differentiable at $0 \in S^1$ and $\psi(0) = 0 \in X$. Let $X_1 = d\psi(0)\mathbb{R}$ and e be a unit vector in X_1 . If $x = ae \in X_1$ and $a \in \mathbb{R}$, let q(x) = a be the coordinate functional of $x_1 \in X_1$ referred to e, whose norm $||q||_{X_1} = 1$. By the Hahn-Banach theorem, the linear functional q admits an extension on the whole space X denoted also by q with the same norm. Define $P_1 x = q(x)e$ and $P_2 = I - P_1$. Then P_1 and P_2 are projections and

$$\|P_1\| = 1, \qquad \|P_2\| \le 2 \tag{5.7}$$

(see Fabian et al [29], page 139 or Li and Muldowney [53], page 459-460). Now $X = X_1 \oplus X_2$ where $X_2 = P_2 X$. Define a map $\mathcal{R} : X \to \mathbb{R}^2$ by

$$\mathcal{R}(x) = (q(x), ||P_2 x||), \tag{5.8}$$

where $P_1 x = q(x) e \in X_1, q(x) \in \mathbb{R}$.

Remark 5.1. The map \mathcal{R} is Lipschitz continuous. Let \prod_0, \prod_+ and \prod_- denote the sets of points $(y_1, y_2) \in \mathbb{R}^2$ such that $y_2 = 0, y_2 > 0$ and $y_2 < 0$, respectively. Then \mathcal{R} can be viewed as a rigid rotation of X about $X_1 \cong \prod_0$ into $\overline{\prod}_+$ with $\mathcal{R}X_1 = \prod_0$.

Remark 5.2. If X is a separable Hilbert space with an orthonormal basis $\{e^i\}_1^{\infty}$, then the rotation map $\mathcal{R}: X \to \mathbb{R}^2$ about $X_1 = \operatorname{span}\{e^1\}$ is

$$x \mapsto \mathcal{R}(x) = \left(x_1, \left(\sum_{i=2}^{\infty} (x_i)^2\right)^{\frac{1}{2}}\right),$$

where $x = \sum_{i=1}^{\infty} x_i e^i$.

Remark 5.3. Since it is assumed that the norm $\|\cdot\|$ of the Banach space X is Gâteaux differentiable, it follows from Lemma 5.1 that Gâteaux derivative of \mathcal{R} exists and is uniformly bounded for all $x \in X$. Thus, there exists a constant M > 0 such that

$$d^{(2)}\mathcal{R}(x): \bigwedge^2 X \to \bigwedge^2 \mathbb{R}^2 \simeq \mathbb{R}$$

satisfies

$$\left\|d^{(2)}\mathcal{R}(x)\right\| \le M.$$

Let $v = \mathcal{R} \circ \psi : S^1 \to \overline{\prod}_+$. Since ψ is Frechét differentiable at $0 \in S^1$ and $\psi(0) = 0 \in X$, it can be assumed that

$$\psi(r) = d\psi(0)s + g(s)$$

where g(s) = o(s) near s = 0. Thus,

$$P_1\psi(s) = d\psi(0)s + P_1g(s) \in X_1,$$

 $P_2\psi(s) = P_2g(s) \in X_2,$

and $v(s) = (v_1(s), v_2(s))$ where $v_1(s) = q(P_1\psi(s)), v_2(s) = ||P_2\psi(s)||$ implies that

$$v_1(s) = ms + o(s), \quad 0 \le v_2(s) = o(s),$$
(5.9)

near s = 0, where $0 \neq m = q(P_1 d\psi(0))(1)$. Without loss of generality, assume that m > 0. If a > m, there exists $0 < \varepsilon < \pi$ such that

$$0 \le v_2(s) \le a|v_1(s)|, \quad -\varepsilon < s < \varepsilon$$

and $sv(s) > 0, s \neq 0$. Thus $v((-\varepsilon, \varepsilon))$ is a curve from $v(-\varepsilon)$ to $v(\varepsilon)$ through 0 and lying in the sectorial region

$$T = \left\{ (v_1, v_2) \in R^2 : 0 \le v_2 \le a |v_1| \right\},\$$

see Figure 5.2.



Figure 5.2: A sector T at 0 in \mathbb{R}^2 .

Next define $\tilde{\psi} : S^1 \to X$ by $\tilde{\psi}(s) = \psi(s)$ on $s \in S^1 \setminus (-\varepsilon, \varepsilon)$ and $\tilde{\psi} : [-\varepsilon, \varepsilon] \to X$ is one-to-one defined by linear interpolation from $\psi(-\varepsilon)$ to $P_1\psi(-\varepsilon)$, from $P_1\psi(-\varepsilon)$ to $P_1\psi(\varepsilon)$ with $\tilde{\psi}(0) = 0$ and from $P_1\psi(\varepsilon)$ to $\psi(\varepsilon)$. If $\tilde{v} = \mathcal{R} \circ \tilde{\psi} : S^1 \to \mathbb{R}^2$, then v, \tilde{v} differ only possibly on $(-\varepsilon, \varepsilon)$. Both

v and \tilde{v} map $[-\varepsilon, \varepsilon]$ to curves from $v(-\varepsilon)$ to $v(\varepsilon)$ through $0 = v(0) = \tilde{v}(0)$ and passing from one half of T to the other as s passes through 0.

The main result in this section, Proposition 5.2, shows the existence of a positive lower bound for $\sigma_2(h)$ independent of $h \in \sum(\psi, X)$. This was deduced in the original paper of Muldowney [78] on systems in \mathbb{R}^n from a classical result on the existence of a surface of minimum area. The proof given here for a general Banach space is adapted from a Brouwer degree argument used in Li [51] and Li and Muldowney [53]. A good reference for degree theory is the book by Lloyd [59].

Proposition 5.2. Let X be a Banach space satisfying the Radon-Nikodym property whose norm is Gâteaux differentiable. Suppose that $\psi \in Lip(\partial B \rightarrow X)$ is a simple closed curve in X. Then there exists $\delta > 0$ such that

 $\sigma_2(h) \ge \delta$

for all $h \in \sum (\psi, X)$.

Proof. As in the preceding discussion, it is assumed that $\psi(0) = 0 \in X$ and ψ is Frechét differentiable at 0. Let $h \in \sum(\psi, X)$ and $\Re = \mathcal{R} \circ h : \overline{B} \to \overline{\prod}_+$ and $v = \mathcal{R} \circ \psi : S^1 \to \overline{\prod}_+$ so that $\Re(\partial B) = v(S^1)$, where \mathcal{R} is defined by (5.8). Remark 5.3 implies that there exists a constant M such that

$$\left\|d^{(2)}\mathcal{R}(x)\right\| \le M$$

for all $x \in X$. Therefore,

$$\int_{B} |d^{(2)}\mathfrak{R}| = \int_{B} |d^{(2)}\mathcal{R}(h) \circ d^{(2)}h(r)| dr$$

$$\leq \int_{B} ||d^{(2)}\mathcal{R}(h)|| ||d^{(2)}h(r)|| dr$$

$$\leq M \int_{B} ||d^{(2)}h(r)||$$

$$\leq M \int_{B} ||d^{(2)}h|| = M\sigma_{2}(h).$$
(5.10)

Since $h|_{\partial B}$ is one-to-one and $h(\partial B) = \psi(S^1)$, the connected components of $\mathbb{R}^2 \setminus \mathfrak{R}(\partial B) = \mathbb{R}^2 \setminus v(\partial B)$ are the same for every $h \in \sum(\psi, X)$ and thus the

Brouwer degree of \mathfrak{R} relative to B, deg(\mathfrak{R}, B, \cdot), is a constant and independent of h on each component. It will be shown that there is a component Δ of $\mathbb{R}^2 \setminus \mathfrak{R}(\partial B)$ such that deg(\mathfrak{R}, B, p) = ± 1 if $p \in \Delta$. It follows from (5.10) that

$$\delta = rac{1}{M} ext{area} \Delta \leq \sigma_2(h)$$

for all $h \in \sum (\psi, X)$.

Let $\tilde{\psi}$ be as in the preliminary discussion. It follows that there exists $\tilde{h} \in \sum(\tilde{\psi}, X)$ and an open arc $\gamma \subset \partial B$ such that $h(\gamma) = \psi((-\varepsilon, \varepsilon)), \tilde{h}(\gamma) = \tilde{\psi}((-\varepsilon, \varepsilon))$ and $r_0 \in \gamma, h(r_0) = \tilde{h}(r_0) = 0$. Recall $\mathfrak{R} = \mathcal{R} \circ h$ and let $\tilde{\mathfrak{R}} = \mathcal{R} \circ \tilde{h}$; $r \in \partial B \setminus \gamma; \mathfrak{R}(\gamma) \subset T, \mathfrak{R}(\gamma) \subset T$ and both curves pass from one convex half of T to the other half as r passed through r_0 . Further, $\mathfrak{R}(\gamma)$ consists of three line



Figure 5.3: The map \mathfrak{R} . Note $v(\pm \varepsilon) = (P_1 \psi(\pm \varepsilon), ||P_2 \psi(\pm \varepsilon)||)$.

segments $[\psi(-\varepsilon), P_1\psi(-\varepsilon)], [P_1\psi(-\varepsilon), P_1\psi(\varepsilon)] \in \Pi_0$ and $[P_1\psi(\varepsilon), \psi(\varepsilon)]$. Since $\Re(x) = 0$ if and only if x = 0, it follows that, if $r \in \partial B$, $0 = \tilde{\Re}(s) = \mathcal{R} \circ \tilde{h}(s)$ if and only if $r = r_0$. Therefore, the compact set $\tilde{\Re}(\partial B \setminus \gamma) = \Re(\partial B \setminus \gamma)$ is a positive distance from $0 \in \mathbb{R}^2$ and so does not intersect $B(0, \rho) \subset \mathbb{R}^2$ for some $\rho > 0$; see Figure 5.3. In particular,

$$\mathfrak{R}(\partial B) \cap B(0,\rho) = (-\rho,\rho) \subset \Pi_0. \tag{5.11}$$
Since $\mathfrak{R}(s) = \mathfrak{\tilde{R}}(s)$ if $r \in \partial B \setminus \gamma$ and the line segment $[\mathfrak{R}(s), \mathfrak{\tilde{R}}(s)] \subset T$ if $r \in \gamma$, the Poincaré-Bohl theorem, [59], page 25, implies that $\deg(\mathfrak{R}, B, \cdot) = \deg(\mathfrak{\tilde{R}}, B, \cdot)$ on $\mathbb{R}^2 \setminus T$. From (5.11), the sets $\Pi_-, B(0, \rho) \cap \Pi_+$ are subsets of Δ_1, Δ_2 , connected components of $\mathbb{R}^2 \setminus \mathfrak{\tilde{R}}(\partial B)$. The interval $[-\rho, \rho] \subset \Pi_0$ is a subset of $\mathfrak{\tilde{R}}(\partial B)$ and is in the boundary of each of these sets. Further, $\mathfrak{\tilde{R}}$ is one-to-one on $\mathfrak{\tilde{R}}^{-1}([-\rho, \rho])$ and $\deg(\mathfrak{\tilde{R}}, B, \cdot) = 0$ on Δ_1 since $\mathfrak{R}(\partial B) \subset \overline{\Pi_+}$. Therefore, $\deg(\mathfrak{\tilde{R}}, B, \cdot) = \pm 1$ on $B(0, \rho) \cap \Pi_+ \subset \Delta_2$. The Poincaré-Bohl theorem then implies $\deg(\mathfrak{R}, B, \cdot) = \pm 1$ on $B(0, \rho) \setminus T$ and hence

$$\delta = \frac{1}{M} \operatorname{area}(B(0,\rho) \setminus T) \le \sigma_2(h).$$

5.2 Bendixson Criterion

In this section, a generalization of the Bendixson criterion to autonomous differential equations in a Banach spaces will be developed. Let A be a sectorial operator in a Banach space X and f(u) be continuously differentiable from X^{α} into X and $0 \leq \alpha < 1$. Consider the initial value problem

$$\frac{du}{dt} + Au = f(u),$$

$$u(0) = u_0 \in X^{\alpha}.$$
(5.12)

In the following, let $u(t, u_0) \in X^{\alpha}$ denote the solution of (5.12). The linear variational differential equation of (5.12) at a solution $u(t, u_0)$ is

$$\frac{dv}{dt} + Av = \frac{\partial f}{\partial u}(u(t, u_0))v, \qquad (5.13)$$

and its second compound differential equation defined on $\bigwedge^2 X^{\alpha}$ is

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f^{[2]}}{\partial u}(u(t, u_0))w.$$
(5.14)

Suppose that X^{α} is a Banach space satisfying the Radon-Nikodym property whose norm is Gâteaux differentiable. Let $D \subset X^{\alpha}$ be a simply connected open subset and $B = \{x \in \mathbb{R}^2 : |x| < 1\}$. Consider a 2-dimensional surface $h_0 \in \operatorname{Lip}(\overline{B} \to D)$. Suppose that $h_t(r) = u(t, h_0(r))$ is defined for all t and $r \in \overline{B}$. Then h_t is a 2-dimensional surface and

$$\sigma_2(h_t) = \int_B \left\| d^{(2)} h_t \right\|_{\alpha} = \int_B \left\| \frac{\partial h_t}{\partial r_1} \wedge \frac{\partial h_t}{\partial r_2} \right\|_{\alpha}$$
(5.15)

where $\|\cdot\|_{\alpha}$ is any norm on $\bigwedge^2 X^{\alpha}$. Proposition 5.2 implies the following Bendixson criterion.

Bendixson Criterion. Suppose that

- (i) $D \subset X^{\alpha}$ is a simply connected open set;
- (ii) the solutions of (5.12) exist for all t if $u_0 \in D$;
- (iii) ψ is a simple closed curve in D;
- (iv) there is a simple 2-dimensional surface $h_0 \in \sum(\psi, D)$ such that $\sigma_2(h_t)$ tends to zero as t tends to infinity under (5.12).

Then ψ cannot be invariant with respect to (5.12).

Remark 5.4. Condition (*iv*) is irrespective of the norm on $\bigwedge^2 X^{\alpha}$ that is used to define $\sigma_2(h_t)$.

Since

$$dh_t(r) = rac{\partial u(t,h_0(r))}{\partial u_0} ~~\cdot~ dh_0(r)$$

satisfies the linear variational differential equation (5.13) with $u_0 = h_0(r)$, it follows from the Binet-Cauchy identity that

$$d^{(2)}h_t(r) = \frac{\partial u(t,h_0(r))}{\partial u_0}^{(2)} \cdot d^{(2)}h_0(r)$$

which satisfies the second compound differential equation (5.14). This observation can be used to derive a concrete sufficient condition for Assumption (iv) of **Bendixson Criterion**.

Theorem 5.3. Suppose that

- (i) $D \subset X^{\alpha}$ is a simply connected open set;
- (ii) the solutions of (5.12) exist for all t if $u_0 \in D$;
- (iii) the family of linear systems

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f^{[2]}}{\partial u}(u(t, u_0))w, \qquad u_0 \in S,$$

is equi-asymptotically stable in $\bigwedge^2 X^{\alpha}$ if S is a compact subset of D.

Then D contains no simple closed curve that is invariant with respect to (5.12).

The equi-asymptotic stability in Assumption (*iii*) of Theorem 5.3 means that if S is a compact subset of D, then, for all $u_0 \in S$, the linear systems (5.14) are asymptotically stable and the limit

$$\lim_{t\to\infty}\|w(t)\|_{\alpha}=0$$

is uniform with respect to $u_0 \in S$, where $w(t) \in \bigwedge^2 X^{\alpha}$ is a solution of (5.14) and $\|\cdot\|_{\alpha}$ is any norm on $\bigwedge^2 X^{\alpha}$.

Proof. Suppose that $\psi \in \operatorname{Lip}(S^1 \to D)$ is a simple closed curve which is invariant with respect to (5.12). Let $h_0 \in \sum(\psi, D)$ be a simple 2dimensional surface which is one-to-one on ∂B and $h_0(\partial B) = \psi(S^1)$, where $B = \{x \in \mathbb{R}^2 : |x| < 1\}$. Then $r \mapsto h_t(r) = u(t, h_0(r))$ is also a simple 2-dimensional surface which is one-to-one on ∂B and $h_t(\partial B) = \psi(S^1)$. Since $dh_t(r) = \frac{\partial u(t, h_0(r))}{\partial u_0} \cdot dh_0(r)$ satisfies the linear variational differential equation (5.13) with $u_0 = h_0(r), d^{(2)}h_t(r) = \frac{\partial u(t, h_0(r))}{\partial u_0}^{(2)} \cdot d^{(2)}h_0(r)$ is a

solution of the second compound differential equation

$$\frac{dw}{dt} + A^{[2]}w = \frac{\partial f^{[2]}}{\partial u}(u(t, u_0))w, \qquad u_0 = h_0(r), \quad r \in \overline{B}.$$

These linear systems are equi-asymptotically stable and

$$\lim_{t \to \infty} \left\| \frac{\partial u(t, h_0(r))}{\partial u_0}^{(2)} \right\|_{\alpha} = 0$$

uniformly with respect to $r \in \overline{B}$, which implies that

$$\lim_{t \to \infty} \|d^{(2)}h_t(r)\|_{\alpha} = 0$$

uniformly with respect to $r \in \overline{B}$. Thus $\lim \sigma_2(h_t) = 0$. But Proposition 5.2 shows that $\delta \leq \sigma_2(h_t)$ for some $\delta > 0$. Therefore, ψ can not be invariant with respect to (5.12).

Remark 5.5. Theorem 5.3 rules out the existence of nontrivial periodic orbits, homoclinic loops and heteroclinic loops. For some systems, the equi-asymptotic stability of the family of linear systems (5.14) can be established by constructing a suitable Lyapunov function in $\bigwedge^2 X^{\alpha}$. For instance, (5.14) is equi-asymptotically stable if there exists a positive definite function V(w), such that $\frac{d^+V}{dt}\Big|_{(5.14)}$ is negative definite, and V and $\frac{d^+V}{dt}\Big|_{(5.14)}$ are both independent of u_0 .

Remark 5.6. The results in this section are motivated by the high-dimensional Bendixson criterion for an autonomous ordinary differential equation in \mathbb{R}^n established by Muldowney [78]. The proof of Theorem 5.3 is adapted from the idea used in Theorem 4.1 of [78].

Remark 5.7. An application of Theorem 5.3 is to combine with Theorem 4.12, Section 4.2 to show the existence of steady state solutions.

Consider a scalar reaction diffusion equation

$$u_t = u_{xx} + f(x, u, u_x), \quad 0 < x < L, \quad t > 0,$$

$$u(t, 0) = u(t, L), \quad u_x(t, 0) = u_x(t, L).$$

(5.16)

Here $f \in C^2((0,L) \times \mathbb{R} \times \mathbb{R} \to \mathbb{R})$. Let $X = H^1_{per}(0,L) = \{\phi \in L^2(0,L) : \phi' \in L^2(0,L), \phi(0) = \phi(L), \phi'(0) = \phi'(L)\}$. Suppose that if $u_0 \in X$, $u(t,x) = u(t,u_0) \in X$ is a solution which exists for all $t \ge 0$. The linear variational equation of (5.16) at a solution u(t,x) is

$$v_t = v_{xx} + a(t, x)v_x + b(t, x)v, \quad 0 < x < L, \quad t > 0,$$

$$v(t, 0) = v(t, L), \quad v_x(t, 0) = v_x(t, L),$$
(5.17)

where

$$a(t,x) = \frac{\partial f}{\partial u_x}(x, u(t, x, u_0(x)), u_x(t, x, u_0(x))),$$

$$b(t,x) = \frac{\partial f}{\partial u}(x, u(t, x, u_0(x)), u_x(t, x, u_0(x))).$$
(5.18)

The pointwise representation of $w(t) \in \bigwedge^2 X$ is

$$w(t, x_1, x_2) = \det \left[\langle x_i, w(t) \rangle \right]$$

which satisfies the second compound differential equation of (5.17) defined on $\bigwedge^2 X$

$$w_t = \sum_{j=1}^2 w_{x_j x_j} + \sum_{j=1}^2 a(t, x_j) w_{x_j} + \sum_{j=1}^2 b(t, x_j) w.$$
(5.19)

The eigenvalues of the Laplace equation

$$\Delta u = -\lambda_i u,$$

$$u(0) = u(L), \quad u_x(0) = u_x(L)$$
(5.20)

are $\lambda_1 = 0, \lambda_{2n} = \left(\frac{n\pi}{L}\right)^2, \lambda_{2n+1} = \left(\frac{n\pi}{L}\right)^2$. Suppose that, for $u, v \in \mathbb{R}$,

$$\begin{aligned} \left| \frac{\partial f}{\partial v}(x, u, v) \right| &\leq A(x), \\ \frac{\partial f}{\partial u}(x, u, v) &\leq B(x). \end{aligned} \tag{5.21}$$

Consider a Lyapunov function

$$V := \frac{1}{2} \int_{(0,L)^2} w^2 = \frac{1}{2} \int_{(0,L)^2} w^2(t,x_1,x_2) dx_1 dx_2.$$

The discussion in the proof of Corollary 3.33, Section 3.4 implies that

$$\begin{aligned} \frac{d^+V}{dt} &\leq \sum_{j=1}^2 \int_{(0,L)^2} \left(\frac{A(x_j)}{2\nu_j(x_j)} - 1 \right) (w_{x_j})^2 \\ &+ \int_{(0,L)^2} \sum_{j=1}^2 \left(B(x_j) + \frac{A(x_j)\nu_j(x_j)}{2} \right) w^2, \end{aligned}$$

for any positive functions $\nu_1(x), \nu_2(x)$. If $\max_x A(x) > 0, 0 \le x \le L$, then a choice

$$\nu_j(x) = \max_x A(x) := \nu, \quad j = 1, 2,$$

implies that

$$\frac{d^+V}{dt} \le \mu V,$$

where

$$\mu = -\lambda_1 - \lambda_2 + 4 \max_x \left\{ B(x) + \frac{A(x)\nu}{2} \right\}.$$

If $\max_x A(x) = 0$, then the function $f(x, u, u_x) = f(x, u)$ which is independent of u_x^{x} , and thus

$$\mu = -\lambda_1 - \lambda_2 + 2 \max_x B(x).$$

Theorem 5.3 implies the following corollary.

Corollary 5.4. Suppose that the solutions of (5.16) exist for all $t \ge 0$ and $u_0 \in X$. If $\max_x |A(x)| > 0$, $0 \le x \le L$, and

$$\max_{x} \left\{ 2B(x) + A(x)\nu \right\} < \frac{\pi^2}{2L^2}$$

where $\nu = \max_{x} A(x)$ or if $f(x, u, u_x) = f(x, u)$ and

$$\max_x B(x) < \frac{\pi^2}{2L^2},$$

where A(x), B(x) is defined by (5.21), then there is no simple closed curve that is invariant with respect to (5.16) in $L^2(0, L)$.

Example 5.3. Consider a scalar reaction diffusion equation

$$u_t = u_{xx} + pu - u^3 + q(x), \quad 0 < x < L, t > 0,$$

$$u(t, 0) = u(t, L), \quad u_x(t, 0) = u_x(t, L)$$
(5.22)

Suppose that p is a positive constant and q(x) is nonnegative continuous in x and $\max_{x} q(x) \leq Q$, $0 \leq x \leq L$. Let $X = H^{1}_{per}(0, L)$. Since

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_0^L u^2 dx &= \int_0^L u_t u dx \\ &= \int_0^L [-u_x^2 + p u^2 - u^4 + p(x) u] dx \\ &\le \left(p + \frac{Q}{2} \right) \int_0^L u^2 dx + \frac{QL}{2}, \end{aligned}$$

and

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_0^L u_x^2 &= -\int_0^L u_t u_{xx} dx \\ &= \int_0^L -u_{xx}^2 + p u_x^2 - q(x) u_{xx}] dx \\ &\le \left(p + \frac{Q}{2} \right) \int_0^L u_x^2 dx + \frac{QL}{2}, \end{aligned}$$

the Gronwall inequality implies that the solution u(t, x) of (5.22) exists in X for all $t \ge 0$. The linear variational equation of (5.22) at a solution u(t, x) is

$$v_t = v_{xx} + (p - 2u^2)v,$$

and the pointwise representation of the second compound differential equation defined on $\bigwedge^2 X$ is

$$w_t = w_{x_1x_1} + w_{x_2x_2} + (2p - u^2(t, x_1) - u^2(t, x_2))w.$$
 (5.23)

If

$$-\lambda_1 - \lambda_2 + 2p < 0,$$

Corollary 5.4 implies that there is no simple closed curve that is invariant with respect to (5.16) in $L^2(0, L)$. Since the ω limit set of a bounded solution is nonempty, Theorem 4.12 of Section 4.2 implies that (5.22) has at least one steady state solution, which is nonconstant if p(x) is nonconstant.

Example 5.4. Consider a reaction diffusion system

$$u_t = d_1 u_{xx} + au - u^3 - uv^2,$$

$$v_t = d_2 v_{xx} - bv - v^3 + u^2 v, \quad 0 < x < L, \quad t > 0,$$
(5.24)

with Dirichlet boundary conditions or Neumann boundary conditions or periodic boundary conditions. Suppose that $a, b, d_1, d_2 > 0$. Let X be a subspace $H^1(0, L) \times H^1(0, L)$ with corresponding boundary conditions. Let X be a subspace $H^1_{per}(0, L)$. Since

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_0^L (u^2 + v^2) dx &= \int_0^L (u_t u + v_t v) dx \\ &= \int_0^L [-d_1 u_x^2 + a u^2 - u^4 - u^2 v^2] dx \\ &+ \int_0^L [-d_2 v_x^2 - b v^2 - v^4 + u^2 v^2] dx \\ &\leq a \int_0^L u^2 dx, \end{aligned}$$

and

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} \int_0^L (u_x^2 + v_x^2) dx &= -\int_0^{2\pi} (u_t u_{xx} + v_t v_{xx}) dx \\ &= \int_0^L [-d_1 u_{xx}^2 + a u_x^2 - 3 u^2 u_x^2 - 2 u u_x v v_x] dx \\ &+ \int_0^L [-d_2 v_{xx}^2 - b v_x^2 - 3 v^2 v_x^2 + 2 u u_x v v_x] dx \\ &\leq a \int_0^{2\pi} u_x^2 dx, \end{aligned}$$

the solutions of (5.24) exist in $X \times X$ for all $t \ge 0$. The linear variational equation of (5.24) at a solution (u(t, x), v(t, x)) is

$$\begin{split} p_t &= d_1 p_{xx} + (a - 2u^2 - v^2) p - 2uvq, \\ q_t &= d_2 q_{xx} + (-b - 2v^2 - u^2) q + 2uvp, \quad 0 < x < L, \quad t > 0, \end{split}$$

and the pointwise representation of the second compound differential equation defined on $\bigwedge^2 X$ is

$$\begin{aligned} \frac{\partial w_{11}}{\partial t} &= d_1 (\Delta_1 w_{11} + \Delta_2 w_{11}) + (a_{11}(t, x_1) + a_{11}(t, x_2)) w_{11} \\ &+ a_{12}(t, x_1) w_{21} + a_{12}(t, x_2) w_{12}, \\ \\ \frac{\partial w_{22}}{\partial t} &= d_2 (\Delta_1 w_{22} + \Delta_2 w_{22}) + (a_{22}(t, x_1) + a_{22}(t, x_2)) w_{22} \\ &+ a_{21}(t, x_1) w_{12} + a_{21}(t, x_2) w_{21}, \\ \\ \frac{\partial w_{12}}{\partial t} &= d_1 \Delta_1 w_{12} + d_2 \Delta_2 w_{12} + (a_{11}(t, x_1) + a_{22}(t, x_2)) w_{12} \\ &+ a_{12}(t, x_1) w_{22} + a_{21}(t, x_2) w_{11}, \end{aligned}$$

where $\Delta_i w = \frac{\partial^2}{\partial x_i \partial x_i}$ and

$$a_{11} = a - 2u^2 - v^2$$
, $a_{12} = -2uv$, $a_{21} = 2uv$, $a_{22} = b - 2v^2 - u^2$

(see (3.26) of Section 3.2 for details). Consider a Lyapunov function

$$V(t) := \frac{1}{2!} \int_{(0,L)^2} \left((w_{11})^2 + (w_{22})^2 + 2(w_{12})^2 \right).$$

Then the derivative of V calculated for (3.28) of Section 3.2 implies that

$$\frac{d^+V}{dt} \leq \int_{(0,L)^2} \left(\mu_1(t)(w_{11})^2 + \mu_2(t)(w_{22})^2 + 2\mu_3(t)(w_{12})^2 \right)$$
(5.25)

where

$$\begin{array}{lll} \mu_1(t) &=& -(\lambda_1 + \lambda_2) \, d_1 + \max_x \left\{ 2a - 4u^2(t, x) - 2v^2(t, x) \right\} \\ &\leq& -(\lambda_1 + \lambda_2) \, d_1 + 2a, \\ \mu_2(t) &=& -(\lambda_1 + \lambda_2) \, d_2 + \max_x \left\{ -2b - 4v^2(t, x) - 2u^2(t, x) \right\} \\ &\leq& -(\lambda_1 + \lambda_2) \, d_2 - 2b, \\ \mu_3(t) &=& -\lambda_1(d_1 + d_2) + \max_x \left\{ a - u^2(t, x) - v^2(t, x) \right\} \\ &\quad + \max_x \left\{ -b - 2v^2(t, x) - u^2(t, x) \right\} \\ &\leq& -\lambda_1(d_1 + d_2) + a - b, \end{array}$$

and $\lambda_1 \leq \lambda_2 \leq \cdots$ are eigenvalues of the Laplace equation $u_{xx} = -\lambda_i u$ with corresponding boundary conditions.

Corollary 5.5. If

$$a<\min\left\{rac{\lambda_1+\lambda_2}{2}d_1,b+\lambda_1(d_1+d_2)
ight\},$$

then the system (5.24) contains no simple closed curve that is invariant in $L^2(0,L)$.

Chapter 6

An Example from Epidemiology

A general account of spatial dispersion of biological populations is contained in the work of Skellam [97], and in the recent books by Okubo [86] and Murray [81]. Early results on models for the spatial dependence of the spread of epidemics are presented by Fisher [32], Kolmogorov, Petrovsky and Piscounov [49] and Turing [105]. In particular, studies on infectious diseases with diffusion have been discussed by Capasso [13], Capasso and Fortunato [14], deMottoni, Orlandi and Tesei [24], Gurtin and MacCamy [36], MacCamy [64], Kallen, Arcuri and Murrary [48], and Webb [108].

In this chapter, as an example, the Bendixson criterion of Chapter 6 is used to preclude the existence of nontrivial periodic solutions to a diffusive epidemiological SIR model

$$\begin{cases} S_t = d_1 S_{xx} + \Lambda - \beta(x) I S - b_1 S, \\ I_t = d_2 I_{xx} + \beta(x) I S - (b_2 + \gamma) I, & 0 < x < 1, \quad t > 0 \\ R_t = d_3 R_{xx} + \gamma I - b_3 R, \end{cases}$$
(6.1)

with homogeneous Neumann boundary conditions:

$$\begin{cases} S_x(t,0) = S_x(t,1) = 0, \\ I_x(t,0) = I_x(t,1) = 0, \\ R_x(t,0) = R_x(t,1) = 0, \end{cases} \quad (6.2)$$

and initial conditions:

$$\begin{cases} S(0,x) = S_0(x) \ge 0, \\ I(0,x) = I_0(x) \ge 0, \\ R(0,x) = R_0(x) \ge 0. \end{cases} \quad 0 \le x \le 1, \quad (6.3)$$

Here S, I, R denote the populations that are susceptible, infectious, and recovered, see Figure 6.1. The parameter $b_i > 0, i = 1, 2, 3$ is the death rate, $\gamma > 0$ is the recovery rate and $d_i \ge 0, i = 1, 2, 3$ is the diffusion coefficients. The parameter $\beta(x)$ is the effective per capita contact rate among individuals which is assumed to be spatially dependent and $\beta(x) \in C([0,1])$ are nonnegative, $\|\beta\|_{\infty} = \max_{0 \le x \le 1} \beta(x) > 0$. It is also assumed that there is a constant recruitment $\Lambda > 0$ for this model.



Figure 6.1: A diffusive SIR model.

The model without diffusion, a system of ordinary differential equations, is well understood; see Hethcote [44] and Brauer, van den Driessche and Wu [9]. This chapter presents some results on the existence and the structure of the global attractor of (6.3), and the nonexistence of periodic solutions, when diffusion is added and the contact rate β is spatial dependent. Consequently, the existence of at least one positive, x-dependent endemic steady state solution is obtained.

The approach to the existence and boundedness is motivated by the work of Dung and Smith [28] in their study of a parabolic system modeling microbial populations with competition.

6.1 Existence and Boundedness

Let $C_+([0,1])$ be the cone of nonnegative functions in the Banach space C([0,1]) with the usual supremum norm $\|\cdot\|_{\infty}$. For $u = (u_1, \cdots, u_k) \in$

$$X_+ \times \cdots \times X_+$$
, define $||u||_{\infty} = \max_{1 \le i \le k} |u_i|$.

Theorem 6.1. The system (6.1)-(6.3) generates a nonlinear local semidynamical system $T(t)_{t\geq 0}$ on the space $C_+([0,1]) \times C_+([0,1]) \times C_+([0,1])$. Moreover, the solutions to (6.1)-(6.3) are in fact classical solutions.

Remark 6.1. The system (6.1)-(6.3) is a system of semilinear parabolic partial differential equations. The standard analysis on the existence by fixed point theory, see Rothe [92], Pazy [87], Henry [42] and Wu [111].

Let $\tau(S_0, I_0, R_0)$ be the maximal interval of existence of the system (6.1)-(6.3). The first two equations in (6.1) do not contain R, which allows the reduction of (6.1) to

$$\begin{cases} S_t = d_1 S_{xx} + \Lambda - \beta(x) IS - b_1 S, \\ I_t = d_2 I_{xx} + \beta(x) IS - (b_2 + \gamma) I. \end{cases}$$
(6.4)

In the following, the system (6.4) is studied instead of (6.1). For simplicity, let $X_+ = C_+([0,1]) \times C_+([0,1])$.

Consider the following ordinary differential equation

$$\begin{cases} u_t = \Lambda - \beta(x)Iu - b_1u, & 0 < x < 1, \quad t > 0 \\ u(0) = \|S_0\|_{\infty}. \end{cases}$$
(6.5)

Then

$$u(t) \leq \left(\|S_0\|_{\infty} - \frac{\Lambda}{b_1} \right) e^{-b_1 t} + \frac{\Lambda}{b_1}$$

$$\leq \max \left\{ \|S_0\|_{\infty}, \frac{\Lambda}{b_1} \right\} := S_{max},$$

and thus

$$\limsup_{t \to \infty} u(t) \le \frac{\Lambda}{b_1}$$

Since

$$\begin{aligned} -S_t + d_1 S_{xx} + \Lambda - \beta(x) IS - b_1 S &= -u_t + \Lambda - \beta(x) Iu - b_1 u, \\ S(0, x) &\leq u(0), \end{aligned}$$

the comparison theorem (see Smoller [103], page 94) implies that

$$S(t, x) \le u(t) \le S_{max}, \quad \text{for all } 0 \le x \le 1, \ t \ge 0,$$
$$\limsup_{t \to \infty} S(t, x) \le \frac{\Lambda}{b_1}, \quad \text{for all } 0 \le x \le 1.$$
(6.6)

Now for any $\varepsilon \geq 0$, define a feasible region

$$\Gamma_{\varepsilon} := \left\{ (S(x), I(x)) \in X_{+} : \|S\|_{\infty} \le \frac{\Lambda}{b_{1}} + \varepsilon \right\}.$$
(6.7)

Proposition 6.2. Γ_{ε} is positively invariant for (6.2)-(6.4) and for $\varepsilon > 0$, Γ_{ε} is an absorbing set in X_+ .

Proof. For any $\varepsilon \geq 0$ and any solution (S(t), I(t)) of (6.2)-(6.4) with $||S_0||_{\infty} \leq \frac{\Lambda}{b_1} + \varepsilon$, since

$$\begin{aligned} -S_t + d_1 S_{xx} + \Lambda - \beta(x) IS - b_1 S &\geq -\beta(x) I\left(\frac{\Lambda}{b_1} + \varepsilon\right) - b_1 \varepsilon, \\ S_x(t,0) &= S_x(t,1) &= 0, \\ S(0,x) &\leq \frac{\Lambda}{b_1} + \varepsilon, \end{aligned}$$

a comparison theorem implies that

$$S(t,x) \leq \frac{\Lambda}{b_1} + \varepsilon,$$
 for all $0 \leq x \leq 1, t \geq 0.$

Thus Γ_{ε} is positively invariant.

For any bounded set \mathcal{B} in X_+ , there exists a M such that $||S||_{\infty} \leq M$ for every $(S(x), I(x)) \in \mathcal{B}$. Suppose that $\varepsilon > 0$. If $M \leq \frac{\Lambda}{b_1} + \varepsilon$, choose T = 0, otherwise choose $T = \frac{1}{b_1} \ln \frac{M - \frac{\Lambda}{b_1}}{\varepsilon}$. Then, for $t \geq T$,

$$S(t,x) \leq \left(\|S_0\|_{\infty} - \frac{\Lambda}{b_1} \right) e^{-b_1 t} + \frac{\Lambda}{b_1}$$

$$\leq \left(M - \frac{\Lambda}{b_1} \right) e^{-b_1 T} + \frac{\Lambda}{b_1}$$

$$\leq \frac{\Lambda}{b_1} + \varepsilon,$$

which implies that Γ_{ε} is an absorbing set in X_+ when $\varepsilon > 0$.

Since the solution of (6.2)-(6.4) is attracted to the set

$$\Gamma_0 = \left\{ (S(x), I(x)) \in X_+ : \|S\|_{\infty} \le \frac{\Lambda}{b_1} \right\}$$

for larger time, in the following the dynamics of (6.2)-(6.4) in Γ_0 is studied. It follows from

$$(S+I)_{t} = d_{1}S_{xx} + d_{2}I_{xx} + \Lambda - b_{1}S - (b_{2}+\gamma)I$$

that

$$\begin{split} \int_0^1 (S+I)_t dx &= \int_0^1 (d_1 S_{xx} + d_2 I_{xx} + \Lambda - b_1 S - (b_2 + \gamma)I) \ dx \\ &= d_1 S_x \left|_{x=0}^{x=1} + d_2 I_x \right|_{x=0}^{x=1} + \int_0^1 (\Lambda - b_1 S - (b_2 + \gamma)I) dx \\ &\leq \Lambda - b_m \int_0^1 (S+I) dx, \end{split}$$

where $b_m = \min\{b_1, b_2 + \gamma\}$, which implies that

$$\int_{0}^{1} (S(t,x) + I(t,x)) dx \leq \left(\int_{0}^{1} (S_{0}(x) + I_{0}(x)) dx - \frac{\Lambda}{b_{m}} \right) e^{-b_{m}t} + \frac{\Lambda}{b_{m}} \\
\leq \max \left\{ \int_{0}^{1} (S_{0}(x) + I_{0}(x)) dx, \frac{\Lambda}{b_{m}} \right\} \\
\leq \frac{\Lambda}{b_{m}} + \|I_{0}\|_{\infty} := C_{1}(\|I_{0}\|_{\infty}),$$
(6.8)

where $C_1(||I_0||_{\infty})$ is a positive continuous function, and thus

$$\limsup_{t\to\infty} \int_0^1 (S(t,x) + I(t,x)) dx \le \frac{\Lambda}{b_m}.$$

In particular,

$$\int_{0}^{1} I(t, x) dx \le C_{1}(\|I_{0}\|_{\infty}), \tag{6.9}$$

and

$$\limsup_{t \to \infty} \int_0^1 I(t, x) dx \le \frac{\Lambda}{b_m} := c_1.$$
(6.10)

The next proposition uses mathematical induction to estimate the L^p norm of *I*. The argument is standard and has been used by Dung and Smith [28].

Proposition 6.3. For each $0 , there exists a positive continuous function <math>C_p(||I_0||_{\infty})$ and a positive constant c_p such that

$$\|I(t)\|_{p} \le C_{p}(\|I_{0}\|_{\infty}), \qquad t \ge 0$$
(6.11)

1

and

$$\limsup_{t \to \infty} \|I(t)\|_p \le c_p. \tag{6.12}$$

Proof. Assume that (6.11) holds for some $p \ge 1$, in particular, it holds for p = 1 by (6.9) and (6.10). It is sufficient to prove that it holds for exponent 2p. Here only the bound on L_2 norm is proved. The general case can be obtained similarly and will be omitted. Since

$$\begin{aligned} \frac{d}{dt} \int_0^1 I^2 dx &= 2d_2 \int_0^1 II_{xx} dx + 2\int_0^1 \beta(x) SI^2 dx - 2(b_2 + \gamma) \int_0^1 I^2 dx \\ &\leq -2d_2 \int_0^1 I_x^2 dx + 2\|\beta\|_\infty \frac{\Lambda}{b_1} \int_0^1 I^2 dx - 2(b_2 + \gamma) \int_0^1 I^2 dx, \end{aligned}$$

the Nirenberg-Gagliardo inequality (see Henry [42], page 37) and Young's inequality imply that

$$\int_{0}^{1} I^{2} dx \leq C \left(\int_{0}^{1} I_{x}^{2} dx + \int_{0}^{1} I^{2} dx \right)^{\frac{1}{2}} \left(\int_{0}^{1} I dx \right) \\
\leq C \varepsilon_{1} \left(\int_{0}^{1} I_{x}^{2} dx + \int_{0}^{1} I^{2} dx \right) + \frac{C}{\varepsilon_{1}} \left(\int_{0}^{1} I dx \right)^{2},$$
(6.13)

where C and ε_1 are positive constants. Thus,

$$\begin{split} \frac{d}{dt} \int_{0}^{1} I^{2} dx \\ &\leq -2d_{2} \int_{0}^{1} I_{x}^{2} dx + 2\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} C \varepsilon_{1} \left(\int_{0}^{1} I_{x}^{2} dx + \int_{0}^{1} I^{2} dx \right) \\ &+ 2\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} \left(\int_{0}^{1} I dx \right)^{2} - 2(b_{2} + \gamma) \int_{0}^{1} I^{2} dx \\ &\leq 2 \left(-d_{2} + \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} C \varepsilon_{1} \right) \int_{0}^{1} I_{x}^{2} dx + 2 \left(-(b_{2} + \gamma) + \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} C \varepsilon_{1} \right) \int_{0}^{1} I^{2} dx \\ &+ 2\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} \left(\int_{0}^{1} I dx \right)^{2} \\ &\leq -2 \left(d_{2} - \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} C \varepsilon_{1} \right) \int_{0}^{1} I_{x}^{2} dx - 2 \left((b_{2} + \gamma) - \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} C \varepsilon_{1} \right) \int_{0}^{1} I^{2} dx \\ &+ 2M(\|I(t)\|_{1}) \\ &\coloneqq -2\alpha_{1} \int_{0}^{1} I_{x}^{2} dx - 2\alpha_{2} \int_{0}^{1} I^{2} dx + 2M(\|I(t)\|_{1}), \end{split}$$

where $M(\|I(t)\|_1) = \|\beta\|_{\infty \frac{\Lambda}{b_1} \frac{C}{c_1}} \left(\int_0^1 I(t) dx \right)^2$. Moreover, from (6.9) and (6.10),

$$M(||I(t)||_{1}) \leq ||\beta||_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} (C_{1}(||I_{0}||_{\infty}))^{2},$$

and

$$\limsup_{t \to \infty} M(\|I(t)\|_1) \le \|\beta\|_{\infty} \frac{\Lambda}{b_1} \frac{C}{\varepsilon_1} \left(\frac{\Lambda}{b_m}\right)^2 = \|\beta\|_{\infty} \frac{\Lambda}{b_1} \frac{C}{\varepsilon_1} (c_1)^2.$$

Choose $\varepsilon_1 < \frac{1}{C \|\beta\|_{\infty} \frac{\Lambda}{b_1}} \min\{d_2, b_2 + \gamma\}$. Then $\alpha_1, \alpha_2 > 0$ and

$$\int_{0}^{1} I^{2}(t, x) dx \leq \left(\int_{0}^{1} I_{0}^{2}(x) dx - \frac{M(\|I(t)\|_{1})}{\alpha_{2}} \right) e^{-2\alpha_{2}t} + \frac{M(\|I(t)\|_{1})}{\alpha_{2}} \\
\leq \max \left\{ \|I_{0}\|_{\infty}^{2}, \frac{\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} (C_{1}(\|I_{0}\|_{\infty}))^{2}}{\alpha_{2}} \right\} \\
\leq \max \left\{ \|I_{0}\|_{\infty}^{2}, \frac{\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} (\frac{\Lambda}{b_{m}} + \|I_{0}\|_{\infty})^{2}}{\alpha_{2}} \right\} := (C_{2}(\|I_{0}\|_{\infty}))^{2}$$
(6.14)

Therefore, $\|I(t)\|_2$ is defined for all $t \ge 0$ and there exists a positive constant $c_2 = \sqrt{\frac{\|\beta\|_{\infty} \frac{\Lambda}{b_1} \frac{C}{c_1}(c_1)^2}{\alpha_2}}$ $\limsup_{t \to \infty} \|I(t)\|_2 \le c_2.$

Theorem 6.4 shows an upper bound of $||I||_{\infty}$, whose proof is based on the integral equation of I(t) and the Sobolev compact embedding property.

Theorem 6.4. The solution of (6.2)-(6.4) exists for all $t \ge 0$ in Γ_0 . Furthermore, there exists a positive continuous function $C(||I_0||_{\infty})$ and a positive constant c, independent of the initial data I_0 , such that

$$\|I(t)\|_{\infty} \le C(\|I_0\|_{\infty}), \qquad t \ge 0, \tag{6.15}$$

and

$$\limsup_{t \to \infty} \|I(t)\|_{\infty} \le c. \tag{6.16}$$

Proof. First, consider problem (6.2)-(6.4) in the larger space $Y = L^p((0, 1))$. The solution I(t) satisfies

$$I(t) = e^{-Bt}I_0 + \int_0^t e^{-B(t-s)}g_2(S(s), I(s))ds, \qquad (6.17)$$

where $B = -d_2 \triangle + (b_2 + \gamma)I$ with homogeneous Neumann boundary conditions,

$$g_2(S(t), I(t))(x) = g_2(x, S(t, x), I(t, x)) = \beta(x)S(t, x)I(t, x).$$

The operator B generates an analytic semigroup $\{T(t) = e^{-Bt}\}_{t>0}$ on Y with $0 \in \rho(B)$. The semigroup of operators $\{T(t) = e^{-Bt}\}_{t>0}$ map Y into the space $Y^{\alpha} = D(B^{\alpha})$ with the norm $||u||_{Y^{\alpha}} = ||B^{\alpha}u||_{p}$, where $0 < \alpha < 1$ and B^{α} is the fractional power of B, see Pazy [87], page 74 and page 242. Choose p such that $\frac{1}{2p} < \alpha < 1$ and the embedding

$$Y^{\alpha} \xrightarrow[\text{cont.}]{} C^{\nu}([0,1]), \qquad 0 \le \nu < 2\alpha - \frac{1}{p}$$
(6.18)

is continuous, see Henry [42], page 39 and Pazy [87], page 243. For each $I_0 \in X_+$,

$$B^{\alpha}I(t) = B^{\alpha}e^{-Bt}I_0 + \int_0^t B^{\alpha}e^{-B(t-s)}g_2(S(s), I(s))ds.$$

From the L^p estimates of I(t) in Proposition 6.3 and $||S(t)||_{\infty} \leq \frac{\Lambda}{b_1}$, for $t \geq 0$, we have that there is a positive function $G_p = G_p(||I_0||_{\infty}) = ||\beta||_{\infty} \frac{\Lambda}{b_1} C_p(||I_0||_{\infty})$ such that

$$\begin{aligned} \|g_2(S(t),I(t))\|_p &= \left(\int_0^1 (\beta(x)S(t,x)I(t,x))^p dx\right)^{\frac{1}{p}} \\ &\leq \|\beta\|_{\infty} \frac{\Lambda}{b_1} \left(\int_0^1 I^p dx\right)^{\frac{1}{p}} \\ &\leq G_p(\|I_0\|_{\infty}), \end{aligned}$$

and there is a positive constant $g_p = \|\beta\|_{\infty} \frac{\Lambda}{b_1} c_p$, independent of $\|I_0\|_{\infty}$, such that

$$\limsup_{t\to\infty} \|g_2(S(t),I(t))\|_p \le g_p.$$

Therefore, there is a $\eta = \eta(I_0) > 0$ such that $||g_2(S(t), I(t))||_p \leq 2g_p$ for

 $t \geq \eta$. Consequently, for $t \geq \eta$,

$$\begin{split} \|I(t)\|_{Y^{\alpha}} &= \|B^{\alpha}I(t)\|_{p} \\ \leq & \|B^{\alpha}e^{-Bt}I_{0}\|_{p} + \int_{0}^{t} \|B^{\alpha}e^{-B(t-s)}\|_{\mathscr{L}(Y,Y)}\|g_{2}(S(s),I(s))\|_{p}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \int_{0}^{t}C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}\|g_{2}(S(s),I(s))\|_{p}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \int_{0}^{\eta}G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}ds \\ &+ \int_{\eta}^{t}2g_{p}C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \eta G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-\eta)^{-\alpha}e^{-\delta(t-\eta)} \\ &+ 2g_{p}C_{\alpha}\int_{0}^{t-\eta}r^{-\alpha}e^{-\delta r}dr \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \eta G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-\eta)^{-\alpha}e^{-\delta(t-\eta)} \\ &+ 2g_{p}C_{\alpha}\int_{0}^{\infty}r^{-\alpha}e^{-\delta r}dr, \end{split}$$

and thus $\|I(t)\|_{Y^\alpha}$ is defined for all $t\geq 0$ and

$$\limsup_{t \to \infty} \|I(t)\|_{Y^{\alpha}} \le 2g_p C_{\alpha} \int_0^\infty r^{-\alpha} e^{-\delta r} dr = 2g_p C_{\alpha} \delta^{-1+\alpha} \Gamma(1-\alpha).$$
(6.19)

Furthermore, the estimate (6.19) and the embedding (6.18) implies that (6.16) holds. For $t \ge 1$,

$$\begin{split} \|I(t)\|_{Y^{\alpha}} &= \|B^{\alpha}I(t)\|_{p} \\ \leq \|B^{\alpha}e^{-Bt}I_{0}\|_{p} + \int_{0}^{t} \|B^{\alpha}e^{-B(t-s)}\|_{\mathscr{L}(Y,Y)}\|g_{2}(S(s),I(s))\|_{p}ds \\ \leq C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \int_{0}^{t}C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}\|g_{2}(S(s),I(s))\|_{p}ds \\ = C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + G_{p}(\|I_{0}\|_{\infty})C_{\alpha}\int_{0}^{t}r^{-\alpha}e^{-\delta r}dr \\ \leq C_{\alpha}e^{-\delta}\|I_{0}\|_{p} + G_{p}(\|I_{0}\|_{\infty})C_{\alpha}\int_{0}^{\infty}r^{-\alpha}e^{-\delta r}dr \\ \leq C_{\alpha}e^{-\delta}\|I_{0}\|_{\infty} + G_{p}(\|I_{0}\|_{\infty})C_{\alpha}\delta^{-1+\alpha}\Gamma(1-\alpha), \end{split}$$
(6.20)

and thus the embedding (6.18) implies that (6.15) holds for $t \ge 1$. For $0 \le t \le 1$, it follows from (6.17) and $\|e^{-Bt}\|_{\mathscr{L}(X_+,X_+)} \le e^{-(d_2+\gamma)t}$ that

$$\begin{split} \|I(t)\|_{\infty} \\ &\leq \|e^{-Bt}I_0\|_{\infty} + \int_0^t \|e^{-B(t-s)}g_2(S(s), I(s))\|_{\infty} ds \\ &\leq \|e^{-Bt}\|_{\mathscr{L}(X,X)} \|I_0\|_{\infty} + C \int_0^t \|e^{-B(t-s)}g_2(S(s), I(s))\|_{Y^{\alpha}} ds \\ &\leq e^{-(d_2+\gamma)t} \|I_0\|_{\infty} + C \int_0^t \|B^{\alpha}e^{-B(t-s)}g_2(S(s), I(s))\|_p ds \\ &\leq \|I_0\|_{\infty} + C \int_0^t \|B^{\alpha}e^{-B(t-s)}\|_{\mathscr{L}(Y,Y)} \|g_2(S(s), I(s))\|_p ds \\ &\leq \|I_0\|_{\infty} + C \int_0^t C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)} \|g_2(S(s), I(s))\|_p ds \\ &\leq \|I_0\|_{\infty} + CG_p(\|I_0\|_{\infty})C_{\alpha} \int_0^t r^{-\alpha}e^{-\delta r} dr \\ &\leq \|I_0\|_{\infty} + CG_p(\|I_0\|_{\infty})C_{\alpha} \int_0^{\infty} r^{-\alpha}e^{-\delta r} dr \end{split}$$

and thus

$$\|I(t)\|_{\infty} \le \|I_0\|_{\infty} + CG_p(\|I_0\|_{\infty})C_{\alpha}\delta^{-1+\alpha}\Gamma(1-\alpha).$$
(6.21)

Therefore, (6.15) holds for all $t \ge 0$.

Now define a new feasible region

$$\Theta := \{ (S(x), I(x))^T \in X_+ : \|S\|_{\infty} \le \frac{\Lambda}{b_1}, \ \|I\|_{\infty} \le 2c \},$$
(6.22)

where c is from Theorem 6.4.

Corollary 6.5. Θ is a bounded absorbing set for (6.2)-(6.4) in Γ_0 . Proof. For any even number $p \geq 2$ such that $\frac{1}{2p} < \alpha < 1$,

$$\begin{split} \|g_{2}(S(t), I(t))\|_{p} \\ &\leq \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \left(\int_{0}^{1} I^{p} dx \right)^{\frac{1}{p}} \\ &\leq \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \left(\left(\int_{0}^{1} I^{p}_{0} dx - \frac{M(\|I(t)\|_{\frac{p}{2}})}{\alpha_{2}} \right) e^{-2p\alpha_{2}t} + \frac{M(\|I(t)\|_{\frac{p}{2}})}{\alpha_{2}} \right)^{\frac{1}{p}}. \end{split}$$

It can be shown that for any bounded set \mathcal{B} in Γ_0 , there is a $\eta_1 = \eta_1(\mathcal{B}) > 0$ such that for $t \ge \eta_1$,

$$\|g_{2}(S(t), I(t))\|_{p} \leq \|\beta\|_{\infty} \frac{\Lambda}{b_{1}} 2 \left(\frac{\|\beta\|_{\infty} \frac{\Lambda}{b_{1}} \frac{C}{\varepsilon_{1}} (c_{\frac{p}{2}})^{p}}{\alpha_{2}}\right)^{\frac{1}{p}} = 2g_{p}$$

holds for all $(S(t), I(t))^T \in \mathcal{B}$. Consequently, for $t \ge \eta_1$,

$$\begin{split} \|I(t)\|_{Y^{\alpha}} &= \|B^{\alpha}I(t)\|_{p} \\ \leq & \|B^{\alpha}e^{-Bt}I_{0}\|_{p} + \int_{0}^{t} \|B^{\alpha}e^{-B(t-s)}\|_{\mathscr{L}(Y,Y)}\|g_{2}(S(s),I(s))\|_{p}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{p} + \int_{0}^{t}C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}\|g_{2}(S(s),I(s))\|_{p}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{\infty} + \int_{0}^{\eta_{1}}G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}ds \\ & + \int_{\eta_{1}}^{t}2g_{p}C_{\alpha}(t-s)^{-\alpha}e^{-\delta(t-s)}ds \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{\infty} + \eta_{1}G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-\eta_{1})^{-\alpha}e^{-\delta(t-\eta_{1})} \\ & + 2g_{p}C_{\alpha}\int_{0}^{t-\eta_{1}}r^{-\alpha}e^{-\delta r}dr \\ \leq & C_{\alpha}t^{-\alpha}e^{-\delta t}\|I_{0}\|_{\infty} + \eta_{1}G_{p}(\|I_{0}\|_{\infty})C_{\alpha}(t-\eta_{1})^{-\alpha}e^{-\delta(t-\eta_{1})} \\ & + 2g_{p}C_{\alpha}\int_{0}^{\infty}r^{-\alpha}e^{-\delta r}dr, \end{split}$$

and thus there exists a $\eta_2 = \eta_2(\mathcal{B}) > \eta_1$ such that for $t > \eta_2$,

$$\|I(t)\|_{Y^{\alpha}} \leq 4g_p C_{\alpha} \int_0^{\infty} r^{-\alpha} e^{-\delta r} dr = 4g_p C_{\alpha} \delta^{-1+\alpha} \Gamma(1-\alpha).$$

Furthermore, for $t > \eta_2$,

$$\|I(t)\|_{\infty} \leq C \|I(t)\|_{Y^{\alpha}} \leq 4Cg_p C_{\alpha} \delta^{-1+\alpha} \Gamma(1-\alpha) = 2c,$$

where C is the embedding constant in (6.18). Therefore, Θ is a bounded absorbing set in Γ_0 .

Theorem 6.6. There exists a compact, connected attractor $\mathcal{A} = \omega(\Theta)$ of (6.2)-(6.4) which attracts the bounded sets in X_+ and is the maximal bounded attractor in X_+ .

Proof. The above result shows that system (6.2)-(6.4) generates a nonlinear semi-dynamical system $T(t)_{t\geq 0}: \Gamma_0 \to \Gamma_0$. Let $T(t) = (T_1(t), T_2(t))$, where $S(t) = T_1(t)S_0$ and $I(t) = T_2(t)I_0$. From (6.20), for any $t_1 \geq 1$,

$$\|I(t_1)\|_{Y^{\alpha}} \le C_{\alpha} e^{-\delta} \|I_0\|_{\infty} + M_p(\|I_0\|_{\infty}) C_{\alpha} \delta^{-1+\alpha} \Gamma(1-\alpha) := M(\|I_0\|_{\infty})$$

and thus

$$||I(t_1)||_{C^{\nu}} \le C ||I(t_1)||_{Y^{\alpha}} \le CM(||I_0||_{\infty}).$$

Since

$$C^{\nu}([0,1]) \xrightarrow[\text{comp.}]{} C([0,1]), \qquad (6.23)$$

 $T_2(t_1)$ is compact. A similar argument shows that $T_1(t_1)$ is also compact. Thus, for every bounded set $\mathcal{B} \subseteq \Gamma_0$ and $t_1 \ge 1$,

$$T(t_1)\mathcal{B} \subseteq \{ (S(x), I(x)) \in X_+ : \|S(t_1)\|_{C^{\nu}} \le N, \|I(t_1)\|_{C^{\nu}} \le M \},\$$

where N and M depend on \mathcal{B} . Theorem 1.1 of Temam [104], page 23 implies that the ω -limit set of Θ , $\mathcal{A} = \omega(\Theta)$, is a compact, connected attractor of (6.2)-(6.4) which attracts the bounded sets in Γ_0 and is the maximal bounded attractor in Γ_0 . Furthermore, Proposition 6.2 implies that Γ_{ε} is an absorbing set in X_+ when $\varepsilon > 0$. It can be shown that $\omega(\Gamma_{\varepsilon}) \subseteq \Gamma_0$. Therefore, $\mathcal{A} = \omega(\Theta)$ is a global attractor for (6.2)-(6.4) in X_+ .

6.2 Nonexistence of Periodic Solutions

In this section, Theorem 5.3 of Section 5.2 will be used to rule out the existence of nontrivial periodic orbits. A particular conclusion that can be drawn from this, as a consequence of Theorem 4.12 of Section 4.2, is the existence of at least one positive steady state solution is obtained.

The linear variational equation of (6.4) at any solution $(S(t, x), I(t, x)) \in \Theta$ of (6.2)-(6.4) is

$$\begin{cases} u_t = d_1 u_{xx} - (\beta(x)I + b_1)u - \beta(x)Sv, \\ v_t = d_2 v_{xx} + \beta(x)Iu + (\beta(x)S - (b_2 + \gamma))v, \end{cases}$$
(6.24)

with homogeneous Neumann boundary conditions. The pointwise representation of the second compound differential equation of (6.24) is

$$\frac{\partial w_{11}}{\partial t} = d_1(\Delta_1 w_{11} + \Delta_2 w_{11}) + (a_{11}(t, x_1) + a_{11}(t, x_2))w_{11} \\
+ a_{12}(t, x_1)w_{21} + a_{12}(t, x_2)w_{12}, \\
\frac{\partial w_{22}}{\partial t} = d_2(\Delta_1 w_{22} + \Delta_2 w_{22}) + (a_{22}(t, x_1) + a_{22}(t, x_2))w_{22} \\
+ a_{21}(t, x_1)w_{12} + a_{21}(t, x_2)w_{21}, \\
\frac{\partial w_{12}}{\partial t} = d_1\Delta_1 w_{12} + d_2\Delta_2 w_{12} + (a_{11}(t, x_1) + a_{22}(t, x_2))w_{12}$$
(6.25)

$$+a_{12}(t,x_1)w_{22}+a_{21}(t,x_2)w_{11}.$$

where $\Delta_i w = \frac{\partial^2}{\partial x_i \partial x_i}$ and

$$a_{11} = -(\beta(x)I + b_1), \ a_{12} = -\beta(x)S, \ a_{21} = \beta(x)I, \ a_{22} = (\beta(x)S - (b_2 + \gamma))$$

(see (3.26) of Section 3.2 for details). Consider a Lyapunov function

$$V(t) := \frac{1}{2!} \int_{(0,L)^2} \left((w_{11})^2 + (w_{22})^2 + 2(w_{12})^2 \right).$$

Then the derivative of V calculated for (3.28) of Section 3.2 implies that

$$\frac{d^+V}{dt} \leq \int_{(0,L)^2} \left(\mu_1(t)(w_{11})^2 + \mu_2(t)(w_{22})^2 + 2\mu_3(t)(w_{12})^2 \right)$$
(6.26)

where

$$\begin{split} \mu_1(t) &= -d_1\pi^2 - 2b_1 + \frac{1}{\varepsilon} \|\beta\|_{\infty} \frac{\Lambda}{b_1} + \frac{2c}{\varepsilon} \|\beta\|_{\infty} \\ \mu_2(t) &= -d_2\pi^2 - 2(b_2 + \gamma) + 2\|\beta\|_{\infty} \frac{\Lambda}{b_1} + \frac{2c}{\varepsilon} \|\beta\|_{\infty} + \frac{1}{\varepsilon} \|\beta\|_{\infty} \frac{\Lambda}{b_1} \\ \mu_3(t) &= -b_1 - (b_2 + \gamma) + \|\beta\|_{\infty} \frac{\Lambda}{b_1} + \varepsilon \|\beta\|_{\infty} \frac{\Lambda}{b_1} + 2c\varepsilon \|\beta\|_{\infty} \end{split}$$

where $\|\beta\|_{\infty} = \max_{0 \le x \le 1} \beta(x)$, $\varepsilon >$ is an arbitrary constant and the constant c is defined by (6.22).

Theorem 6.7. If the diffusion terms d_1 and d_2 are sufficiently large and

$$b_1 + (b_2 + \gamma) > \|\beta\|_{\infty} \frac{\Lambda}{b_1},$$

then system (6.2)-(6.4) has no periodic solutions in Θ and thus there exists a positive, x-dependent steady state solution.

Appendix A Lozinskiĭ Measures

This appendix is a summary on Lozinskiĭ measures. Detailed information on Lozinskiĭ measures of matrices will be referred to Coppel [19], Muldowney [76–78] and of a bounded linear operators on a Banach space will be referred to Daleckiĭ and Kreĭn [23], page 61.

Let X be a Banach space. For any $x, y \in X$, the limit

$$\partial_y \|x\| := \lim_{h \to 0^+} \frac{\|x + hy\| - \|x\|}{h},$$
 (A.1)

always exists, see Coppel [19], page 3 and Daleckiĭ and Kreĭn [23], page 61. Let $\mathscr{B}(X)$ denote the space of all bounded linear operators from X to X.

Definition A.1. For any $A \in \mathscr{B}(X)$, the *Lozinskiĭ measure* of A is defined by

$$\mu(A) := \lim_{h \to 0^+} \frac{\|I + hA\| - 1}{h}.$$
 (A.2)

Remark A.1. It follows from (A.1) that $\mu(A)$ exists. The "Lozinskii measure" μ had previously been called the "Lozinskii logarithmic norm"; however, it is not a norm since it can take negative values. The term "Lozinskii measure" has been used by Coppel [19].

The following propositions are stated in Daleckiĭ and Kreĭn [23], page 61-62.

Proposition A.1. For any $A, B \in \mathscr{B}(X)$,

(i) $\mu(\alpha A) = \alpha \mu(A)$, if $\alpha \ge 0$;

- (*ii*) $|\mu(A)| \le ||A||;$
- (*iii*) $\mu(A+B) \le \mu(A) + \mu(B);$
- (*iv*) $|\mu(A) \mu(B)| \le ||A B||.$

Proposition A.2. If X is a Banach space, then

 $-\mu(-A) \leq Re\lambda(A) \leq \mu(A), \quad for \ all \ \lambda \in \sigma(A)$

where $\sigma(A)$ denotes the spectrum of A.

Proposition A.3. If X is a Hilbert space with an inner product $\langle \cdot, \cdot \rangle$, then

$$\mu(A) = \lambda_M(A_R), \qquad A_R = Re \ A = \frac{A + A^*}{2}$$

where A^* is the adjoint operator of A and $\lambda_M(A_R) = \sup_{\|x\|=1} \langle A_R x, x \rangle$. Moreover,

$$-\mu(-A) = \lambda_m(A_{\mathcal{R}})$$

where $\lambda_m(A_{\mathcal{R}}) = \inf_{\|x\|=1} \langle A_{\mathcal{R}}x, x \rangle$.

Example A.1. Let X be a Banach space of sequences of complex or real numbers, $x = \{\xi_j\}$. Then each operator $A \in \mathscr{B}(X)$ corresponds to a matrix $[a_j^k(t)]$.

$$X \qquad ||A|| \qquad \mu(A)$$

$$l_{\infty} = \left\{ x : ||x||_{\infty} = \sup_{j} |\xi_{j}| < \infty \right\} \qquad \sup_{i} \sum_{k} |a_{i}^{k}| \qquad \sup_{i} \left(\operatorname{Re} a_{i}^{i} + \sum_{k \neq i} |a_{i}^{k}| \right)$$

$$l_{1} = \left\{ x : ||x||_{l_{1}} = \sum_{j=1}^{\infty} |\xi_{j}| < \infty \right\} \qquad \sup_{k} \sum_{i} |a_{i}^{k}| \qquad \sup_{k} \left(\operatorname{Re} a_{k}^{k} + \sum_{i \neq k} |a_{i}^{k}| \right)$$

$$l_{2} = \left\{ x : ||x||_{l_{2}} = \left(\sum_{j=1}^{\infty} |\xi_{i}|^{2} \right)^{\frac{1}{2}} < \infty \right\} \qquad \sqrt{\lambda_{M}(A^{*}A)} \qquad \lambda_{M} \left(\frac{A + A^{*}}{2} \right)$$

Table A.1: Norms and Lozinskiĭ measures of A

Remark A.2. The above results are given by Daleckiĭ and Kreĭn [23] as exercises, see page 61-62. When $X = \mathbb{C}^n$ and $A = [a_i^j]$ is a $n \times n$ matrix, a similar table is given by Coppel [19], page 41.

Proposition A.4. Suppose that x(t) is a solution of

$$\frac{dx}{dt} = A(t)x, \quad x \in X, \tag{A.3}$$

where $A(t) \in \mathscr{B}(X)$. Then, for $t \geq t_0$,

$$\|x(t_0)\| \exp\left(-\int_{t_0}^t \mu(-A(s))ds\right) \le \|x(t)\| \le \|x(t_0)\| \exp\left(\int_{t_0}^t \mu(A(s))ds\right).$$
(A.4)

Corollary A.5. The equation (A.3) is unstable if

$$\liminf_{t\to\infty}\int_{t_0}^t\mu(-A(s))ds=-\infty;$$

stable if

$$\limsup_{t\to\infty}\int_{t_0}^t\mu(A(s))ds<\infty;$$

asymptotically stable if

$$\lim_{t\to\infty}\int_{t_0}^t\mu(A(s))ds=-\infty;$$

uniformly stable if

$$\mu(A(t)) \le 0, \quad for \ t \ge t_0;$$

 $uniformly \ asymptotically \ stable \ if$

$$\mu(A(t)) \le -\delta < 0, \quad for \ t \ge t_0.$$

Appendix B Sectorial Operators

This appendix is a summary on sectorial operators and detailed information is referred to Henry [42], page 16-29.

B.1 Sectorial Operators and Analytic Semigroups

Definition B.1. A linear operator A in a Banach space X is called a *sectorial operator* if it is a closed densely defined operator such that, for some $\phi \in (0, \pi/2)$ and some $M \ge 1$ and real a, the sector

$$S_{a,\phi} = \{\lambda \mid \phi \le |arg(\lambda - a)| \le \pi, \ \lambda \neq a\}$$

is in the resolvent set of A and

$$\|(\lambda - A)^{-1}\| \le \frac{M}{|\lambda - a|}$$
 for all $\lambda \in S_{a,\phi}$.

Example B.1.

- 1. If A is a bounded linear operator on a Banach space, then A is sectorial.
- 2. If A is a self adjoint densely defined operator in a Hilbert space, and if A is bounded below, then A is sectorial.
- 3. If $Au(x) = -\Delta u(x)$, $x \in \Omega$, when $u \in C_0^2(\Omega)$ where Ω is an open set in \mathbb{R}^n , and A is the closure in $L_p(\Omega)$ of $-\Delta|_{C_0^2(\Omega)}, 1 \leq p < \infty$, then A is sectorial if its resolvent set meets the left half-plane.

Definition B.2. An analytic semigroup on a Banach space is a family of continuous linear operators on X, $\{T(t)\}_{t\geq 0}$, satisfying

- (i) T(0) = I, T(t)T(s) = T(t+S) for $t, s \ge 0$,
- (ii) $T(t)x \mapsto x$ as $t \to 0^+$, for each $x \in X$,
- (*iii*) $t \mapsto T(t)x$ is real analytic on $0 < t < \infty$ for each $x \in X$.

The infinitesimal generator L of this semigroup is defined by

$$Lx = \lim_{t \to 0^+} \frac{T(t)x - x}{t},$$

its domain $\mathcal{D}(L)$ consisting of all $x \in X$ for which this limit (in X) exists and T(t) is usually written as $T(t) = e^{Lt}$.

Theorem B.1. If A is a sectorial operator, then -A is the infinitesimal generator of an analytic semigroup $\{e^{-tA}\}_{t>0}$,

$$e^{-tA} = \frac{1}{2\pi i} \int_{\Gamma} (\lambda + A)^{-1} e^{\lambda t} d\lambda,$$

where Γ is a contour in $\rho(-A)$ with $\arg \lambda \to \pm \theta$ as $|\lambda| \to \infty$ for some $\theta \in (\frac{\pi}{2}, \pi)$.

Further e^{-tA} can be continued analytically into a sector $\{t \neq 0 : |arg t| < \varepsilon\}$ containing the positive real axis, and if $\text{Re } \sigma(A) > a$, that is, if $\text{Re}\lambda > a$ whenever $\lambda \in \sigma(A)$, then for t > 0

$$\|e^{-tA}\| \le Ce^{-\alpha t}, \quad \|Ae^{-tA}\| \le \frac{C}{t}e^{-\alpha t}$$

for some constant C.

Finally
$$\frac{d}{dt}e^{-tA} = -Ae^{-tA}$$
 for $t > 0$.

B.2 Fractional Powers of Operators

Definition B.3. Suppose A is a sectorial operator and $Re \ \sigma(A) > 0$, define for any $\alpha > 0$, $1 \qquad 1 \qquad f^{\infty}$

$$A^{-\alpha} = \frac{1}{\Gamma(\alpha)} \int_0^\infty t^{\alpha-1} e^{-tA} dt,$$

where $\Gamma(\cdot)$ is the Gamma function.

Theorem B.2. If A is a sectorial operator and Re $\sigma(A) > 0$, then for any $\alpha > 0$, $A^{-\alpha}$ is a bounded linear operator on X which is one-one and satisfies

$$A^{-\alpha}A^{-\beta} = A^{-(\alpha+\beta)}$$

whenever $\alpha, \beta > 0$. Also for $0 < \alpha < 1$,

$$A^{-lpha} = rac{sin\pilpha}{\pi} \int_0^\infty \lambda^{-lpha} (\lambda+A)^{-1} d\lambda.$$

Definition B.4. Define A^{α} to be the inverse of $A^{-\alpha}$ ($\alpha > 0$), $\mathcal{D}(A^{\alpha}) = \mathcal{R}(A^{-\alpha})$. Set $A^0 = I$ on X.

Remark B.1.

1.
$$A^{\alpha} = AA^{\alpha-1} = \frac{1}{\Gamma(\alpha)} \int_0^{\infty} t^{-\alpha} A e^{-At} dt.$$

- 2. If $\alpha > 0, A^{\alpha}$ is a closed and densely defined.
- 3. If $\alpha \geq \beta$, then $\mathcal{D}(A^{\alpha}) \subset \mathcal{D}(A^{\beta})$.
- 4. $A^{\alpha}A^{\beta} = A^{\beta}A^{\alpha} = A^{\alpha+\beta}$ on $\mathcal{D}(A^{\gamma})$ where $\gamma = \max\{\alpha, \beta, \alpha+\beta\}$.
- 5. $A^{\alpha}e^{-At} = e^{-At}A^{\alpha}$ on $\mathcal{D}(A^{\alpha}), t > 0$.

Theorem B.3. If A is a sectorial operator and Re $\sigma(A) > \delta > 0$, then for any $\alpha \ge 0$, there exists $C_{\alpha} < \infty$ such that

$$||A^{-\alpha}e^{-At}|| \le C_{\alpha}t^{-\alpha}e^{-\delta t} \quad for \ t > 0,$$

and if $0 < \alpha \leq 1, x \in D(A^{\alpha})$,

$$||(e^{-At} - 1)x|| \le \frac{1}{\alpha} C_{1-\alpha} t^{\alpha} ||A^{\alpha}x||.$$

In addition, C_{α} is bounded for all α in any compact interval of $(0, \infty)$, and bounded as $\alpha \to 0^+$.

Definition B.5. Suppose A is a sectorial operator, define for any $\alpha \ge 0$,

$$X^{\alpha} = D(A_1^{\alpha})$$

with the graph norm

 $||x||_{\alpha} = ||A_1^{\alpha}x||,$

where $A_1 = A + aI$ with a chosen so $Re \sigma(A_1) > 0$.

Remark B.2. It can be shown that different choices of a give equivalent norms on X^{α} .

Theorem B.4. If A is a sectorial operator in a Banach space X, then X^{α} is a Banach space in the norm $\|\cdot\|_{\alpha}$ for $\alpha \geq 0, X^0 = X$, and for $\alpha \geq \beta \geq 0, X^{\alpha}$ is dense subspace of X^{β} with continuous inclusion. If A has compact resolvent, the inclusion $X^{\alpha} \subset X^{\beta}$ is compact when $\alpha > \beta \geq 0$.

Bibliography

- [1] R. Abraham and J. E. Marsden. *Foundations of Mechanics*. Benjamin/Cummings, London, 1978.
- [2] S. Agmon. Lectures on elliptic boundary value problems, Van Nostrand Math. Studies, no. 2. Princeton, New Jersy, 1965.
- [3] A. C. Aitken. *Determinants and matrices*. Oliver and Boyd, Edinburgh, 1956.
- [4] S. B. Angenent and B. Fiedler. The dynamics of rotating waves in scalar reaction diffusion equations. *Trans. Amer. Math. Soc.*, 307(2):545–568, 1988.
- [5] W. Arendt, C. J. K. Batty, M. Hieber, and F. Neubrander. Vectorvalued Laplace transforms and Cauchy problems. Springer, Birkhäuser, 2001.
- [6] I. Bendixson. Sur les curbes définiés par des équations différentielles. Acta Math., 24:1–88, 1901.
- [7] N. Bourbaki. Espaces Vectoriels Topologiques. Masson, Paris, 1953.
- [8] N. Bourbaki. Algèbre. Masson, Paris, 1970.
- [9] F. Brauer, P. van den Driessche, and J. (Eds) Wu. Mathematical Epidemiology. Lecture Notes in Mathematics, Vol. 1945, Springer, Berlin Heidelberg, 2008.
- [10] P. Brunovský and B. Fiedler. Numbers of zeros on invariant manifolds in reaction-diffusion equations. Nonlinear Anal., 10(2):179–193, 1986.

- [11] P. Brunovský and B. Fiedler. Simplicity of zeros in scalar parabolic equations. J. Differential Equations, 62(2):237-241, 1986.
- [12] S. Busenberg and P. van den Driessche. A method for proving the nonexistence of limit cycles. J. Math. Anal. Appl., 172(2):463-479, 1993.
- [13] V. Capasso. Global solution for a diffusive nonlinear deterministic epidemic model. SIAM J. Appl. Math., 35:274–284, 1978.
- [14] V. Capasso and D. Fortunato. Stability results for semilinear evolution equations and their application to some reaction-diffusion problems. SIAM J. Appl. Math., 39:37–47, 1980.
- [15] M. L. Cartwright. van der Pol's equation for relaxation oscillation. In S. Lefschetz, editor, Contribution to the theory of non-linear oscillations II, Annals of Mathematical Studies, No. 29, pages 3–18. Princeton University Press, Princeton, New Jersey, 1952.
- [16] R. Casten and C. Holland. Stability properties of solutions to systems of reaction-diffusion equations. *SIAM J. Appl. Math.*, 33:353–364, 1977.
- [17] R. Casten and C. Holland. Instability results for reaction diffusion equations with Neumann boundary conditions. J. Differential Equations, 27:266-273, 1978.
- [18] K. S. Cheng. Uniqueness of limit cycle for a predator-prey system. SIAM J. Math. Anal., 12:541-548, 1981.
- [19] W. A. Coppel. Stability and asymptotic behavior of differential equations. D. C. Heath and Company, Boston, 1965.
- [20] R. Courant and D. Hilbert. Methods of Mathematical Physics, Volume I. InterScience Publishers, INC., New York, 1953.
- [21] R. Courant and D. Hilbert. Methods of Mathematical Physics, Volume II Partial Differential Equations. Intervenience Publishers, a division of John Wiley & Sons, New York, 1962.

- [22] J. Cronin. Differential equations: introduction and qualitative theory. Marcel Dekker, New York, second edition, 1994.
- [23] Ju. L. Daleckiĭ and M. G. Kreĭn. Stability of solutions of differential equations in Banach spaces. Translations of Mathematical Monographs, American Mathematical Society, Vol. 43, Providence, Rhode Island, 1974.
- [24] P. deMottoni, E. Orlandi, and A. Tesei. Asymptotic behavior for a system describing epidemics with migration and spatial spread of infection. Non. Anal. Theory Math. Appl., 3:663-675, 1979.
- [25] A. A. Dorodnicyn. Asymptotic solutions of van der Pol's equation. Prikl. Mat. Mech. J., 11:313–328, 1947.
- [26] H. Dulac. Recherche des cycles limites. C. R. Acad. Sci. Paris, 204:1703-1706, 1937.
- [27] N. Dunford and B. J. Pettis. Linear operations on summable functions. Trans. Amer. Math. Soc., 47:323–392, 1940.
- [28] L. Dung and H. L. Smith. A parabolic system modeling Microbial competition in an unmixed bio-reactor. J. Differential Equations, 130:59-91, 1996.
- [29] M. J. Fabian, P. Habala, P. Hajek, M. Santalucia, J. Pelant, and V. Zizler. Functional analysis and infinite-dimensional geometry. Springer, New York, 2001.
- [30] B. Fiedler and J. Mallet-Paret. A Poincaré-Bendixson theorem for scalar reaction diffusion equations. Arch. Rational Mech. Anal., 107(4):325–345, 1989.
- [31] M. Fiedler. Additive compound matrices and an inequality for eigenvalues of symmetric stochastic matrices. *Czech. Math. J.*, 24:392– 402, 1974.
- [32] R. A. Fisher. The wave of advance of advantageous genes. Ann. Eugen. London, 7:355–369, 1937.

- [33] F. R. Gantmacher. The theory of matrices. Chelsea Publ. Co., New York, 1959.
- [34] J. A. Goldstein. Semigroups of linear operators and applications. Oxford Mathematical Monographs, Oxford University Press, New York, 1985.
- [35] R. T. Gong and S. B. Hsu. Stability analysis for a class of diffusive coupled systems with applications to population biology. *Canadian Appl. Math. Quart.*, 8(1):79–96, 2000.
- [36] M. E. Gurtin and R. C. MacCamy. On the diffusion of biological populations. *Math. Biosc.*, 33:35–49, 1977.
- [37] J. K. Hale. Ordinary differential equations. Robert E. Krieger Publishing Company, Huntington, New York, 1980.
- [38] J. K. Hale and P. Massatt. Asymptotic behavior of gradient-like systems. In A. R. Bednarek and L. Cesari, editors, *Dynamical Systems II*, pages 85–101. Academic Press, New York, 1982.
- [39] J. K. Hale and G. Raugel. Convergence in gradient-like systems with applications to PDE. Z. Angew. Math. Phys., 43(1):63-124, 1992.
- [40] P. Hartman. The existence of large or small solutions of linear differential equations. *Duke Math. J.*, 28:421–429, 1961.
- [41] P. Hartman. Ordinary differential equations. John Wiley & Sons, Inc., New York, 1964.
- [42] D. Henry. Geometric theory of semilinear parabolic equations. Lecture Notes in Mathematics, Vol. 840, Springer-Verlag, Berlin Heidelberg, 1981.
- [43] D. Henry. Some infinite-dimensional Morse-Smale systems defined by parabolic differential equations. J. Differential Equations, 59:165–205, 1985.
- [44] H. W. Hethcote. Qualitative analysis for communicable disease models. Math. Biosci., 28:335–356, 1976.

- [45] M. W. Hirsch. Systems of differential equations which are competitive or cooperative i: Limit sets. SIAM J. Math. Anal., 13(2):167–179, 1982.
- [46] R. B. Holmes. Geometric functional analysis and its application. Springer-Verlag, New York, 1975.
- [47] A. S. Householder. The theory of matrices in numerical analysis. Blaisdell Pub. Co., New York, 1964.
- [48] A. Kallen, Arcuri P., and J. D. Murrary. A simple model for the spatial spread and control of rabies. J. Theoret. Biol., 116:377–393, 1985.
- [49] A. Kolmogorov, I. Petrovsky, and N. Piscounov. Étude de l'équation de la diffusion avec croissance de la quantité de matière et son application à un problème biologique. *Mosc. Univ. Bull. Ser. Inte. Sect. A*, 1:1–25, 1937.
- [50] H. Leiva. Stability of a periodic solution for a system of parabolic equations. *Appl. Anal.*, 60(3-4):277-300, 1996.
- [51] M. Y. Li. Geometrical studies on the global asymptotic behaviors of dissipative dynamical systems. PhD thesis, University of Alberta, 1993.
- [52] M. Y. Li and J. S. Muldowney. Global stability for the SEIR model in epidemiology. *Math. Biosci.*, 125(2):155–164, 1995.
- [53] M. Y. Li and J. S. Muldowney. Lower bounds for the Hausdorff dimension of attractors. J. Dynam. Differential Equations, 7(3):457– 469, 1995.
- [54] M. Y. Li and J. S. Muldowney. A geometric approach to global stability problems. SIAM J. Math. Anal., 27(4):1070–1083, 1996.
- [55] M. Y. Li and J. S. Muldowney. Phase asymptotic semiflows, Poincaré's stability condition and the existence of stable limit cycles. J. Differential Equations, 124:425–448, 1996.
- [56] M. Y. Li and J. S. Muldowney. Dynamics of differential equations on invariant manifolds. J. Differential Equations, 168(2):295–320, 2000.

- [57] M. Y. Li and L. Wang. A criterion for stability of matrices. J. Math. Anal. Appl., 225(1):249-264, 1998.
- [58] Y. Li and J. S. Muldowney. On Bendixson's criterion. J. Differential Equations, 106(1):27–39, 1993.
- [59] N. G. Lloyd. Degree theory. Cambridge University Press, New York, 1978.
- [60] N. G. Lloyd. A note on the number of limit cycles in certain twodimensional systems. J. London Math. Soc., 20(2):277-286, 1979.
- [61] D. London. On derivations arising in differential equations. Linear and Multilinear Algebra, 4:179–189, 1976.
- [62] Y. Lou and W.-M. Ni. Diffusion, self-diffusion and cross-diffusion. J. Differential Equations, 131(1):79–131, 1996.
- [63] Y. Lou and W.-M. Ni. Diffusion vs cross-diffusion: an elliptic approach. J. Differential Equations, 154:157–190, 1999.
- [64] R. C. MacCamy. Simple population models with diffusion. Comput. Math. Appl., 9:341-344, 1983.
- [65] J. W. Macki and J. S. Muldowney. The asymptotic behaviour of solutions to linear systems of ordinary differential equations. *Pacific J. Math.*, 33:695–706, 1970.
- [66] J. Mallet-Paret and H. L. Smith. The Poincaré-Bendixson theorem for monotone cyclic feedback systems. J. Dynamics and Diff. Eqns., 2(4):367-421, 1990.
- [67] A. W. Marshall and I. Olkin. Inequalities: Theory of majorization and its applications. Academic Press, New York, 1979.
- [68] P. Massatt. The convergence of scalar parabolic equations with convection to periodic solutions. *Preprint*, 1986.
- [69] J. L. Massera. The existence of periodic solutions of systems of differential equations. Duke Math. J., 17:457–475, 1950.
- [70] H. Matano. Convergence of solutions of one-dimensional semilinear parabolic equations. J. Math. Kyoto Univ., 18(2):221-227, 1978.
- [71] H. Matano. Asymptotic behavior of solutions of semilinear heat equations on S¹. In W. M. Ni, B. Peletier, and J. Serrin, editors, Nonlinear diffusion equations and their equilibrium states II, pages 139– 162. Springer-Verlag, New York, 1988.
- [72] R. E. Megginson. An introduction to Banach space theory. Springer-Verlag, New York, 1998.
- [73] J. Mikusiński. Sur l'equation $x^{(n)} + a(t)x = 0$. Ann. Polon. Math., 1:207–221, 1955.
- [74] H. Milloux. Sur l'équation différentielle x'' + xA(t) = 0. Prace Mat. Fiz., 41:39–53, 1934.
- [75] T. Muir. The theory of determinants in the historical order of development. Macmillan, London, 1906.
- [76] J. S. Muldowney. On the dimension of the zero or infinity tending sets for linear differential equations. *Proc. Amer. Math. Soc.*, 83(4):705– 709, 1981.
- [77] J. S. Muldowney. Dichotomies and asymptotic behaviour for linear differential systems. Trans. Amer. Math. Soc., 283(2):465–484, 1984.
- [78] J. S. Muldowney. Compound matrices and ordinary differential equations. Rocky Mountain J. Math., 20(4):857–872, 1990.
- [79] J. S. Muldowney. Stable hyperbolic limit cycles. In M. Martelli, K. Cooke, E. Cumberbatch, B. Tang, and H. Thieme, editors, *Differential equations and applications to biology and to industry*, pages 393–399. World Scientific Publ., River Edge, New Jersey, 1996.
- [80] J. S. Muldowney. Implications of the stability of an orbit for its omega limit set. In A. A. Martynyuk, editor, Advances of stability theory at the end of 20th century, volume 13 of Stability and Control: Theory,

Methods and Applications, pages 217–229. Taylor & Francis, London, 2003.

- [81] J. D. Murray. Mathematical Biology. Springer, New York, third edition, 2003.
- [82] Z. Nehari. Disconjugate linear differential operators. Trans. Amer. Math. Soc., 129:500–516, 1967.
- [83] W.-M. Ni. Diffusion, cross-diffusion, and their spike-layer steady states. Notices Amer. Math. Soc., 45:9–18, 1998.
- [84] K. Nickel. Gestaltaussagen über lösungen parabolischer differentialgleichungen. J. Reine Angew. Math., 211:78–94, 1962.
- [85] A. Okubo. Diffusion and ecological problems: mathematical models. Biomathematics, 10, Springer-Verlag, New York, 1980.
- [86] A. Okubo. Diffusion and Ecological Problems: Mathematical Models. Springer, New York, 1980.
- [87] A. Pazy. Semigroups of linear operators and applications to partial differential equations. Applied Mathematical Sciences, Vol. 44, Springer-Verlag, New York, 1983.
- [88] R. R. Phelps. Gaussian null set and differentiability of Lipschitz map on Banach spaces. *Pacific J. Math.*, 77:523–531, 1978.
- [89] V. A. Pliss. Nonlocal problems of the theory of oscillations. Academic Press, New York, 1966.
- [90] J. P. Ponzo and N. Wax. On the periodic solution of the van der Pol equation. *IEEE Trans. Circuit Theory*, CT-12:135–136, 1965.
- [91] G. B. Price. Some identities in the theory of determinants. Amer. Math. Monthly, 54:75-90, 1947.
- [92] F. Rothe. Global solutions of reaction-diffusion systems. Lecture Notes in Mathematics, Vol. 1072, Springer-Verlag, Berlin, 1984.

- [93] S. H. Saperstone. Semidynamical systems in infinite dimensional spaces. Applied Mathematical Sciences, Vol. 37, Springer-Verlag, New York, 1981.
- [94] L. Schwartz. Les Tenseurs. Hermann, Paris, 1975.
- [95] B. Schwarz. Totally positive differential systems. *Pacific J. Math.*, 32:203–229, 1970.
- [96] G. R. Sell. Periodic solutions and asymptotic stability. J. Differential Equations, 2:143–157, 1966.
- [97] J. G. Skellam. Random dispersal in theoretical populations. Biometrika, 38:196-218, 1951.
- [98] R. A. Smith. The Poincaré-Bendixson theorem for certain differential equations of higher order. Proc. Roy. Soc. Edinburgh Sect. A, 83:63–79, 1979.
- [99] R. A. Smith. Existence of periodic orbits of autonomous ordinary differential equations. Proc. Roy. Soc. Edinburgh Sect. A, 85:153–172, 1980.
- [100] R. A. Smith. An index theorem and Bendixson's negative criterion for certain differential equations of higher dimension. Proc. Roy. Soc. Edinburgh Sect. A, 91:63-77, 1981.
- [101] R. A. Smith. Massera's convergence theorem for periodic nonlinear differential equations. J. Math. Anal. Appl., 120:679–708, 1986.
- [102] R. A. Smith. Some applications of Hausdorff dimension inequalities for ordinary differential equations. Proc. Roy. Soc. Edinburgh Sect. A, 104(3-4):235-259, 1986.
- [103] J. Smoller. Shock waves and reaction-diffusion equations. Springer-Verlag, New York, second edition, 1994.
- [104] R. Temam. Infinite-dimensional dynamical systems in mechanics and physics. Springer-Verlag, New York, second edition, 1997.

- [105] A. M. Turing. On the chemical basis of morphogenesis. Philos. Trans. Roy. Soc. Londer Ser. B, 237:37–72, 1952.
- [106] M. Urabe. Periodic solutions of the van der Pol's equation with damping coefficients $\lambda = 0 \sim 10$. *IRE Trans. Circuit Theory*, CT-7:382–386, 1960.
- [107] M. Urabe. Nonlinear autonomous oscillation: Analytical Theory. Academic Press, New York, 1967.
- [108] G. F. Webb. A reaction-diffusion model for a deterministic diffusive epidemic. J. Math. Anal. Appl., 84:150–161, 1981.
- [109] J. H. M. Wedderburn. Lectures on matrices. Amer. math. Soc., New York, 1934.
- [110] H. W. Wielandt. Topics in the analytic theory of matrices. In Lecture notes prepared by R. R. Meyer. University of Wisconsin, Madison, 1967.
- [111] J. Wu. Theory and applications of partial functional differential equations. Applied Mathematical Sciences, Vol. 119, Springer-Verlag, New York, 1996.
- [112] T. Yoshizawa. Stability and existence of periodic and almost periodic solutions. In W. A. Harris and Y. Sibuya, editors, *Proceedings United* States-Japan seminar on differential and functional equations, pages 411-427. Benjamin, New York, 1967.
- [113] K. Yosida. Functional Analysis. Springer-Verlag, New York, 5 edition, 1978.
- [114] T. I. Zelenjak. Stabilization of solutions of boundary value problems for a second-order parabolic equation with one space variable. *Differential'nye Uravnenija*, 4:34–45, 1968.